

National Library of Canada

Acquisitions and

Bibliothèque nationale du Canada

Direction des acquisitions et des services bibliographiques **Bibliographic Services Branch** 

395 Wellington Street Ottawa, Ontario **K1A 0N4** 

395, rue Wellington Ottava (Ontario) K1A 0N4

Your lile Votre rélérence

Our lile Notre reférence

### AVIS

The quality of this microform is heavily dependent upon the quality of the original thesis submitted for microfilming. Every effort has been made to ensure the highest quality of reproduction possible.

NOTICE

If pages are missing, contact the university which granted the degree.

Some pages may have indistinct print especially if the original pages were typed with a poor typewriter ribbon the or if university sent us an inferior photocopy.

Reproduction in full or in part of this microform is governed by the Canadian Copyright Act. R.S.C. 1970. C. C-30, and subsequent amendments.

La qualité de cette microforme dépend grandement de la qualité de thèse soumise la ลน microfilmage. Nous avons tout fait pour assurer une qualité supérieure de reproduction.

S'il manque des pages, veuillez communiquer avec l'université qui a conféré le grade.

qualité d'impression La de certaines pages peut laisser à désirer, surtout si les pages originales ont été dactylographiées à l'aide d'un ruban usé ou si l'université nous a fait parvenir une photocopie de qualité inférieure.

La reproduction, même partielle, de cette microforme est soumise à la Lo, canadienne sur le droit d'auteur, SRC 1970, c. C-30, et ses amendements subséquents.



# New Nitrogen-Phosphorus Chemistry Dependent on the 2,4,6-Tri-*tert*-Butylphenyl Substituent

1

J,

.

by

Jason A. C. Clyburne

Submitted in partial fulfilment of the requirements

for the degree of Doctor of Philosophy

at

Dalhousie University Halifax, Nova Scotia March, 1995

© Copyright by Jason A. C. Clyburne, 1995



National Library of Canada

Acquisitions and Bibliographic Services Branch

395 Wellington Street Ottawa, Ontario K1A 0N4 Bibliothèque nationale du Canada

Direction des acquisitions et des services bibliographiques

395, rue Wellington Ottawa (Ontario) K1A 0N4

Your file Votre référence

Our file Notre rélérence

THE AUTHOR HAS GRANTED AN IRREVOCABLE NON-EXCLUSIVE LICENCE ALLOWING THE NATIONAL LIBRARY OF CANADA TO REPRODUCE, LOAN, DISTRIBUTE OR SELL COPIES OF HIS/HER THESIS BY ANY MEANS AND IN ANY FORM OR FORMAT, MAKING THIS THESIS AVAILABLE TO INTERESTED PERSONS. L'AUTEUR A ACCORDE UNE LICENCE IRREVOCABLE ET NON EXCLUSIVE PERMETTANT A LA BIBLIOTHEQUE NATIONALE DU CANADA DE REPRODUIRE, PRETER, DISTRIBUER OU VENDRE DES COPIES DE SA THESE DE QUELQUE MANIERE E''' SOUS QUELQUE FORME QUE CE SOIT POUR METTRE DES EXEMPLAIRES DE CETTE THESE A LA DISPOSITION DES PERSONNE INTERESSEES.

THE AUTHOR RETAINS OWNERSHIP OF THE COPYRIGHT IN HIS/HER THESIS. NEITHER THE THESIS NOR SUBSTANTIAL EXTRACTS FROM IT MAY BE PRINTED OR OTHERWISE REPRODUCED WITHOUT HIS/HER PERMISSION. L'AUTEUR CONSERVE LA PROPRIETE DU DROIT D'AUTEUR QUI PROTEGE SA THESE. NI LA THESE NI DES EXTRAITS SUBSTANTIELS DE CELLE-CI NE DOIVENT ETRE IMPRIMES OU AUTREMENT REPRODUITS SANS SON AUTORISATION.

;

ISBN 0-612-05\_39-1

# Canadä

Alexander Cameron Name JASON ne

Dissertation Abstracts International is arranged by broad, general subject categories. Please select the one subject which most nearly describes the content of your dissertation. Enter the corresponding four-digit code in the spaces provided.

Dra ani NL SUBJECT THRM



### **Subject** Categories

### THE HUMANITIES AND SOCIAL SCIENCES

<b>COMMUNICATIONS AND THE</b>	ARTS
Architecture	0729
Art History	0377
Cinema	0900
Dance	0378
Fine Arts	0357
Information Science	0723
Journalism	0391
Library Science	0399
Mass Communications	0708
Music	0413
Speech Communication	0459
Theoter	0465

#### EDUCATION

EDD/WILLING	
General	0515
Administration	0514
Adult and Continuing	0216
Agricultural	0517
Ad	0272
Bilingual and Multicultural	0282
Business	UY BB
Comments to 1 more	
VUCTOR AT 15 YO M A 15 YO	
Early Childhood	0518
Elementary	0524
Finance	0277
Cuidenes and Counceling	0610
Guidance and Courseling	
Health	0680
Higher	0745
History of	0520
nome economics	
Industrial .	0521
10930 J.C.S	
201.6	
Music	
Philosophy of	0998
Physical	0522

Psychology	0525
Reading	0535
Religious	0527
Sciences	0714
Secondary	0533
Social Sciences	0534
Sociology of	0340
Special	0529
Teacher Training	0530
Technology	ູດກາດ
ers children in the se	. <b>.</b>
<ul> <li>M 0.1 0.10</li> </ul>	·
I MAHAOF LITTOATURE AND	

### LANGUAGE, LITERATURE AND LINGUISTICS ۵

Language	
General	0679
Ancient	
v none o	0207
longuistics	0290
Modern .	0291
Literature	
General	0401
Classical	0204
Comparative	0295
Medieval	0297
Modern	0298
African	0116
American	0591
Asian	0305
Canadian (English)	0352
Conadian (French)	0355
<u>c</u> nglisn	0393
Germanic	0311
Latin American	0312
Middle Fostern	0315
Damage Lusier in the state of the	
Komance	0313
Slavic and East European	0314

# PHILOSOPHY, RELIGION AND THEOLOGY Philosophy ......0422 **SOCIAL SCIENCES** American Studies ...... 0323 Economics General General 0501 Agricultural 0503 Commerce-Business 0505 Finance 0508 History 0509 Labor 0510 Theory 0511 Folklore 0358 Geography 0366 Gerontology 0351 0501

Ancient	057	70
Medieval	059	ιí
Modern	050	áż
Rlack	030	20
African	22	21
Asia Australia and Ossania	03	31
Canadian	23	24
European	03.	24
	033	55
Attable Content	033	ŝõ
Middle Edstern	033	53
United States	03	5/
History of Science	055	35
Law	035	18
Political Science		
General	061	5
International Law and		
Relations	061	6
Public Administration	061	7
Recreation	081	4
Social Work	045	52
Sociology		
General	062	26
Criminology and Penology	062	27
Demography	093	8
Ethnic and Racial Studies	063	ŝī
Individual and Family		
Studies	062	2
Industrial and Labor		
Relations	062	90
Public and Social Walfara	ñ4 7	ñ
Social Structure and	005	0
Dovelopment	070	ŝ
Theory and Mathede	0/0	14
Transportation	034	40
Humportation	0/0	22
Urban and kegional Planning	022	2
women s atudies	04.5	1.5

### THE SCIENCES AND ENGINEERING

#### **BIOLOGICAL SCIENCES** Aminuthu

General	0473
Aaronomy	0285
Animal Culture and	
Nutrition	0475
Animal Pathology	0476
Food Stience and	
Tabada and	0350
orestry and Wildlife	
Plant Culture .,	
Plant Pathology	0480
Piont Physiology	0812
ans e Vollagene i	
AN ANT OCCUPANT OF A DESCRIPTION OF A DE	
Biology	
General	0306
Anatomy	0287
Biostatistics	0308
Botony	0309
Cell	0379
Ecology	0320
Fotomology	0352
Genetics	0340
Limalogy	0702
Adianabialami	0/73
Microbiology	0410
MOIOCUIOF	
Neuroscience	
Oceanography	
Physiology	0433
Radiation	0821
Velerinary Science	0778
Zoology	0472
Biophysics	
General	0786
Medical .	0760
	••••
EARTH SCIENCES	
Biogeochemistry	0425
Geochemistry	0996

RTH SCIENCE	S	
ogeochemistry		425
ocnemistry .	···· ······ ····· ····· ······ ········	/70

# Geodesy 0370 Geology 0372 Geophysics 0373 Hydrology 0388 Mineralogy 0411 Palebototany 0345 Palebotany 0426 Paleocology 0418 Paleocology 0426 Paleocology 0485 HEALTH AND ENVIRONMENTAL SCIENCES á

Immunology	.0982
Medicine and Surgery	.0564
Mental Health	.0347
Nursing	.0569
Nutrition	0570
Obstetrics and Gynecology.	0380
Occupational Health and	
Therapy	.0354
Ophthalmology	0381
Patholegy	.0571
Pharmacology	0419
Pharmacy	0572
Physical Therapy	.0382
Public Health	.0573
Radiology	.0574
Recreation	0575

Speech Patho	loay	 
Toxicology		 
ome Economics		 0386

0495

### **PHYSICAL SCIENCES**

Pure Sciences	
Chemistry	
General	

Н

1

2

	.,0400
Agricultural	0749
Analytical	.0486
Biochemistry	0487
Inorganic	0488
Nuclear	.0738
Organic	0490
Pharmaceutical	0491
Physical	0494
Polymer	0495
Radiation	0754
Mothemotics	0405
Physics	.,0405
General	0605
Acoustics	10000
Astronomy and	0780
Astronousies	0404
Amorphosic Science	
Almospheric Science	
Electron for and Claster 1	0/ 40
Electronics and Electricity	,,000/
Lienentary Particles and	0700
righ chergy	
Fiuld and Plasma	
Molecular	.,0609
Nuclear	
Optics	0752
Rodiation	.0756
Solid State	0611
statistics	0463
Applied Sciences	
Applied Machanics	0244
Computer Science	.0340
Lompoler Science	

Engineering	
General	0537
Aerospace	0538
A	
A	•
1	
Chemical	0542
Civil	0543
Electronics and Electrical	0544
Heat and Thermodynamics.	0348
Hydraulic	0545
Industrial	0546
Marine	0547
Materials Science	0794
Mechanical	0548
Metallurgy	0743
Mining	0551
Nuclear	0552
Packaging	0549
Petroleum	0765
Sanitary and Municipal	.0554
System Science	
Geolechnology	
Plastics Tashaalaan	0/ 90
Taxtile Technology	.0793
Texnie Technology	0794
PSYCHOLOGY	
General	0621
Behavioral	0384

### P

General	.0521
Behavioral	.0384
Clinical	.0622
Developmental	.0620
Experimental	.0623
Industrial	.0624
Personality	.0625
Physiological	.0989
Psychobiology	.0349
Psychometrics	.0632
Social	.0451

 $\textcircled{\blue}{\blue}$ 

signature page

.

•

,

.

For Alison

ł

•

.

,

### Table of Contents

Table of Contents v
List of Tables viii
List of Figures ix
Abstract
List of Abbreviations Used xii
Acknowledgements xii
Chapter 1 Introduction 1
1.1 2,4,6-Tri- <i>tert</i> -Butylphenyl (Mes*)
1.2 The Chemistry of Mes*N-P 3
<b>1.3</b> The Report
Chapter 2 Stability of Low Coordinate Environments for Main Group
Elements: Theoretical and Experimental Considerations
2.1 Thermodynamic Consequences of Steric Crowding
2.2 Experimental Support for Thermodynamic Consequences of Steric
<b>Crowding</b> 11
Chapter 3 Monomeric Trihalogenophosphine Imides
3.1 Synthesis of Mes*NPCl <sub>3</sub> and Mes*NPBr <sub>3</sub> 14
3.2 Reactions of Iminophosphine Halides with Molecular Halogens 16
3.3 Formation of a Phosphine Imide by the Reaction of Mes*NPCl with
$[Ph_3C][BF_4]$ 17
3.4 Structural Features of Monomeric Trihalogenophosphine Imides 18
3.5 Steric and Electronic Control of Geometries at Nitrogen 22
Chapter 4 The Chemistry of the Phosphoazonium Cation
4.1 Coordination Chemistry of the Main Group Elements

.

4.2	Identification of the First Arene Complexes of Phosphoru	29
4.3	Structural Features of Arene Complexes of the Phosphoazonium	
	Cation	30
4.4	Spectroscopic Features of Arene Complexes of the Phosphoazonium	
	Cation	34
4.5	Isolable $\pi$ -Complexes as Models for Intermediates in Electrophilic	
	Aromatic Substitution	38
4.6	Reaction of the Phosphoazonium Cation with Hydridic Bases	41
4.7	Structural and Spectroscopic Features of Two New Phosphenium	
	Cations	41
4.8	Proposed Mechanism for Phosphoazonium Insertion Reactions	45
Chapter 5	Fluorenylidene Methylenephosphonium Cations	48
5.1	Precursors to Methylenephosphonium Cations	49
5.2	Ring Expansion Isomerism of a Fluorenylidenephosphonium	
	Cation	55
Chapter 6	Bis-Arenes as Four Site Donors to Antimony Trichloride	59
6.1	Preparation of 4 SbC' <sub>3</sub> :Bis-Arene Complexes	60
6.2	Solid State and Spectroscopic Features 4 SbCl <sub>3</sub> :Bis-Arene	
	Complexes	62
6.3	Factors Governing the Structure of SbCl <sub>3</sub> :Arene Complexes	65
Chapter 7	Future Work	69
7.1	Novel Ligands with Interesting Cations	<b>69</b>
7.2	Synthesis of New Phosphorus Cations	70
7.3	Nitrenium Cations	70
7.4	Polymers Based on the SbCl <sub>3</sub> Arene Interaction	71
Chapter	8 Experimental	72

.

•

Ş

.

L

8.1	Handling Air Sensitive Reagents	72
8.2	General Procedures	72
8.3	Specific Procedures	74
References	s	95

•

.

f

-

,

.

,

,

### List of Tables

ı

,

Table 2.1	PCModel calculated absolute strain energies (kJ mol <sup>-1</sup> ) for tetra-substituted (all <i>trans</i> ) cyclobutane rings and the corresponding <i>t</i> -alkene.	7
Table 2.2	Enthalpy of dimerization (kJ/mol) calculated for substituted alkenes incorporating strain energy.	8
Table 3.1	Selected bond lengths (Å) and angles (°) for three phosphine imides.	19
Table 4.1	Important bond lengths (Å), bond angles (°), and contact distances (Å) for arene complexes of the phosphoazonium cation. Parene measured to the centroid of the ring.	31
Table 4.2	Comparison of IR data (cm <sup>-1</sup> ) of the arene in [Mes*NP-arene] <sup>+</sup> complexes to the free arene.	34
Table 4.3	<sup>31</sup> P NMR data for arene complexes of the phosphoazonium cation (ppm).	36
Table 4.4	Solution <sup>13</sup> C and <sup>1</sup> H NMR chemical shifts ( $\delta$ , ppm, CD <sub>2</sub> Cl <sub>2</sub> ) of the arenes in [Mes*NP-arene] <sup>+</sup> complexes.	37
Table 4.5	Experimental and calculated phosphorus chemical shift tensors (in ppm) of aminophosphenium and iminophosphine centers ( $R = Mes^*$ ).	42
Table 6.1	Comparison of melting points (°C) for 4 SbCl <sub>3</sub> :bis-arene complexes.	60
Table 6.2	IR data (cm <sup>-1</sup> ) for pure bis-arenes and 4 SbCl <sub>3</sub> :bis-arene complexes.	62
Table 6.3	Comparison of Sb-arene and SbC <sub>arene</sub> distances (maxmin.) (Å) for 4 SbCl <sub>3</sub> :bis-arene complexes. Sbarene measured to the centroid of the ring.	62
Table 6.4	Sb-Cl bond lengths and SbCl contacts (Å) for SbCl <sub>3</sub> :bis-arene complexes.	63
Table 6.5	Comparison of crystallographic data for 2 SbCl <sub>3</sub> :benzene and 4 SbCl <sub>3</sub> :t-stilbene.	65

### List of Figures

Figure 2.1	Schematic representation of the thermodynamic consequences of sterically demanding substituents on the stability of iminophosphines. The substituents on $2.1$ and $2.2$ are small, such as H or Me.	10
Figure 2.2	Schematic representation of the thermodynamic consequences of sterically demanding substituents on the addition of amines to iminophosphines.	13
Figure 3.1	ORTEP view of Mes*NPCl <sub>3</sub> (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity.	21
Figure 3.2	ORTEP view of Mes*NPBr <sub>3</sub> (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity.	21
Figure 3.3	ORTEP view of Mes* $NPF_2(CPh_3)$ (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity.	22
Figure 4.1	Lewis structures illustrating charge delocalization in diaminophosphenium cations	32
Figure 4.2	PLUTO view of [Mes*NP-benzene][GaCl4] illustrating the distorted trigonal bipyramidal local coordination sphere around phosphorus.	35
Figure 4.3	PLUTO view of [Mes*NP-benzene][Ga <sub>2</sub> Cl <sub>7</sub> ] illustrating the distorted tetrahedral local coordination sphere around phosphorus.	35
Figure 4.4	PLUTO view of [Mes*N(H)PN(H)Mes*][GaCl <sub>4</sub> ] <u>4.19</u> showing two cations and one anion. The hydrogen atoms (except N-H) have been omitted for clarity.	43
Figure 4.5	ORTEP view of Mes*N(H)P=fluorenylidene <u>4.22</u> (50% probability ellipsoids). The hydrogen atoms (except N-H) have been omitted for clarity.	43
Figure 4.6	<sup>31</sup> P CP NMR spectrum for a static powder sample of [Mes*N(H)PN(H)Mes*] <sup>+</sup> <u>4.19</u> obtained on an MSL-200 spectrometer.	44
Figure 4.7	<sup>31</sup> P CP-MAS NMR spectrum for a spinning (6103 Hz) powder sample of [Mes*N(H)PN(H)Mes*] <sup>+</sup> <u>4.19</u> obtained on an Bruker AMX 400 spectrometer.	44

Figure 5.1	ORTEP view of $(iPr_2N)_2P$ -fluorenyl <u>5.9b</u> (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity.	52
Figure 5.2	ORTEP view of $(iPr_2N)_2P(Cl)$ =fluorenylidene 5.10b (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity.	52
Figure 5.3	Comparison of three fluorenyl phosphorus derivatives with fluorene and the fluorenyl anion. Distances are in Å.	54
Figure 5.4	ORTEP view of 9-di- <i>i</i> -propylamino-10-di- <i>iso</i> - propyliminium-9,10-dihydro-9-phosphaphenanthrene 5.18 (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity.	57
Figure 5.5	Proposed mechanism for the isomerization of <u>5.17a</u> .	58
Figure 6.1	ORTEP view of the 4 SbCl <sub>3</sub> :diphenylacetylene complex (50% probability ellipsoids). The hydrogen atoms on the arene rings have been omitted for clarity.	61
Figure 6.2	ORTEP view of the 4 SbCl <sub>3</sub> : <i>trans</i> -stilbene complex (50% probability ellipsoids). The hydrogen atoms on the arene rings have been omitted for clarity.	64
Figure 6.3	PLUTO view of $SbCl_3$ illustrating the local coordination environments of antimony.	64
Figure 6.4	Packing diagram of the 2 SbCl <sub>3</sub> :benzene complex illustrating the alternating layers of SbCl <sub>3</sub> and benzene.	б8
Figure 6.5	Packing diagram of the 4 SbCl <sub>3</sub> :trans-stilbene complex illustrating the alternating layers of SbCl <sub>3</sub> and trans - stilbene.	68

•

1

.

•

#### Abstract

Large, sterically demanding groups have facilitated preparation of many compounds containing low coordinate environments for main group elements. Assessment of strain energy for molecules containing main group elements in low coordinate environments indicates that thermodynamic stability with respect to oligomers and structural alternatives involving higher coordination numbers is imposed by large substituents. These results help rationalize some of the observations made in recent years.

The phosphoazonium cation [Mes\*NP]<sup>+</sup> (Mes\* = 2,4,6-tri-tert-butylphenyl) is a



phosphorus analogue of the diazonium cation, and its reaction with hydridic bases results in the formation of aminophosphenium cations. A mechanism is proposed, and the first example of an oxyphosphenium cation is reported. The first arene complexes of phosphorus (shown in figure) have been isolated and comprehensively characterized,

arene = benzene, toluene, or mesitylene and they represent stable models of  $\pi$ -complex anion = [GaCl<sub>4</sub>] or [Ga<sub>2</sub>Cl<sub>7</sub>] intermediates in electrophilic aromatic substitution.

Spectroscopic studies indicate that the coordination geometries observed in the solid are maintained in solution.

A convenient synthesis of rare monomeric trihalogenophosphine imides (Mes\*N=PX<sub>3</sub>) is described, and attempts to prepare stable methylenephosphonium cations have resulted in the discovery an unprecedented ring expansion reaction for methylenephosphoranes. The factors governing the structure of SbCl<sub>3</sub> arene complexes are discussed.

# List of Abbreviations Used

F

м

Ad	adamantyl
avg.	average
bp.	boiling point
Bu	butyl, C4H9
СР	cross polarization
Et	ethyl, C <sub>2</sub> H <sub>5</sub>
FT	Fourier Transform
HOMO	highest occupied molecular orbital
IR	infrared
LUMO	lowest unoccupied molecular orbital
MAS	magic angle spinning
Me	methyl, CH <sub>3</sub>
Mes	2,4,6-trimethylphenyl, C <sub>6</sub> H <sub>2</sub> Me <sub>3</sub>
Mes*	2,4,6-tri-tert-butylphenyl, C <sub>6</sub> H <sub>2</sub> (tert-Bu) <sub>3</sub>
MHz	megahertz
mp.	melting point
NA	not available
NMR	nuclear magnetic resonance
ppm	parts per million
Ph	phenyl, C <sub>6</sub> H <sub>5</sub>
Pr	propyl, C <sub>3</sub> H <sub>7</sub>
Ру	pyridine, NC <sub>5</sub> H <sub>5</sub>
TMS	tetramethylsilane, SiMe <sub>4</sub>

#### **Acknowledgements**

Many people have been involved in this work and without their contributions it would not have been possible.

First, I would like to thank Neil Burford, my supervisor, whose keen interest in science is reflected by his enthusiasm and ability to teach. I hope that I will provide similar experiences to students of my own.

I have experienced great satisfaction with helping several excellent honours students finish their projects and these include D. Gates, N. Aucoin, J. Wiles, and S. Warner. Very great thanks go to T. Parks, P. Losier, D. Silvert, S. Mason, L. Meadows, B. Yhard, and C. Macdonald for their opinions and discussions.

I would like to thank Dr. S. Cameron, P. Bakshi, K. Robertson, Dr. S. Sereda (Dalhousie University, DALX), and Dr. J. Richardson (Louisville) for obtaining the crystal structures. Dr. D. Hooper and, especially, M. Lumsden and G. Wu, of the Atlantic Regional Magnetic Resonance Centre have been invaluable in the collection and interpretation of NMR spectra. J. Müller's excellent design and construction of glassware has been essential. M. Chan introduced me to PCModel during her spare time

as she wrote her preliminary report.

I have benefitted greatly from conversations with and the criticisms of Drs. S. Grossert, J. Pincock, R. Boyd, R. Wasylishen, K. Eichele, S. Cameron, R. White, B. Grindley, and M. Scheer (Karlsruhe). Neil has generously allowed me the opportunity to pursue my interests in geology with Drs. D. Buckley and P. Stoffyn at the Bedford Institute of Oceanography.

I appreciate the financial support provided by Dalhousie University and the Walter C. Sumner Foundation. General lab facilities have been supported by the Natural Sciences and Engineering Research Council of Canada and the Petroleum Research Fund.

I have also enjoyed the company, support, and patience of Alison Sherwood during

the writing of this thesis. When not in the lab, we crossed the country, paddled the lakes in Mazurka (a red canoe), toured the highways of Eastern Canada in Eudromedaries (a blue 1981 Volvo), played and marched with the Halifax 33rd Pipes and Drums, danced Irish Ceili style, and enjoyed fresh coffee from Robin's Donuts. I would like to thank J., G. and F. Sherwood for providing creature comforts to a graduate student.

Lillian and Hilda (LSC) make the best coffee on campus, a fuel too often needed.

.

.

### Chapter 1 Introduction

Compounds containing either multiple bonds or electron deficient sites are considered to be low coordinate, and as such offer great synthetic potential *via* addition or coordination reactions. High reactivity and structural simplicity make these species ideal molecular building blocks, and this capacity is best illustrated by the established chemistry of alkenes, alkynes, carbenes, and aromatic compounds.<sup>1</sup>

The *p*-block elements of the second row, which include carbon, are unique because of the relative ease with which they engage in multiple  $(\sigma + \pi)$  bonding. In contrast, much of the chemistry of the heavier *p*-block elements shows a preference for  $\sigma$  bond formation; multiple bonds are rare.<sup>2</sup> Perhaps the most striking comparison is between azobenzene, (PhN=NPh) and its heavier congener, phosphobenzene (PhP=PPh). Early studies by Michaelis and Köhler reported formation of PhP=PPh *via* direct combination of PhPCl<sub>2</sub>



and PhPH<sub>2</sub>.<sup>3</sup> It was later shown by cryoscopic techniques<sup>4</sup> and X-ray crystallographic studies<sup>5</sup> that the products of this reaction were cyclic polyphosphines <u>1.1</u> and <u>1.2</u>.

Despite numerous synthetic difficulties (*i. e.*, tendency to oligomerize, high reactivity, and air and moisture sensitivity), compounds containing stable (*vide infra*)  $p_{\pi} - p_{\pi}$  bonds between heavy elements have been prepared and characterized.<sup>6</sup> Common to

all of these molecules are large, sterically demanding substituents which "protect" the reactive low coordinate site. This thesis shows that for low coordinate main group element environments, thermodynamic stability with respect to alternatives with higher coordination is imposed by large substituents.

### 1.1 2,4,6-Tri-tert-butylphenyl (Mes\*)

The first comprehensively characterized phosphobenzene (diphosphene 1.3) was synthesized in 1981 by the reaction of Mes\*PCl<sub>2</sub> and magnesium in THF.<sup>7</sup> This paper introduced the new sterically demanding group, 2,4,6-tri-*tert*-butylphenyl (Mes\*, supermesityl), as the substituent on the first molecule containing a  $3p_{\pi}$  -  $3p_{\pi}$  bond. Since



then, the Mes\* group has played a central role in the chemistry of compounds containing elements with low coordination numbers.<sup>8</sup>

Most elements adopt novel bonding environments with the Mes\* substituent. For example, compounds of the Group 13 elements (B, Al, Ga, and In) with a Mes\* substituent typically exist as three coordinate monomers.<sup>9</sup> Lower coordination numbers (*i. e.*, one and two) have been observed for Group 14 (C, Si, Ge, Sn, Pb) and Group 15 (N, P, As, Sb, Bi) elements with examples such as germylene **1.4**.<sup>10</sup> silaamidide **1.5**.<sup>11</sup> persistent radicals Mes\*O,<sup>12</sup> iminophosphines (*vide infra*), diphosphene Mes\*P=PMes\*,<sup>7,13</sup> phosphaarsene Mes\*P=AsMes\*,<sup>14,13</sup> phosphastibene Mes\*P=SbMes\*,<sup>13</sup> diarsene Mes\*As=AsMes\*,<sup>15</sup> phosphaalkyne Mes\*CP,<sup>16</sup> arsaalkyne Mes\*CAs,<sup>17</sup> methyleneoxophosphorane <u>1.6</u>,<sup>18</sup> phosphasilene <u>1.7</u>,<sup>19</sup> phosphaketene





1.7

**<u>1.8</u>**,<sup>20</sup> and phosphaallene **<u>1.9</u>**.<sup>21</sup> Finally, the stable iodoselenide Mes\*SeI represents a unique molecule containing a single Se-I bond.<sup>22</sup>

Other bulky substituents employed in a similar manner include 2,6-di-*tert*-butylphenyl,<sup>23</sup> 2,6-di-*tert*-butyl-4-methylphenyl,<sup>24</sup> adamantyl,<sup>25</sup> 1,2,3,4,5-pentamethylcyclopentadienyl,<sup>26</sup> tris(trimethylsilyl)methyl,<sup>27</sup> and the recently reported 2,4,6tris[bis(trimethylsilyl)methyl]phenyl group.<sup>28</sup> Mes\* is the most common due to its low cost, availability, and ease of derivatization *via* bromination, lithiation, and nitration.

### **1.2** The Chemistry of Mes\*N-P

Iminophosphines (R-P=N-R) represent one of the most extensive series of compounds containing dicoordinate phosphorus,<sup>29</sup> and they are observed to dimerize in the absence of sterically demanding substituents.<sup>30</sup> The most interesting and synthetically versatile iminophosphine is Mes\*NPCl.<sup>31</sup> First reported in 1988, this compound has

played an important role in many of the recent developments in the chemistry of iminophosphines and phosphorus chemistry in general. The most obvious synthetic potential of Mes\*NPCl is nucleophilic substitution of chloride ion with lithium reagents.<sup>32</sup> In addition, this compound has been part of investigations into the chemistry of iminophosphines with transition metal complexes,<sup>33</sup> alkynes,<sup>34</sup> and alkyl azides.<sup>35</sup>

More interesting is the phosphoazonium cation [Mes\*NP]<sup>+</sup>, which is obtained by halide ion abstraction from Mes\*NPCl.<sup>31</sup> Also known as the iminophosphenium cation, this cation is isoelectronic to the diazonium cation and the analogy implies a diverse chemistry. Consistently, the phosphoazonium cation is extremely reactive.

### **1.3** The Report

The chemistry of Mes\*N-P fragments plays a central role in this thesis, which begins in Chapter 2 with an assessment of the thermodynamic consequences of large substituents bound to low coordinate main group centers. Chapter 3 describes the oxidation of Mes\*NPX (X = Cl, Br, I) with halogens (Cl<sub>2</sub>, Br<sub>2</sub>, I<sub>2</sub>), as well as a novel reaction between Mes\*NPCl and [CPh<sub>3</sub>][BF<sub>4</sub>]. Chapter 4 explores the analogy between diazonium and phosphoazonium cations. A brief analysis of the coordination chemistry of main group elements is presented to introduce the first examples of arene complexes of phosphorus. Structural and spectroscopic features of five arene complexes [Mes\*NP-arene]<sup>+</sup> are reported. These compounds represent stable models of  $\pi$ -complex intermediates in electrophilic aromatic substitution reactions. The reactivity of [Mes\*NP]<sup>+</sup> with simple hydridic bases establishes the chemical similarity between diazonium and phosphonazonium cations. The preparation and spectroscopic characterization of the first oxyphosphenium cation is described. Chapter 5 focuses on the chemistry of methylenephosphonium cations and includes an unprecedented rearrangement of a phosphorus ylide. Chapter 6 describes the factors governing the coordination chemistry of SbCl<sub>3</sub> with arenes. A proposal for further research is presented, highlighting the synthetic versatility of main group element cations, particularly [Mes\*NP]<sup>+</sup>.

Note: In many cases, trivial compound names are used in preference to the IUPAC nomenclature. This is to be consistent with much of the established literature and to help the reader easily appreciate structural comparisons.

.

# <u>Chapter 2</u> Stability of Low Coordinate Environments for Main Group Elements: Theoretical and Experimental Considerations

Stability is a relative term and refers to the reluctance of a system to adjust to another energy state. There are two general types of stability: thermodynamic and kinetic. A thermodynamically stable molecule is of lower absolute energy than any other structural arrangement. A kinetically stable molecule has alternative structures of lower energy; however, access to these alternatives is prevented by an activation barrier which is insurmountable under normal conditions, such as room temperature. For a synthetic chemist, a realistic criterion for a stable molecule is isolation and storage in pure form without decay for indefinite periods of time, although shelf lifetimes of several years are more realistic. It is important to assess the thermodynamic and kinetic factors contributing to the stability of a compound in order to predict both structure and reactivity.

There has been much discussion concerning the role played by bulky substituents in the stability of low coordinate environments for main group elements. It is generally thought that large groups hinder the approach of molecules which, under normal circumstances, would oligomerize or undergo other reactions. As early as 1984, Cowley proposed that the stability of disilenes ( $R_2Si=SiR_2$ ) and diphosphenes (RP=PR) may also involve a significant thermodynamic component (*i.e.*, an oligomer possessing large groups may be thermodynamically unstable with respect to monomers containing multiple bonds).<sup>6, 36</sup>

Despite extensive use of large substituents in main group element chemistry, there has been no attempt to assess the thermodynamic effects of these bulky groups. This study examines these effects using molecular mechanics calculations on a series of model compounds. These results have been extrapolated to iminophosphines and other low coordinate main group element environments.

### 2.1 Thermodynamic Consequences of Steric Crowding

For all elements of the *p*-block other than those of the second row (B-Ne), examination of single and multiple bond energies reveals that the energy of two  $\sigma$  bonds is much greater than that of a  $\sigma$  and  $\pi$  bond (a double bond).<sup>2</sup> Consistent with this, iminophosphines can dimerize in the absence of large substituents *via* [2 + 2] cycloaddition, giving P<sub>2</sub>N<sub>2</sub> rings. Although bond energy stabilization is maximized with formation of  $\sigma$  bonds, there is a concomitant increase in strain energy due to steric interactions between substituents and ring strain for four membered rings.

Calculation of strain energy for these cyclic oligomers is restricted by the large number of atoms involved and the presence of heavy elements. To appreciate the size of the calculation required, the [2 + 2] dimer of Mes\*P=NMes\* has a molecular formula of  $C_{72}H_{116}N_2P_2$ , 192 atoms in total. Such a calculation is clearly beyond the scope of most *ab initio* methods. Semi-empirical molecular mechanics packages such as PCModel<sup>37</sup>

Table 2.1: PCModel calculated absolute strain energies (kJ/mol) for tetra-substituted (all <i>trans</i> ) cyclobutane rings and the corresponding <i>trans</i> -alkene.			
	cyclobutane	2 x t-alkene	
<b>R</b> = H	110	<b>≈</b> 0	
R = Me	100	≈ 0	
$\mathbf{R} = t\mathbf{B}\mathbf{u}$	190	10	
R = Ph	290	200	
R = Mes	360	300	
R = Mes*	1200	660	

using the MM2 force field can handle large numbers of atoms; however, parameters are not available for P-N systems. Nevertheless, a model involving various disubstituted alkenes and corresponding tetra-substituted cyclobutanes can assess the

magnitude of strain energy due to unavoidable steric crowding. The calculated strain energy for these molecules, reported to two significant figures, is presented in Table 2.1.

Minimum strain energies are obtained for alkenes and cyclobutanes with all substituents in the *trans* configuration. The strain energy calculated for cyclobutane compares well with the literature value (110 kJ/mol) and is primarily due to bond bending

to make the ring.<sup>38</sup> The calculated strain energies for the larger substituents include this energy, but also incorporate repulsions which result from non-bonded interactions (*i. e.*, steric crowding). For large substituents, this factor dominates the calculated strain energy. Strain energies are also calculated for cyclohexane rings, which, unsubstituted, can be considered strain-free.<sup>39</sup> This is not true for hexa-substituted rings with Me, Ph, Mes,

Table 2.2: Enthalpy of dimerization (kJ/mol) calculated for substituted alkenes incorporating strain energy.		
R	ΔH	
н	-30	
Me	-40	
<i>t</i> Bu	40	
Ph	-40	
Mes	-80	
Mes*	400	

*t*Bu, and Mes<sup>\*40</sup> groups in the equatorial positions, which have calculated strain energies of 50, 320, 450, 1300, and 3500 kJ/mol, respectively. The calculated values are consistent with other sterically crowded hydrocarbons such as *cis*-1,2-di-*tert*butylethylene (46 kJ/mol), tri-*tert*-butylethylene (130 kJ/mol) and the unknown tetra-*tert*-butlylethene (250 kJ/mol).<sup>41</sup> Cyclo-C<sub>4</sub>H<sub>4</sub>Mes<sup>\*</sup><sub>4</sub> and Mes<sup>\*</sup>CH=CHMes<sup>\*</sup> are also unknown; however, highly strained molecules can be prepared, such as *t*-butyl substituted tetrahedrane which has an estimated strain

energy of  $\approx 550 \text{ kJ/mol.}^{42}$ 

The calculated data show a rapid increase of strain energy with increased steric bulk of the substituent, and this trend is evident for both alkenes and cyclobutanes. The effect is more pronounced in the cyclobutane rings. Estimates of  $\Delta H_{dimerization}$  (Table 2.2) for the alkenes incorporating strain energy are calculated using carbon-carbon bond strengths (C-C 340 kJ/mol, C=C 610 kJ/mol),<sup>43</sup> and the following expression:

 $\Delta H_{\text{dimerization}} \approx (\sum \Delta H_{C=C} + \text{strain}_{\text{alkene}}) - (\sum \Delta H_{C-C} + \text{strain}_{\text{cyclobutane}})$ Most important, the dimerization of Mes\*CH=CHMes\* is estimated to be 400 kJ/mol endothermic. That is, the alkene is thermodynamically stable with respect to dimerization.

Extrapolation of these values to the heavier main group elements is enlightening. Specifically for P-N systems, bond energy considerations provide a convenient assessment of the thermodynamic stability of these compounds. The simplest cyclic diazadiphosphetidine  $P_2N_2$  ring 2.2 has four P-N bonds which account for 920 kJ/mol (4 x 230 kJ/mol P-N),<sup>44</sup> whereas for iminophosphine 2.1 the energy due to P=N bonds is 810 kJ/mol (2 x 405 kJ/mol P=N).<sup>45</sup> Therefore, with small substituents, iminophosphines are thermodynamically unstable with respect to dimerization by 110 kJ/mol (Figure 2.1). Values for strain energy in the  $P_2N_2$  ring are not available; however, it is probable that this value is between the strain energy calculated for cyclobutane (110 kJ/mol) and that for cyclo-P<sub>4</sub>H<sub>4</sub> (28 kJ/mol).<sup>46</sup>

Internuclear distances are larger for phosphorus-containing compounds, and consequently the strain energy due to steric interactions is decreased. Nevertheless, the strain energy will be comparable since C-C  $(1.47-1.53 \text{ Å})^{47}$  and P-N  $(1.59-1.77 \text{ Å})^{48}$  bond lengths are similar. An estimate for the strain energy in P-N systems is difficult, although a conservative estimate is approximately half the value calculated for the organic model. Bond energy and strain considerations predict that Mes\*-substituted iminophosphines are thermodynamically stable with respect to dimerization by 160 kJ/mol. With reference to Figure 2.1, iminophosphine **2.4** is thermodynamically stable with respect to its dimeric alternative **2.3**.

Entropy considerations ( $\Delta G^{\circ} = \Delta H^{\circ} - T\Delta S^{\circ}$ ) have not been discussed since dissociation reactions are generally entropically favoured.<sup>49</sup> With large groups, both components of  $\Delta G^{\circ}$  favour multiply bonded systems. Kinetic factors, such as the energy of activation, can prevent reaction, but the relative energy between alternative structures remains unchanged. In the absence of kinetic factors (*i. e.*, barriers to reaction are overcome), compounds will adopt the lowest energy structure.



Figure 2.1: Schematic representation of the thermodynamic consequences of sterically demanding substituents on the stability of iminophosphines. The substituents on 2.1 and 2.2 are small, such as H or Me.

## 2.2 Experimental Support for Thermodynamic Consequences of Steric Crowding<sup>50</sup>

the state

The consequences of steric shielding are expressed in addition as well as dimerization reactions of P-N systems. Dicoordinate environments for phosphorus are generally considered unstable with respect to compounds having higher coordination



(2.1)

numbers (vide supra).<sup>45</sup> Consistently, iminophosphines <u>2.5</u> readily undergo addition reactions with primary or secondary amines [Equation 2.1], giving three or four coordinate phosphorus compounds, <u>2.6</u> and <u>2.7</u>, respectively.<sup>51, 52</sup> These species adopt an equilibrium in which tautomer stability is dependent upon the substituents. <sup>53</sup>

Dr. Simon Mason<sup>54</sup> showed that bis(dialkylamino)phosphine chlorides, (NR<sub>2</sub>)<sub>2</sub>PCl, react quantitatively with Mes\*N(H)Li to give the corresponding trisaminophosphines <u>2.8</u>. These coordinatively saturated molecules (*i. e.*, three single bonds) spontaneously eliminate a molecule of secondary amine giving iminophosphines <u>2.10</u> via a tetracoordinate intermediate <u>2.9</u> [Equation 2.2]. This suggests thermodynamic preference for the dicoordinate phosphine over the tri- and tetracoordinate alternatives. This situation is schematically represented in Figure 2.2. Slower rates of elimination of amine from 2.8 result with increasingly large NR<sub>2</sub> groups (*i. e.*, a kinetic barrier).



### (2.2)

Phenylaminophosphines such as (PhNH)<sub>3</sub>P have been reported to decompose on heating *via* amine elimination.<sup>55</sup> Based on the observations with sterically crowded trisaminophosphines <u>2.8</u>, it is conceivable that amine eliminations from phenylaminophosphines proceed through iminophosphine intermediates; however, without large substituents, these intermediates are thermodynamically unstable with respect to oligomeric alternatives that are reminiscent of cyclic polyphosphines <u>1.1</u> and <u>1.2</u>.

Sterically imposed thermodynamic stability with respect to oligomers is illustrated by numerous molecules containing multiple bonds between heavy elements, such as diphosphenes (R-P=P-R) and disilenes (R<sub>2</sub>Si=SiR<sub>2</sub>). Another system where strain concepts are applicable is the Lewis acid-base adducts of R<sub>3</sub>E (E = Al, Ga, In ) compounds. It has been said that sterically demanding groups decrease the acidity of R<sub>3</sub>E acids, but no explanation is offered.<sup>56</sup> Recognizing that steric repulsion between substituents in a trigonal planar arrangement is less than that in a tetrahedral environment, adducts of R<sub>3</sub>E Lewis acids (R = a large substituent) may be thermodynamically unstable with respect to the free acid and base.

Iodoselenides (R-SeI) are thermodynamically unstable with respect to I<sub>2</sub> and a

diselane -Se-Se- ( $\Delta H = -1.3 \text{ kJ/mol}$ ).<sup>57</sup> The iodo selenide, Mes\*SeI, is unique since it is the only example of a neutral molecule with a stable Se-I bond.<sup>22</sup> This compound is considered kinetically stable;<sup>57</sup> however, Mes\*SeI is clearly thermodynamically stable with respect to I<sub>2</sub> and Mes\*Se-SeMes\* since these species adopt an equilibrium with Mes\*SeI in benzene solution, the major component being Mes\*SeI (K = 10<sup>2</sup> mol<sup>2</sup>/L<sup>2</sup>).<sup>22</sup> The origin of the thermodynamic stability is likely the release of strain energy from Mes\*Se-SeMes\*.

.....



Figure 2.2: Schematic representation of the thermodynamic consequences of sterically demanding substituents on the addition of amines to iminophosphines.

### Chapter 3 Monomeric Trihalogenophosphine Imides<sup>58</sup>

Trichlorophosphine imides (chlorophosphazenes) have been extensively studied,<sup>59</sup> and they typically adopt four-membered cyclic dimeric structures in both the solid state and in solution. As such they are categorized as cyclodiphosphazanen, and, depending on the substituents on nitrogen and phosphorus, may be covalent <u>3.1</u><sup>60</sup> or ionic <u>3.2</u>.<sup>61</sup> Hundreds of derivatives have been reported<sup>62</sup> and yet, surprisingly, only two<sup>63</sup> examples



of monomeric trihalogenophosphine imides,  $3.3a^{64}$  and  $3.3b^{65}$  have been confirmed by structural studies. The Mes\* substituent provides a unique opportunity to easily prepare monomeric compounds of the form Mes\*NPX<sub>3</sub>, which are useful synthons by virtue of their potential for substitution and elimination.

### 3.1 Synthesis of Mes\*NPCl<sub>3</sub> and Mes\*NPBr<sub>3</sub>

Preliminary observations<sup>66</sup> had shown that Mes\*NPCl reacts with chlorine in CH<sub>2</sub>Cl<sub>2</sub> to give the corresponding trichlorophosphine imide, Mes\*NPCl<sub>3</sub> [Equation. 3.1].

The compound was isolated from the reaction solids by vacuum sublimation, and its identity was determined by <sup>31</sup>P NMR, elemental analysis, and a single crystal X-ray structure which established its monomeric nature. <sup>31</sup>P NMR studies of the reaction mixtures indicated the presence of at least five other phosphorus containing compounds in



ł

solution, with Mes\*NPCl<sub>3</sub> representing only 35% (by integration) of the mixture. In addition to the generation of multiple reaction products, the isolated yield of Mes\*NPCl<sub>3</sub> was less than 20 mg.

(Dichloroiodo)benzene, which is readily prepared by direct combination of  $Cl_2$  with iodobenzene in cold (0 °C) CH<sub>2</sub>Cl<sub>2</sub>, reacts quickly and quantitatively (as shown by <sup>31</sup>P NMR) with Mes\*NPCl to form Mes\*NPCl<sub>3</sub>. Iodobenzene, the only by-product of the reaction, is easily removed under vacuum. Mes\*NPCl<sub>3</sub> can be purified by either vacuum sublimation or recrystallization from hexanes or pentanes. <sup>31</sup>P NMR studies indicate that Mes\*NPCl<sub>3</sub> is not formed by the Kirsanov reaction [Mes\*NH<sub>2</sub> + PCl<sub>5</sub>].<sup>59</sup>

In contrast to the reaction of Mes\*NPCl with Cl<sub>2</sub>, Mes\*NPBr<sub>3</sub> is generated quantitatively by the reaction of Mes\*NPBr with Br<sub>2</sub> in hexane, and red crystals of Mes\*NPBr<sub>3</sub> can be isolated in high yield. NMR studies of Mes\*NPI and I<sub>2</sub> indicate that no reaction occurs (Mes\*NPI<sub>3</sub> not formed), reflecting the decreasing oxidizing power of the halogens with increasing atomic number (*vide infra*). Attempts to prepare the corresponding trifluoride, Mes\*NPF<sub>3</sub>, *via* halide exchange of Mes\*NPCl<sub>3</sub> with SbF<sub>3</sub> and CsF were unsuccessful, and the starting material remained unchanged. No reaction is observed between Mes\*NPCl<sub>3</sub> and SbPh<sub>3</sub>.

#### **3.2** Reactions of Iminophosphine Halides With Molecular Halogens

Reactions involving the iminophosphine halides, Mes\*NPX (X = Cl, Br, l), with halogens (PhICl<sub>2</sub>, Br<sub>2</sub>, and I<sub>2</sub>) have been comprehensively assessed by <sup>31</sup>P NMR spectroscopy, and most derivatives of Mes\*NPX<sub>n</sub>X'<sub>3-n</sub> have been identified. For example, the reaction mixture of Mes\*NPCl with Br<sub>2</sub> exhibits <sup>31</sup>P NMR signals assigned to Mes\*NPCl<sub>3</sub> (-96 ppm), Mes\*NPBrCl<sub>2</sub> (-136 ppm), Mes\*NPBr<sub>2</sub>Cl (-183 ppm) and Mes\*NPBr<sub>3</sub> (-236 ppm). These compounds crystallize as a mixture of crystals from which a single crystal of Mes\*NPBr<sub>3</sub> was isolated and characterized by an X-ray study. Chemical analyses of the recrystallized bulk sample are consistent with a mixture in the relative ratios revealed in the solution <sup>31</sup>P NMR spectrum of the sample. The chlorine rich reaction products Mes\*NPBrCl<sub>2</sub> (20% by integration) and Mes\*NPCl<sub>3</sub> (trace) observed in the relation mixture are accounted for by a mechanism involving a halide exchange process. Other combinations of Mes\*NPX and dihalogens X'<sub>2</sub> reveal similar product mixtures.

Reactions of I<sub>2</sub> with Mes\*NPCl and Mes\*NPBr are an exception. Independent of stoichiometry, reaction of Mes\*NPCl and I<sub>2</sub> produces Mes\*NPCl<sub>2</sub>I (-211 ppm), of which several crystals have been isolated and characterized by X-ray crystallography. The remaining reaction products are a halide exchanging mixture of Mes\*NPCl and Mes\*NPI whose average chemical shift is dependent upon the stoichiometry. Reaction mixtures containing Mes\*NPCl and trace I<sub>2</sub> give a signal at  $\approx$  150 ppm, close to that of Mes\*NPCl (139 ppm in C<sub>6</sub>D<sub>6</sub>),<sup>31</sup> whereas reaction mixtures with an excess of I<sub>2</sub> give a signal at  $\approx$  200 ppm, close to that of Mes\*NPI (218 ppm in C<sub>6</sub>D<sub>6</sub>).<sup>31</sup> Reaction of Mes\*NPBr with I<sub>2</sub> produces a single signal at 170 ppm which has been assigned to an exchanging mixture of

Mes\*NPBr and Mes\*NPI. Mixtures of Mes\*NPBr and Mes\*NPI also show an average signal (159 ppm), indicating an exchange process which presumably takes place *via* the phosphoazonium cation, [Mes\*NP]<sup>+</sup>, or perhaps *via* trihalide anion formation,  $[X'_2X]^-$ . Mixtures of Mes\*NPCl<sub>3</sub> and Mes\*NPBr<sub>3</sub> do not show evidence for halide exchange, even after one month. On this basis, formation of the mixed halides likely involves halide exchange at the monohalide followed by oxidation of the resulting monohalides by the now mixed dihalogens. These reactions are summarized below [Equations 3.2 - 3.5]:

Mes*NPX	+	X'2	$\rightarrow$	Mes*NPX'	(3.2)
Mes*NPX'	+	<b>X</b> <sub>2</sub>	>	Mes*NPX'X <sub>2</sub>	(3.3)
Mes*NPX	+	XX'	>	Mes*NPX'X <sub>2</sub>	(3.4)
Mes*NPX	+	X'2	>	Mes*NPXX'2	(3.5)

A reaction process involving halide exchange accounts for the formation of Mes\*NPCl<sub>2</sub>I from the reaction of Mes\*NPCl + I<sub>2</sub>, but does not account for the absence of the diiodo or triiodo trihalides, Mes\*NPI<sub>2</sub>X and Mes\*NPI<sub>3</sub>. This may be due to either the low reduction potential of iodine (+0.620 V, *cf.*, Br<sub>2</sub>, +1.078 V; Cl<sub>2</sub>, +1.360 V),<sup>67</sup> or to the increase in steric demands of the halogen (Covalent radii: 0.99 Å, Cl; 1.14 Å, Br; 1.33 Å, I).<sup>67</sup>

# 3.3 Formation of a Phosphine Imide by the Reaction of Mes\*NPC1 with [Ph<sub>3</sub>C][BF<sub>4</sub>]<sup>68</sup>

Reaction of Mes\*NPCl with AlCl<sub>3</sub> produces the phosphoazonium cation, [Mes\*NP]<sup>+</sup>;<sup>31</sup> however, combination of Mes\*NPCl with stoichiometric quantities of [Ph<sub>3</sub>C][BF<sub>4</sub>] in CH<sub>2</sub>Cl<sub>2</sub>/hexanes solution results in the quantitative formation of aminodifluorophosphine <u>3.4</u>. Structural assignment for this compound is based on <sup>31</sup>P



NMR data (146 ppm,  ${}^{1}J_{PF}$  1105 Hz), which is typical of other aminodifluorophosphines: R(TMS)N-PF<sub>2</sub>,  ${}^{69}R = Ad$  (165 ppm,  ${}^{1}J_{PF}$  1208 Hz), Mes (151 ppm,  ${}^{1}J_{PF}$  1225 Hz). Compound <u>3.4</u> could not be isolated in pure form as it readily isomerizes to the difluoroalkylphosphine imide <u>3.5</u>, a rearrangement which is reminiscent of an Arbusov-type reaction.<sup>70</sup> This isomerization is likely driven by the steric crowding at the nitrogen center in <u>3.4</u>. Solution studies indicate that <u>3.5</u> is formed in excess of 90 % yield, and can be isolated readily from the reaction mixtures as light yellow block crystals.

### 3.4 Structural Features of Monomeric Trihalogenophosphine Imides

The solid state structures of Mes\*NPCl<sub>3</sub>, Mes\*NPBr<sub>3</sub> and Mes\*NPF<sub>2</sub>(CPh<sub>3</sub>) have been determined by X-ray crystallography. ORTEP views are presented in Figures 3.1-3.3. The crystal structure of Mes\*NPCl<sub>2</sub>I has also been determined and confirms the covalent monomeric arrangement; however, disorder prevents accurate assessment of bond lengths and angles. Significant intermolecular contacts are not observed in the structures of Mes\*NPCl<sub>3</sub>, Mes\*NPBr<sub>3</sub>, and Mes\*NPF<sub>2</sub>(CPh<sub>3</sub>).

The molecules all exhibit unusually large angles at the nitrogen center [Mes\*NPCl<sub>3</sub>, 160.9(3)°; Mes\*NPBr<sub>3</sub>, 160.1(9)°; Mes\*NPF<sub>2</sub>(CPh<sub>3</sub>), 156.3(3)°]. Similarly, a wide angle at nitrogen is also observed for Mes\*N=P(Cl)(NEt<sub>2</sub>)<sub>2</sub> (161.5(2)°).<sup>50</sup> For comparison, monomeric triarylphosphine imides (ArNPAr<sub>3</sub>) have geometries at the nitrogen center that are in accord with the VSEPR model. The C-N-P bond angles in <u>3.6</u>,<sup>71</sup> <u>3.7</u>,<sup>72</sup> and <u>3.8</u> 73



(130.4(3)°, 128.6(3)°, and 121°, respectively) are typical of other phosphine imides. Many derivatives of the iminato ligand (R<sub>3</sub>PN<sup>-</sup>) are known for the transition metals, and these exhibit a wide range of angles for the metal-bound nitrogen center (130-177°).<sup>74</sup> The wider angles are rationalized in terms of P-N and N-M  $\pi$ -interactions and *sp* hybridization at the nitrogen center. The phosphorus centers in Mes\*NPCl<sub>3</sub> and Mes\*NPBr<sub>3</sub> are distorted

Table 3.1: Selected bond lengths (Å) and angles (*) for three phosphine imides.			
	Mes*NPCl <sub>3</sub>	Mes*NPBr3	Mes*NPF2(CPh3)
PN	1.467(4)	1.46(1)	1.475(4)
NC <sub>Mes</sub> +	1.410(4)	1.45(1)	1.431(5)
C <sub>Mes</sub> +-N-F	160.9(3)	160.1(9)	156.3(3)

from a tetrahedral geometry, and they exhibit close to  $C_{3v}$  symmetry with the narrow angle for X-

P-X of 100°, which can be compared to the Cl-P-Cl angle in OPCl<sub>3</sub> (103.3°).<sup>75</sup>

The P-N bonds for these Mes\* substituted phosphine imides are consistently short (Mes\*NPCl<sub>3</sub>, 1.467(4) Å; Mes\*NPBr<sub>3</sub>, 1.46(1) Å; Mes\*NPF<sub>2</sub>(CPh<sub>3</sub>), 1.475(4) Å); and

are much shorter than often observed for other phosphazenes. For example, the P-N bonds in the arylphosphine imides are <u>3.6</u>, <u>3.7</u>, and <u>3.8</u> are 1.602(3) Å, 1.574(2) Å, and 1.622(5) Å, respectively. The Mes\* substituted phosphine imide Mes\*N=P(Cl)(NEt<sub>2</sub>)<sub>2</sub> also has a short P=N bond (1.492(2) Å).<sup>50</sup> The P-N bond in Mes\*NPCl<sub>3</sub> represents one of the shortest P-N bonds reported to date and is comparable to the P-N bond observed for the phosphoazonium cation, [Mes\*NP]<sup>+</sup>, 1.475(8) Å.<sup>31</sup> This molecule is considered to contain a P-N triple bond.<sup>76,77</sup>

MacAuliffe has recently shown that the solid state structures of diiodo- and dibromophosphoranes adopt linear 'spoke' structures <u>3.9</u> involving tetracoordinate phosphorus centers, rather than the expected pentacoordinate arrangement.<sup>78</sup> The linear spoke arrangement is not observed for the phosphine imides, with the preferential adoption of tetracoordinate environments for phosphorus rather than a three coordinate trigonal planar geometry. The tricoordinate phosphonium center has been demonstrated to be thermodynamically unstable with respect to other structural alternatives (*vide infra*).



<u>3.9</u>


<u>Figure 3.1:</u> ORTEP view of Mes\*NPCl<sub>3</sub> (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity. Important bond lengths (Å) and angles (\*): P-N 1.467(4), P-Cl(1) 2.017(1), P-Cl(2) 2.004(1), N-C<sub>Mes\*</sub> 1.410(4), C<sub>Mes\*</sub>-N-P 160.9(3), Cl(1)-P-Cl(2) 100.71(5), N-P-Cl(1) 118.9(2), N-P-Cl(2) 116.69(9).



**<u>Figure 3.2</u>**: ORTEP view of Mes\*NPBr<sub>3</sub> (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity. Important bond lengths (Å) and angles (\*): P-N 1.46(1), P-Br(1) 2.165(4), P-Br(2) 2.171(4), P-Br(3) 2.148(4), N-C<sub>Mes\*</sub> 1.45(1), C<sub>Mes\*</sub>-N-P 160.1(9), Br(1)-P-Br(2) 101.8(2), Br(1)-P-Br(3) 100.8(2), Br(2)-P-Br(3) 100.2(2), N-P-Br(1) 117.2(5), N-P-Br(2) 118.5(4), N-P-Br(2) 115.4(4).



Figure 3.3: ORTEP view of Mes\*NPF<sub>2</sub>(CPh<sub>3</sub>) (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity. Important bond lengths (Å) and angles (\*): P-N 1.475(4), P-C(19) 1.847(4), P-F(1) 1.557(3), P-F(2) 1.562(3), C<sub>Mes\*</sub>-N 1.563(3), P-N-C<sub>Mes\*</sub> 156.3(3), F(1)-P-F(2) 96.97(15), F(1)-P-N 117.7(2), F(1)-P-C(19) 99.3(2), F(2)-P-N 114.9(2), F(2)-P-C(19) 104.8(2), N-P-C(19) 119.7(2).

#### 3.5 Steric and Electronic Control of Geometries at Nitrogen

Trihalogenophosphine imides typically dimerize (*vide supra*); however, dimerization of Mes\*NPX<sub>3</sub> derivatives is not anticipated since the dimeric alternatives should possess a large strain energy due to steric interactions. That is, the monomeric Mes\*NPX<sub>3</sub> derivatives are thermodynamically stable with respect to dimerization.

Large bond angles at nitrogen and short P-N bonds are evident in Mes\*NPCl<sub>3</sub>, Mes\*NPBr<sub>3</sub>, and Mes\*NPF<sub>2</sub>(CPh<sub>3</sub>). For comparison, the P-N bond length of Mes\*NPCl<sub>3</sub> is significantly shorter and the C-N-P angle is substantially larger than those in the only other examples of monomeric trichlorides <u>3.3a</u> [1.504(2) Å and 144.6(2)°] and <u>3.3b</u> [1.505(3) Å and 142.9(3)°].

In all cases, Mes\* substituted iminophosphines and phosphine imides possess

short P-N bonds, and this is likely due to steric enforcement of linearity. For example, the P-N bond in the parent iminophosphine chloride, Mes\*NPCl,<sup>31</sup> is 1.495(4) Å and there is only a small, yet significant, shortening of the P-N bond for any of the Mes\* substituted phosphine imides.

The short P-N bond and large C-N-P angle for Mes\*NPCl<sub>3</sub>, Mes\*NPBr<sub>3</sub>, and Mes\*NPF<sub>2</sub>(CPh<sub>3</sub>) implies sp hybridization at the nitrogen center. Two factors that can affect the bond length are the extent of  $\pi$  bonding between the atoms, as well as the hybridization of the atoms ingaged in the bond. It is difficult to assess the extent of the shortening of the P-N bond due to enhanced  $\pi$  bonding, and how much is due to the sterically enforced sp hybridization at the nitrogen center. This problem has been addressed in the case of P-P bond shortening in diphosphenes (RP=PR) and diphosphines (R<sub>2</sub>P-PR<sub>2</sub>).<sup>79</sup> Another factor to consider is the ionic bonding contribution to the P-N bond which has been illustrated for monomeric aluminum aryloxides, where the lack of a significant rotation barrier around the O-Al bond suggests that  $\pi$  bonding is minimal.<sup>80</sup> The shortness of the O-Al bonds in these compounds may be explained by a significant ionic contribution to the bond, noting that a large electronegativity difference exists between oxygen and aluminum (Pauling scale: O, 3.44; Al, 1.61, diff. 1.83).<sup>67</sup> The electronegativity of nitrogen and phosphorus are comparable (Pauling scale: N, 3.04; P, 2.19; diff. 0.85).<sup>67</sup> and it is questionable if there is a significant ionic contribution to the short P-N bonds observed. It is tempting to conclude that the P-N triple bond model 3.10



3.10

is a significant resonance contributor to the electronic structure with P-N  $\pi$ -interaction enhanced in the presence of electronegative substituents.

•

•

-

••

#### Chapter 4 The Chemistry of the Phosphoazonium Cation

The diazonium cation <u>4.1</u> is one of the most versatile functional groups in organic chemistry.<sup>81</sup> The phosphoazonium cation ( $R = Mes^* 4.2$ ) can be considered a derivative of <u>4.1</u>, representing a formal replacement of the terminal nitrogen atom by a phosphorus atom. Numerous derivatives of the diazonium cation have been isolated; however, only one example of a phosphoazonium cation <u>4.2</u> has been isolated.<sup>31</sup> Recently, [MeNP]<sup>+</sup> and [MeNAs]<sup>+</sup> have been observed in the gas phase.<sup>82</sup>

The chemistry of 4.2 is still in its infancy and includes reactions with diphenylacetylene,<sup>83</sup> alkyl azides, and iminophosphines;<sup>84</sup> each giving cyclic phosphenium





cations <u>4.3</u>, <u>4.4</u>, and <u>4.5</u>, respectively. The iminophosphine, Mes\*NPCp\*, exhibits ring whizzing in the solid state and can be viewed as an  $\eta^1$  complex of the phosphoazonium cation and the Cp\* anion.<sup>85</sup>

Recent theoretical studies indicate that loss of  $N_2$  from **4.1** should occur much easier than loss of PN from **4.2** (methyl cation affinity for  $N_2$  an PN are 180 and 420 kJ/mol respectively).<sup>76</sup> In addition, calculation of the charge distribution for these cations suggests that whereas **4.1** is the best resonance structure for the diazonium cation, the charge for phosphoazonium is localized on the terminal phosphorus atom. That is, resonance structure **4.2b** dominates over **4.2a**. The phosphoazonium cation offers a unique electron deficient, coordinatively unsaturated phosphorus center as a potential coordination site which is exploited for the first time in this study. By way of introduction

a



to the coordination chemistry of 4.2, some general aspects of the coordination chemistry of main group elements are presented.

# 4.1 Coordination Chemistry of the Main Group Elements

The coordination chemistry of the metallic elements,<sup>86</sup> including those from the s and p blocks,<sup>87</sup> is well established. These compounds are structurally diverse with many types of ligands attached to the metal centers. The coordination chemistry of the lighter main group elements is not as extensive.

The concept of coordination chemistry is simply described: an electron deficient species (a Lewis acid) reacts with an electron rich species (a Lewis base) to form a coordinate covalent bond. This acid-base adduct, or complex, can be neutral, as in the case

of  $Et_2O \rightarrow BF_3$ , or charged (a complex ion), as in the anionic species  $[GaCl_4]^-$ . Other examples of anionic complex ions include  $[I_3]^-$ ,  $[SF_5]^-$  and  $[PCl_6]^-$ . These represent  $\sigma$ complexes of halide ions coordinated to non-metal acids ( $I_2$ ,  $SF_4$ , and  $PCl_5$ , respectively). Cationic species include oxonium and sulfoxonium cations,  $[Me_3O]^+$  and  $[Me_3S]^+$ , which are complexes between Me<sub>2</sub>E (E = O or S) and  $[Me]^+$ . The iodine cation  $[Py_2I]^+$  can be considered as a complex between two pyridine molecules and  $[I]^+$ .<sup>88</sup>

A recent focus of synthetic inorganic chemistry has been the preparation of cations containing elements in low coordination environments. Species such as phosphenium,<sup>89</sup> silylium,<sup>90</sup> and arsenium<sup>91</sup> cations have been prepared, and their potential as coordination sites is being studied.



The most spectacular examples of *p*-block  $\sigma$ -complex ions are the dimeric arsenium cations <u>4.6a</u>, which exist in the solid state as complex ions <u>4.6b</u>, with mutual donor/acceptor interactions (S $\rightarrow$ As or N $\rightarrow$ As) evident in each unit.<sup>91</sup> The  $\sigma$ -complex of the silylium cation, [RCN $\rightarrow$ Si(*i*Pr)<sub>3</sub>]<sup>+</sup>, has been characterized by NMR spectroscopy,<sup>92</sup> and, more recently, a novel pentacoordinate silicon cation <u>4.7</u> has been prepared.<sup>93</sup>

Several  $\sigma$ -complexes of phosphorus cations have been reported. The Lewis acidity

of phosphenium cations was first demonstrated in 1976 with the spectroscopic characterization of  $[R_3P \rightarrow PR_2]^+$  complexes in solution;<sup>94</sup> however, the first comprehensively characterized examples of  $\sigma$ -complexes of phosphenium cations are the DBN and DBU complexes of  $[(iPr_2N)_2P]^+$  4.8.<sup>95</sup> The cation 4.2 has been described as a



complex demonstrating the ability of a Lewis base (Me<sub>2</sub>PCl) to induce coordination from the prototypical phosphenium cation  $[Me_2P]^+$  to a second Lewis acid (GaCl<sub>3</sub>).<sup>96</sup> Most interesting is the donor stabilized methylenephosphenium cation <u>4.10</u>.<sup>97</sup> The report of this cation is perhaps one of the best examples of base stabilization, a concept whereby



extremely reactive, electron deficient compounds are isolated as their Lewis base adducts.<sup>98</sup> Surprisingly, only one  $\sigma$ -complex of the phosphoazonium cation <u>4.11</u> with a neutral ligand has been reported; characterized only by <sup>31</sup>P NMR spectroscopy.<sup>99</sup>

 $\pi$ -Complexes are somewhat rare for the main group element cations. There are a

number of main group element cations which engage  $\pi$ -ligands, such as pentamethylcyclopentadienide (Cp<sup>\*</sup>),<sup>100</sup> as in [Cp<sup>\*</sup><sub>2</sub>As]<sup>+</sup>,<sup>101</sup> which may be considered a non-metal analogue of the metallocenes (*e. g.*, ferrocene). Examples of  $\eta^6$ -arene complexation to the heavier *p*-block metals and metalloids are also well established.<sup>102</sup> Examples include a series of complexes involving substituted benzenes coordinated to group 15 trihalides (EX<sub>3</sub>: E = As, Sb, Bi; X = Cl, Br, I) (*vide infra*) and cationic centers such as Ga (I),<sup>103</sup> In (I)<sup>104</sup>, Sn (II)<sup>105</sup>, Pb (II).<sup>106</sup> An example of a bismuthenium cation, [Me<sub>6</sub>C<sub>6</sub>---BiCl<sub>2</sub>] [AlCl<sub>4</sub>], has been reported.<sup>107</sup> The classical benzene---dihalogen complexes,<sup>108</sup> referred to as charge transfer complexes, represent examples of  $\pi$ -complexes to neutral non-metal elements. Noticeably absent from the list of elements (cationic or neutral) which form arene complexes is phosphorus, despite extensive studies into its chemistry.<sup>109</sup>

# **4.2** Identification of the First Arene Complexes of Phosphorus<sup>110</sup>

The phosphoazonium cation, [Mes\*NP]<sup>+</sup>, was first prepared as the [AlCl<sub>4</sub>]<sup>-</sup> salt, and X-ray analysis showed the presence of toluene in the crystal lattice. Described as a solvate, the formula was reported as [Mes\*NP][AlCl<sub>4</sub>]<sup>-</sup> C<sub>7</sub>H<sub>8</sub>.<sup>31</sup> Reactions of Mes\*NPCl with a stoichiometric quantity of GaCl<sub>3</sub> in toluene quantitatively generate [Mes\*NP-toluene][GaCl<sub>4</sub>] with a <sup>31</sup>P NMR chemical shift of 78 ppm. X-ray analysis of crystalline [Mes\*NP-toluene][GaCl<sub>4</sub>] clearly reveals the presence of a significant contact between the cation, [Mes\*NP], and the arene, toluene. Stoichiometric studies of Mes\*NPCl with GaCl<sub>3</sub> in toluene indicate that with a two-fold excess of GaCl<sub>3</sub>, a compound with a <sup>31</sup>P NMR chemical shift of 91 ppm is formed. Attempts to isolate this compound are made difficult because the reaction mixtures generally separate into oily layered solutions, but crystalline material forms in high yield upon standing overnight. This material has been characterized as [Mes\*NP-toluene][Ga<sub>2</sub>Cl<sub>7</sub>], and structural analysis of this compound reveals a much closer association of the arene with the phosphoazonium cation than that found for [Mes\*NP-toluene][GaCl4].

Five compounds of the general form [Mes\*NP-arene][anion] 4.12 have been isolated and characterized: two gallate salts, [Mes\*NP-benzene][GaCl<sub>4</sub>] and



arene = benzene, toluene, or mesitylene anion =  $GaCl_4$  or  $Ga_2Cl_7$ 

### 4.12

[Mes\*NP-toluene][GaCl4], and three digallate salts, [Mes\*NP-benzene][Ga<sub>2</sub>Cl<sub>7</sub>], [Mes\*NP-toluene][Ga<sub>2</sub>Cl<sub>7</sub>], and [Mes\*NP-mesitylene][Ga<sub>2</sub>Cl<sub>7</sub>]. Observation of a single signal in the <sup>31</sup>P NMR spectra of the reaction mixtures indicates a quantitative reaction in each case. The gallate salts are susceptible to loss of the arene from the solid,<sup>111</sup> and under dynamic vacuum (24 hours at 10<sup>-3</sup> Torr) benzene and toluene are quantitatively removed as shown by IR and NMR spectroscopy and elemental analysis. In contrast, the digallate salts remain unchanged under vacuum over long periods of time (>60 hours).

# **4.3** Structural Features of Arene Complexes of the Phosphoazonium Cation

Each salt <u>4.12</u> is composed of discrete ions with tetrahedral geometry observed for  $[GaCl_4]^-$  and typical structural parameters for the previously described  $[Ga_2Cl_7]^-$  anion.<sup>112</sup>

The salt [Mes\*NP-toluene][GaCl<sub>4</sub>] is isostructural to the original report of [Mes\*NP-toluene][AlCl<sub>4</sub>].<sup>31</sup> The cation in each case is [Mes\*NP]<sup>+</sup>, and shows the structural features anticipated for the cation (Table 4.1), such as the linear  $C_{Mes}$ -N-P

<u>Table 4.1:</u> Important bond lengths (Å), bond angles (*), and contact distances (Å) for arene complexes of the phosphoazonium cation. Parene measured to the centroid of the arene ring.				
Compound	P-N	P-N-Mes*	PArene	PCl
[Mes*NP-benzene][GaCl4]	1.484(7)	175.5(7)	3.0012(6)	3.083(4) 3.389(4) 3.429(4)
[Mes*NP-toluene][GaCl4]	1.529(15)	177.0(13)	3.038(13)	3.139(9) 3.237(8) 3.474(8)
[Mes*NP-benzene][Ga2Cl7]	1.463(5)	178.5(4)	2.820(4)	3.395(3) 3.570(3)
[Mes*NP-toluene][Ga2Cl7]	1.464(9)	178.7(8)	2.767(7) ~	3.487(4) 3.584(4)
[Mes*NP-mesitylene][Ga <sub>2</sub> Cl <sub>7</sub> ]	1.471(6)	175.7(5)	2.687(4)	3.513(4) 3.805(4)

fragment with extremely short P-N bonds, irrespective of the anion or the coordinated arene.

Of particular interest is the local coordination sphere around the cation. Each compound displays a number of close contacts between the phosphorus center, the anion, and the arene. In the gallate salts the cation interacts with two anions, having phosphorus to chlorine contacts (P---Cl) as short as 3.083(4) Å. The digallate salts adopt an ion pair arrangement with the closest contacts (P---Cl) at 3.395(3) Å. Complete listings of the anion---P contacts are presented in Table 4.1. The contacts are all substantially longer than typical covalent P-Cl bonds found in PCl<sub>3</sub> (2.043 Å) and [PCl<sub>6</sub>]<sup>-</sup> (2.14 Å),<sup>113</sup> or than in the covalent compounds Mes\*NPCl (2.142(4) Å)<sup>31</sup> and Mes\*NPCl<sub>3</sub> (2.017(1) and 2.004(1) Å).<sup>58</sup> The interactions are significantly shorter than the anion-cation contacts observed in other salts of phosphorus cations. For example, the closest P---Cl contacts for

the diaminophosphenium cations 4.13 and 4.14 are 3.867(6) Å and 3.548(1) Å, respectively.<sup>114</sup> These cation-anion contacts are often observed in salts of non-metal cations, and have been interpreted as donations from the anion to the cation.<sup>115</sup> They do not appear to affect the structural features of the cation.<sup>116</sup> Differences in contact distance can be rationalized in terms of the relative acidic character of the phosphorus centers. The positive charge in diaminophosphenium cations in 4.13 and 4.14 can, by virtue of  $\pi$ -resonance, be delocalized over three atomic centers (Figure 4.1); whereas, for the

÷



Anion =  $[GaCl_4]$ 



4.14



Figure 4.1: Lewis structures illustrating charge delocalization for diaminophosphenium cations.

phosphoazonium cation there is only one adjacent nitrogen to engage in delocalization.

Significant changes occur in the local coordination geometry of the phosphoazonium cation upon changing the anion. Shown in Figures 4.2 and 4.3 are crystallographic views of [Mes\*NP-benzene]<sup>+</sup> with two different anions; [GaCl<sub>4</sub>]<sup>-</sup> and [Ga<sub>2</sub>Cl<sub>7</sub>]<sup>-</sup>. In the case of the gallate salt, three close cation-anion contacts are evident

between the phosphorus center and the chlorine centers of two anions, as well as the P---arene contact. This results in a distorted trigonal bipyramidal geometry for the local coordination sphere of the phosphorus center which is illustrated by Cl(1)a and Cl(4)b occupying axial positions, and Cl(2)a, N(1) and benzene occupying equatorial positions. The larger digallate anion affords one less P---Cl contact to the phosphorus center; thus the local coordination sphere for the phosphorus center is a distorted tetrahedron.

For all salts the arene is best described as hexa-hapto bound ( $n^6$ ) to the phosphorus center, with the arene carbon atoms equidistant from phosphorus, and the largest angle deviation from 90° between the phosphorus center, the arene centroid, and a carbon center of the arene, is  $6.2(8)^{\circ}$ . Two trends are evident in the data presented in Table 4.1: the P---arenecentroid distance is shorter in the digallate salts; and the P---arenecentroid distance decreases for the three digallate salts (benzene > toluene > mesitylene), in accord with the  $\pi$ -donor capabilities of the arene. This is surprising due to the substantial increase in steric demands of the ligand as the number of alkyl groups on the arene is increased. The benzene units in [Mes\*NP-benzene][GaCl<sub>4</sub>] and [Mes\*NP-benzene][Ga<sub>2</sub>Cl<sub>7</sub>] are each structurally identical to free benzene,<sup>117</sup> and there are no obvious distortions from planarity of the arene in any of the structures. The maximum deviations from planarity are: benzene [GaCl<sub>4</sub>]<sup>-</sup>, 0.0029 Å; [Ga<sub>2</sub>Cl<sub>7</sub>]<sup>-</sup>, 0.005 Å; toluene [GaCl<sub>4</sub>]<sup>-</sup>, 0.033 Å; [Ga<sub>2</sub>Cl<sub>7</sub>]<sup>-</sup>, 0.028 Å; and mesitylene [Ga<sub>2</sub>Cl<sub>7</sub>], 0.014 Å. Also, the P---Cl contacts increase in length as the P---arenecentroid distances decrease. Elongation of the P---Cl contacts can be rationalized in two ways: steric interactions restrict close approach of the anion; or arene coordination reduces the electrostatic attraction between phosphorus and the chlorine centers of the anion.

# **4.4** Spectroscopic Features of Arene Complexes of the Phosphoazonium Cation

The observed solid state features, specifically the change in the local coordination sphere of phosphorus with different anions and the variation of the P---arene interaction

<u>Table 4.2</u> ; Comparison of IR data $(cm^{-1})$ for the arene in [Mes*NP-arene] complexes to the free arene.			
Compound	Coordinated Arene	Free Arene	
[Mes*NP-benzene][GaCl4]	1979 1842	1960 1815	
[Mes*NP-toluene][GaCl4]	weak and broad	1943 1860 1803 1735	
[Mes*NP-benzene][Ga2Cl7]	1981 1849	1960 1815	
[Mes*NP-toluene][Ga <sub>2</sub> Cl <sub>7</sub> ]	1971 1892 1842	1943 1860 1803	
[Mes*NP-mesitylene][Ga2Cl7]	1787 1813	1735 1759	
	1784 1759	1713	

with different arenes, produce significant spectroscopic differences with respect to the free arene. The FT-IR spectra of the salts are complex; however, overtone and combination bands in the 2000-1750 cm<sup>-1</sup> region corresponding to the C--H outof-plane deformations<sup>118</sup> are well separated from other bands in the spectra. These bands are distinctive for the

• ,

arene (see Table 4.2), and their significant shifts with respect to those of the pure (free) arene<sup>119</sup> are further evidence for interaction between the arene and the phosphorus center.



Figure 4.2: PLUTO view of [Mes\*NP-benzene][GaCl<sub>4</sub>] illustrating the distorted trigonal bipyramidal local coordination sphere around phosphorus.



Figure 4.3: PLUTO view of [Mes\*NP-benzene][Ga<sub>2</sub>Cl<sub>7</sub>] illustrating the distorted tetrahedral local coordination sphere around phosphorus.

Solution NMR spectra have been obtained for all salts (Table 4.3), both in the corresponding arene and in  $CD_2Cl_2$ . <sup>31</sup>P NMR chemical shifts for the salts seem to be independent of the solvent, but there is a significant chemical shift difference (as large as 15 ppm) between the gallate and digallate salts. In general, the solution chemical shifts are

Table 4.3: <sup>31</sup> P NMR data for arene comlexes of the phosphoazonium cation (ppm).				
Compound	In Arene	In CD <sub>2</sub> Cl <sub>2</sub>	In Solid	
[Mes*NP][GaCl4]	'	76	63	
[Mes*NP-benzene][GaCl4]	78	76	NA	
[Mes*NP-toluene][GaCl4]	78	76	79111	
[Mes*NP-benzene][Ga2Cl7]	93	93	<b>90</b>	
[Mes*NP-toluene][Ga2Cl7]	91	95	89	
[Mes*NP-mesitylene][Ga2Cl7]	91	91	NA	

consistent with those observed in the solid state. The arene free salt [Mes\*NP][GaCl4] reacts slowly with dichloromethane; however, spectra obtained within minutes

of sample preparation have a similar <sup>31</sup>P chemical shift (76 ppm) as those of the benzene and toluene complexes. This is in contrast to the  $\approx$  15 ppm difference observed between [Mes\*NP][GaCl<sub>4</sub>] and [Mes\*NP-toluene][GaCl<sub>4</sub>] in the solid state spectra, and suggests that solvation of the arene free salt involves a significant interaction with dichloromethane, which has an effect comparable to that of the coordinated arene. It is interesting to note that the nature of the anion has a dramatic effect on the <sup>31</sup>P chemical shift, supporting the existence of substantial ion pairing in solution.<sup>114</sup> The coordinated arenes of the complexes in dichloromethane solution exhibit deshielding of the <sup>13</sup>C and <sup>1</sup>H nuclei (Table 4.4) with respect to those of the pure arene.<sup>120</sup>

Compound	<sup>13</sup> C NMR	<sup>1</sup> H NMR	
[Mes*NP-benzene][GaCl4]	128.7	7.38	anticipated if the
[Mes*NP-toluene][GaCl_1	C <sub>2</sub> 132.4	7.56-7.47	Parene coordination
	C <sub>3</sub> 131.4 C <sub>4</sub> 128.1 CH <sub>3</sub> 22.0	(multiplet) 7.49 Max CH <sub>3</sub> 2.48	observed in the solid state
Manih MD hannana (Co. Cl.)	120.6	7 00	is retained in $CD_2Cl_2$
[Mes*NP-toluene][Ga2Cl7]	C <sub>2</sub> 134.4	7.76-7.66	solution. The results
	$C_3 133.1$ $C_4 130.0$	(multiplet) 7.69 Max	support maintenance of
	CH3 22.5	CH3 2.30	arene coordination in
[Mes*NP- mesitylene][Ga <sub>2</sub> Cl <sub>7</sub> ]	131.1 CH <sub>3</sub> 21.6	7.21 CH <sub>3</sub> 2.42	solution, and the
benzene	128.5	7.23	deshielding of the arene
toluene	C <sub>2</sub> 129.3 C <sub>3</sub> 128.5	7.04 Max CH <sub>3</sub> 2.29	for [Mes*NP-benzene]
	C <sub>4</sub> 125.7 CH <sub>3</sub> 21.4		[Ga <sub>2</sub> Cl <sub>7</sub> ] ( <sup>13</sup> C, 128.7,
mesitylene	127.4 CH- 21.2	6.69 CH- 2.23	<sup>1</sup> H 7.38) and [Mes*NP-
	CH3 21.2	CH3 2.25	benzene][GaCl <sub>4</sub> ] ( <sup>13</sup> C,

132.6, <sup>1</sup>H 7.80) is consistent with the P---arene<sub>centroid</sub> distances observed in the solid state, 3.0012(6) Å and 2.820(4) Å, respectively.

.

1

#### 4.5 Isolable $\pi$ -Complexes as Models for Intermediates in Electrophilic Aromatic Substitution<sup>121</sup>

The electrophilic aromatic substitution reaction is one of the most important reactions in organic chemistry, and its mechanism has fascinated chemists for decades.<sup>122</sup> There are a number of intermediates proposed in this mechanism [Equation 4.1]. Kinetic



(4.1)

studies first demonstrated the existence the  $\sigma$ -complex intermediate (also known as the Wheland intermediate).<sup>123</sup> Numerous examples of stable  $\sigma$ -complexes have been characterized spectroscopically, <sup>124</sup> and recently, two arene  $\sigma$ -complexes (4.15 <sup>125, 126</sup> and 4.16)<sup>127</sup> have been isolated and characterized by X-ray crystallography. Furthermore, Olah et al., have shown that solutions of 4.15 treated with strong bases ultimately produce phenylsilanes which can be rationalized as products of electrophilic





4.16

aromatic substitution.<sup>128</sup> This observation confirms the description of these compounds as stable Wheland intermediates.

A  $\pi$ -complex intermediate was proposed for the electrophilic aromatic substitution reaction by Dewar.<sup>129</sup> Kinetic data supporting the formation of  $\pi$ -complex intermediates was provided by Olah, who defined the complex as "an electrophile interacting with the aromatic substrate with little deformation of the latter in the transition state."<sup>130</sup> In other words, the electrophile is in weak association with the arene. A rate determining formation of  $\pi$ -complex intermediates has been disputed,<sup>131</sup> and evidence for such systems is limited.

The arene (benzene, toluene and mesitylene) complexes of the phosphoazonium cation, [Mes\*NP-arene], exhibit hexahapto ( $\eta^6$ ) association of the arene with the phosphorus center and represent models of  $\pi$ -complex intermediates in electrophilic aromatic substitution. A careful examination of the literature also reveals a number of other structures which have been overlooked in this context. For example, arenes have been found to form  $\pi$ -complexes with the nitrosyl cation, [NO]<sup>+</sup>, which are reported to have much greater stability than the group 15 trihalide (*i. e.*, SbCl<sub>3</sub>) complexes (*vide infra*).<sup>132</sup>



(4.2)

The arene complexes of the phosphoazorium cation, [Mes\*NP-arene]<sup>+</sup>, provide

excellent models of  $\pi$ -complex intermediates in electrophilic aromatic substitution. The kinetic stability of the arene coordination complexes of the phosphoazonium cation [Mes\*NP-arene]<sup>+</sup> with respect to addition products is implied by the fact that reaction of the phenyliminophosphine <u>4.17</u><sup>133</sup> with trifluoromethane<sup>-------</sup> aric acid (HOSO<sub>2</sub>CF<sub>3</sub>) [Equation 3.2] quantitatively generates a compound with a <sup>31</sup>P NMR shift of 187 ppm. This signal is assigned to the aminophenylphosphenium cation <u>4.18</u>. This demonstrates the relative thermodynamic stability of the complexes with respect to the electrophilic addition product, or other possible structural arrangements. Phenylsilanes, in contrast, react when treated with strong acid to generate an arene and a silane (protodesilylation).<sup>134</sup> The presence of the large Mes\* substituent may render the  $\sigma$ -complex thermodynamically unstable with respect to the  $\pi$ -complex, thus facilitating the isolation of these intermediates.

The complexes, [Mes\*NP-arene]<sup>+</sup>, offer important comparisons with the recently reported silicon complex [Et<sub>3</sub>Si-toluene]<sup>+</sup>. The salt, [Et<sub>3</sub>Si-toluene][B(C<sub>6</sub>F<sub>5</sub>)<sub>4</sub>],<sup>125,126</sup> contains a formally cationic silicon center bound to a toluene molecule in an  $\eta^1 \pi$ coordination <sup>90</sup> or as a  $\sigma$ -complex<sup>135</sup> and is viewed as an example of a stabilized Wheland
intermediate.<sup>90</sup> The dramatic differences in coordinative strength and hapticity (E-arene
distance, Si---C = 2.18(1) Å in [Et<sub>3</sub>Si- $\eta^1$ -toluene]<sup>+</sup> <sup>125,126</sup> P---C<sub>avg</sub> = 3.31(3) Å for
[Mes\*NP- $\eta^6$ -toluene][GaCl<sub>4</sub>]; P---C<sub>avg</sub> = 3.080(3) Å for [Mes\*NP- $\eta^6$ -toluene][Ga<sub>2</sub>Cl<sub>7</sub>])
are related to the relative acidities of the cationic centers. While the silicon center of
[Et<sub>3</sub>Si]<sup>+</sup> **4.15** is truly electron deficient with the only possible resonance provided by
hyperconjugation with the  $\alpha$ -H's of the ethyl substituents, the phosphoazonium cation,
[Mes\*NP], can be described as the iminiumphosphine **4.2a**; isoelectronic to the
phosphaalkynes RCP.<sup>136</sup> In view of the significantly shorter Si---toluene interaction
observed in the crystal structure of [Et<sub>3</sub>Si-toluene][B(C<sub>6</sub>F<sub>5</sub>)<sub>4</sub>],<sup>125,126</sup> spectroscopic
features for the coordinated arene in the solid state, such as IR, should have alerted the
researchers to the significance of the interaction.

# 4.6 Reaction of the Phosphoazonium Cation with Hydridic Bases

The phosphoazonium cation reacts rapidly and quantitatively with hydridic bases resulting in formal insertion of the P-N moiety into H-N and H-O bonds. For example, reaction of  $[Mes*NP]^+$  with  $Mes*NH_2$  in  $CH_2Cl_2$  gives the symmetrical phosphenium



cation <u>4.19</u>. The structure of this molecule is established by spectroscopic techniques and is confirmed by an X-ray crystal structure. Similarly, <sup>31</sup>P NMR studies of the reaction of [Mes\*NP]<sup>+</sup> with Ph<sub>2</sub>NH show rapid formation of the phosphenium cation [Mes\*N(H)PNPh<sub>2</sub>]<sup>+</sup> (249 ppm). Reaction of [Mes\*NP]<sup>+</sup> with Mes\*OH produces a single phosphorus-containing species with a <sup>31</sup>P NMR shift of 290 ppm. The bright yellow microcrystalline powder from this reaction is characterized by elemental analysis, IR, and solution and solid state NMR as the first aryloxy-substituted phosphenium cation <u>4.20</u>.

# **4.7** Structural and Spectroscopic Features of Two New Phosphenium Cations

The structure of <u>4.19</u> exhibits the structural features typical of other aminophosphenium centers (Figure 5.1).<sup>114</sup> The phosphorus atom of <u>4.19</u> exhibits two contacts with the anion,  $[GaCl_4]^-$  (P---Cl 3.854(2) Å), as well as hydrogen bonding with each  $[GaCl_4]^-$  with two H---Cl bonds (2.52 Å). As illustrated in Figure 4.4, each  $[GaCl_4]^$ exhibits four-fold coordination with two P---Cl contacts and two H---Cl contacts. Spectroscopic data are consistent with the structures of 4.19 and 4.20. For example, characteristic IR absorptions for N-H ( $v_{NH}$  4.19 3175 cm<sup>-1</sup>, 4.20 3120 cm<sup>-1</sup>) and [GaCl<sub>4</sub>]<sup>-</sup> are evident. Also, <sup>1</sup>H NMR data confirm the symmetric nature of 4.19 and display a distinct N-H resonance at 9.94 ppm. <sup>1</sup>H NMR data for 4.20 is consistent with the formation of an aminoxyphosphenium cation with a broad N-H resonance at 10.9 ppm.

Solid state <sup>31</sup>P CP-MAS NMR data show large chemical shift anisotropy (span =  $\Omega$  =  $\delta_{11} - \delta_{33}$ ) characteristic for phosphenium<sup>137</sup> and related carbene sites.<sup>138</sup> Figures 4.6 and Figure 4.7 show, respectively, a <sup>31</sup>P CP static powder spectrum and a <sup>31</sup>P CP-MAS

<u>Table 4.5:</u> Experimental and calculated phosphorus chemical shift tensors (in ppm) of aminophosphenium and iminophosphine centers ( $R = Mes^*$ ).					
Compound	δίεο	δ <sub>11</sub>	δ22	δ33	Ω
RNH-P=NR	281	628	124	90	538
[RNH-P-NHR] <sup>+</sup>	281	624	167	51	573
[RNH-P-OR] <sup>+</sup>	308	670	124	124	546
H <sub>2</sub> N-P=NH	301	815	58	31	784
[H <sub>2</sub> N-P-NH <sub>2</sub> ] <sup>+</sup>	242	623	136	-32	655
[H <sub>2</sub> N-P-OH] <sup>+</sup>	268	692	59	54	638

powder spectrum for **4.19**. Chemical shift tensors for **4.19** and **4.20** are presented in Table 4.5. Included in this table are chemical shift tensors which have been calculated by Dr. Klaus Eichele and Prof. Rod Wasylishen for model

compounds H<sub>2</sub>NP=NH, [H<sub>2</sub>NPNH<sub>2</sub>]<sup>+</sup>, and [H<sub>2</sub>NPOH]<sup>+</sup>. These values are in good agreement with the observed Mes\* substituted compounds. Also note that the chemical shift anisotropy of **4.19** is similar to that of its conjugate base, Mes\*N(H)PNMes\* **4.21**.<sup>139</sup> It is obvious is that the formal positive charge at phosphorus is not reflected in the <sup>31</sup>P chemical shift, contrary to popular understanding.<sup>140</sup> The effect of charge on <sup>31</sup>P chemical shifts is readily appreciated when one considers the chemical shift of 328 ppm for the bare phosphorus nucleus P<sup>15+</sup>.<sup>141</sup> It is also interesting to note that **4.19** has similar structural features to **4.21**, and these include N-P-N, 103.8(5)\*; P-N, 1.633(8) Å; and P=N, 1.573(8) Å.



Figure 4.4: PLUTO view of [Mes\*N(H)PN(H)Mes\*][GaCl<sub>4</sub>] <u>4.19</u> showing two cations and one anion. The hydrogen atoms (except N-H) have been omitted for clarity. Important bond lengths (Å) and angles (\*): N-P 1.613(3), N-C<sub>Mes\*</sub> 1.473 (4), N-P-N 104.2(2), P-N-C<sub>Mes\*</sub> 126.6(2).



**Figure 4.5:** ORTEP view of Mes\*N(H)P=fluorenylidene **4.22** (50 % probability ellipsoids). The hydrogen atoms (except N-H) have been omitted for clarity. Important bond lengths (Å) and angles (\*): P-C(1) 1.706(7), P-N 1.639(6), N-C<sub>Mes\*</sub> 1.458(8), N-C<sub>Mes\*</sub> 1.458(8), N-P-C(1) 109.9(4), P-N-C<sub>Mes\*</sub> 121.2(5).



Figure 4.6: <sup>31</sup>P CP NMR spectrum for a static powder sample of [Mes\*N(H)PN(H)Mes\*] 4.19 obtained on an MSL-200 spectrometer.



Figure 4.7: <sup>31</sup>P CP-MAS NMR spectrum for a spinning (6103 Hz) powder sample of [Mes\*N(H)PN(H)Mes\*] 4.19 obtained on an Bruker AMX 400 spectrometer.

.

# **4.8** Proposed Mechanism For Phosphoazonium Cation Insertion Reactions

Formation of cations <u>4.19</u> and <u>4.20</u> represents a formal insertion of the NP group into the N-H and O-H bonds of the hydridic bases. Given the demonstrated ability of the



(4.3)

phosphoazonium cation to form complexes with both  $\sigma$  and  $\pi$  donors, a mechanism is proposed [Equation 4.3] involving a  $\sigma$ -coordination complex, followed by a 1, 3 hydrogen shift from the base to the imino-nitrogen center. 1, 3 hydrogen shifts,<sup>142,143</sup> as well as related silyl shifts,<sup>144</sup> have been documented for neutral iminophosphines, but are usually facilitated by an external base or heating, except in the case of Mes\*N(H)PNMes\*.<sup>145</sup> A slow tautomerism had previously been observed *via* Nuclear Overhauser Effects for



(4.4)

Mes\*N(H)PNMes\* [Equation 4.4], and this is confirmed *via* <sup>15</sup>N-labelled experiments. Mes\*<sup>15</sup>NPCl reacts with LiN(H)Mes\* in ether (0 °C), and isolation of the reaction product after only twenty minutes in solution indicates a 1:1 mixture of two isotopomers, Mes\*N(H)P<sup>15</sup>NMes\* and Mes\*<sup>15</sup>N(H)PNMes\*, as characterized by IR ( $v_{15}N_{H}$  3338 and  $v_{14}N_{H}$  3344 cm<sup>-1</sup>) and <sup>31</sup>P and <sup>15</sup>N NMR spectroscopy. The rapid equilibration of these two isotopomers suggests that the prototropic tautomerization [Equation 4.4] has a low kinetic barrier.

The reaction of Mes\*NPCl with fluorenyllithium [Equation 4.5] results in the quantitative formation of the aminophosphaalkene 4.22, which illustrates the prevalence of the 1, 3 hydrogen shift for dicoordinate phosphorus systems. The structure of the



phosphaalkene Mes\*N(H)-P=fluorenylidene is confirmed by the distinct  $v_{N-H}$  absorption in the IR spectrum at 3413 cm<sup>-1</sup> and the corresponding doublet in the <sup>1</sup>H NMR at 6.85 (<sup>2</sup>J<sub>PH</sub> = 11 Hz). An X-ray crystal structure of this molecule is shown in Figure 4.5.

The well established reaction of a diazonium cation with a 1° or 2° amine to give a triazene<sup>81</sup> 4.23 likely involves a similar reaction pathway [Equation 4.3]; however, since these reactions are typically performed in protic media (e. g., water or alcohol), isolation of the intermediate nitrenium cations may not be possible. The cationic intermediates may be deprotonated in these solvents to give the observed triazenes. Indeed, the 1, 3 hydrogen shift observed for Mes\*N(H)PNMes\* is analogous to tautomerisms well established for triazenes 4.23.<sup>81</sup>

These results show the synthetic versatility of the insertion of [Mes\*NP]<sup>+</sup> into element-H bonds to generate new phosphenium cations. For example, this strategy provides the first example of a cationic N-P-O framework, in spite of numerous examples of neutral oxy-substituted iminophosphines.<sup>146</sup>

••



# Chapter 5 Fluorenylidene Methylenephosphonium Cations

Sterically demanding groups can stabilize multiple bonds between heavy elements via thermodynamic destabilization of alternative structures (vide supra). In contrast, the stability of phosphenium cations 5.1 and 5.2 results from the incorporation of the low



coordinate phosphorus center within a Hückel  $10\pi$ -electron system.<sup>147</sup> The fluorenylidene unit 5.3 is an attractive substituent for the development of new bonding environments, as it offers both steric and electronic stabilizing features. Consequently, it has been used to prepare the first methylenegermane 5.4.<sup>148</sup> and also novel boron 5.5<sup>149</sup> and phosphorus compounds 5.6.<sup>150</sup>



On the basis of spectroscopic data, the first methylenephosphonium cation 5.7 was

proposed as the result of halide abstraction from a fluorenylidene substituted phosphorane.<sup>151</sup> These conclusions have been refuted after comparison with a structurally characterized methylenephosphonium cation <u>5.8</u>.<sup>152</sup> Tricoordinate phosphonium cations are isoelectronic to alkenes and are of theoretical interest.<sup>153</sup> It is surprising that very little chemistry of these compounds has been explored, and much of our knowledge of these species is dominated by the fact that they are thermodynamically unstable with respect to



intramolecular electrocyclic ring closure,<sup>154</sup> covalent alternatives,<sup>155</sup> or other decomposition pathways.<sup>156</sup>

## 5.1 Precursors to Methylenephosphonium Cations

Preparation of chlorophosphoranes involves a two step reaction sequence beginning with a chlorophosphine  $[R_2N]_2P$ -Cl, R = iPr, and R = Et. Reaction of fluorenyllithium with the  $[iPr_2N]_2P$ -Cl and  $[Et_2N]_2P$ -Cl generates the fluorenyl-phosphorus bond giving fluorenyl phosphines <u>5.9a</u> and <u>5.9b</u>, respectively. The reactions proceed quantitatively in ether solution (0 °C), and crystalline  $[iPr_2N]_2P$ -fluorenyl <u>5.9a</u> is readily obtained from hexane solution.

The structure of <u>5.9a</u> is shown in Figure 5.1 and contains a severely disordered phosphorus atom. This disorder is modelled with two phosphorus positions. The carbon

C(1) of the fluorenyl fragment is tetrahedral and the P-C bond length is 1.99(7) Å. Unfortunately, the disorder in the molecule does not allow for meaningful comparison with other P-C bonds such as in PMe<sub>3</sub> (1.85 Å),<sup>157</sup> and related pentamethylcyclopentadienyl-



phosphorus compounds such as Cp\*P=PCp\* 1.893(7) Å and 1.883(7) Å,<sup>158</sup> and [Cp\*-P-N(H)-*t*Bu][AlCl<sub>4</sub>] 1.990(2) Å.<sup>159</sup> Although in each case the Cp\* substituted compounds exhibit ring whizzing as shown by NMR spectroscopy in both solution and the solid state, the structures show attachment of the phosphorus atom to a single distorted tetrahedral carbon center of the Cp\* ring; indicating of a monohapto ( $\eta^1$ ) interaction. Recognizing fluorenyl as a derivative of Cp\*, **5.9a** can be viewed as an intimate-ion pair coordination complex between the fluorenyl anion and the diaminophosphenium cation, [(*i*Pr<sub>2</sub>N)<sub>2</sub>P]. The <sup>31</sup>P NMR chemical shift for **5.9a** (71 ppm) is substantially upfield from that of the free cation [(*i*Pr<sub>2</sub>N)<sub>2</sub>P][GaCl<sub>4</sub>] <sup>31</sup>P 313 ppm,<sup>114</sup> but is consistent with the DBU complex [(*i*Pr<sub>2</sub>N)<sub>2</sub>P  $\leftarrow$  DBU][PF<sub>6</sub>] 108 ppm.<sup>160</sup>

The phosphines <u>5.9a</u> and <u>5.9b</u> react quantitatively with CCl<sub>4</sub> to yield the chlorophosphoranes <u>5.10a</u> and <u>5.10b</u>. An ORTEP view of <u>5.10a</u> is shown in Figure 5.2 and illustrates the anticipated steric crowding. This crowding is manifest in CD<sub>2</sub>Cl<sub>2</sub> solution with <sup>1</sup>H and <sup>13</sup>C NMR spectra, indicating two distinct types of *i*Pr- groups. This implies restricted rotation around the P-N bonds. Similarly, <u>5.10b</u> exhibits a single resonance for the Et groups at room temperature; however, this resonance broadens on

cooling, suggesting a barrier to rotation around the P-N bond. The P=C bond is short, 1.698(7) Å, and is typical of other methylenephosphoranes such as  $tBu_2(Cl)P=CPh_2$ , 1.673 (5) Å, 1.667 (5) Å,<sup>156</sup> 1.668(5) Å; Ph<sub>3</sub>P=C(cyclopropyl), 1.696(6) Å;<sup>161</sup>  $iPr_3P=CMe_2$  1.731(3) Å.<sup>162</sup>



Two reasonable Lewis structures can be drawn to represent methylenephosphoranes, **5.11** (*ylene*) and **5.12** (*ylide*). There has been much controversy concerning their relative contributions; however, most experimental and theoretical data supports the dominance of the *ylide* form, with little participation of the *ylene* form.<sup>163</sup> X-ray crystal structures of methylenephosphoranes indicate slightly distorted tetrahedral environments for phosphorus (*sp*<sup>3</sup>), and in the case of compounds which lack substituents on carbon to engage in delocalization, the carbon centers are pyramidal. The shortness of the P-C bond can be attributed to electrostatic attraction between phosphorus and carbon. The upfield <sup>13</sup>C chemical shifts of the carbon centers are consistent with carbanionic character.<sup>164</sup> Caution should therefore be taken where methylenephosphoranes are illustrated in the *ylene* form. For the purposes of chemical reactivity the dipolar *ylide* form is more appropriate.

The ylidic carbon center in <u>5.10a</u> is planar, consistent with other Cp and Cp\* substituted phosphoranes; thus suggesting that the resonance structure <u>5.13a</u> is important.



Figure 5.1: ORTEP view of  $[iPr_2N]_2P$ -fluorenyl (5.9b) (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity.



**<u>Figure 5.2</u>**: ORTEP view of  $[iPr_2N]_2P(Cl)=$ fluorenylidene (<u>5.10b</u>) (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity. Important bond lengths (Å) and angles (\*): P(1)-C(1) 1.698(7), P(1)-Cl(1) 2.085(3), P(1)-N(1) 1.634(6), P(1)-N(2) 1.668(5), Cl(1)-P(1)-N(1) 102.9(2), Cl(1)-P(1)-N(2) 104.4(2), Cl(1)-P(1)-C(1) 107.3(3), N(1)-P(1)-N(2) 111.4(3), N(1)-P(1)-C(1) 117.3(3), N(2)-P(1)-C(1) 112.1(3).

in the extremely large  ${}^{1}J_{PC}$  values (5.10a, 201 Hz; 5.10b, 205 Hz), which are among the largest yet observed for methylenephosphoranes.<sup>163</sup>





The chlorine atom of the phosphorane <u>5.10a</u> is almost in the plane defined by the fluorenylidene substituent [Cl-P-C<sub>1</sub>-C<sub>2</sub> 151.5(6)\*; Cl-P-C<sub>1</sub>-C<sub>8</sub>-33.7(7)\*]. Such an arrangement precludes interaction between the  $\pi$  framework of the fluorenylidene moiety and the  $\sigma^*$  orbital of the P-Cl bond, and as a result, the P-Cl bond in <u>5.10a</u> is short (2.085(3) Å). It is argued that hyperconjugation accounts for the long P-Cl bonds in  $tBu_2(Cl)P=CPh_2$ , 2.195(2) Å, 2.228(2) Å, and 2.235(2) Å<sup>156</sup> where even weak Lewis acids, such as SnCl<sub>2</sub>, can effect halide ion abstraction to give methylenephosphonium cations.

X-ray structures of three fluorenyl and fluorenylidene compounds provide an opportunity to assess the ligand for different coordination environments at phosphorus (Figure 5.3). The bond alternation observed in fluorene <u>5.14</u><sup>165</sup> arises from the presence of a saturated C(1), whereas in the "naked" fluorenyl anion <u>5.15</u>.<sup>166</sup> the 14  $\pi$ -electron delocalized framework results in shorter bonds for bridging C(1). Based on these data, <u>5.9a</u> is best viewed as a fluorenyl ligand with a saturated C(1) center, whereas <u>5.10a</u> and <u>5.16</u> have delocalized fluorenylidene rings.





5.14











Figure 5.3: Comparison of three fluorenyl phosphorus derivatives with fluorene and the fluorenyl anion. Distances are in Å.

.

#### 5.2 Ring Expansion Isomerism of a Fluorenylidenephosphonium Cation

Methylenephosphorane 5.10a reacts rapidly with a stoichiometric amount of GaCl<sub>3</sub> in CH<sub>2</sub>Cl<sub>2</sub> to give a single phosphorus containing species with a <sup>31</sup>P NMR chemical shift of 71 ppm. This material has a very low solubility in benzene and can be quantitatively precipitated. It has not been possible to recrystallize this material because of its facile isomerization (*vide infra*). <sup>1</sup>H and <sup>13</sup>C NMR data are not informative; however, it is



5.17a

reasonable to speculate on the formation of a coordination complex involving a carbon-gallium coordinate bond, <u>5.17a</u>, similar to those reported for other ylide/Lewis acid adducts.<sup>167</sup> The complex can be viewed as a zwitterionic fluorenylphosphonium derivative with the fluorenyl C-H replaced by C-GaCl<sub>3</sub>. Formation of this covalent complex is in contrast to the ionic structure observed for [ $tBu_2P=C(SiMe_3)_2$ ][AlCl<sub>4</sub>] (<sup>31</sup>P NMR 130 ppm).<sup>168</sup> Similarly, <u>5.10b</u> reacts with GaCl<sub>3</sub> in CH<sub>2</sub>Cl<sub>2</sub> to give a single phosphorus- containing species with a chemical shift of 71 ppm.

Further support for the formation of the Lewis acid adduct is provided by the reaction of <u>5.10b</u> with HOSO<sub>2</sub>CF<sub>3</sub>. The reaction is quantitative and produces a compound whose solution NMR spectra are consistent with the formation of  $[(Et_2N)_2P(Cl)-fluorenyl][OSO_2CF_3]$ . For example, the magnitude of <sup>1</sup>J<sub>PC</sub> (79 Hz) and <sup>13</sup>C multiplicity analysis of this carbon center are in accord with the protonation of the

ylidic carbon to give a phosphonium cation.<sup>169</sup> Indeed, the *ylide* resonance structure <u>5.12</u> would predict that electrophilic attack by GaCl<sub>3</sub> would occur on the *ylidic* carbon in preference to halide ion abstraction from a phosphonium center.

The GaCl<sub>3</sub> complex <u>5.17a</u> is soluble in polar solvents such as  $CH_2Cl_2$ , but reacts giving a red solution with a single <sup>31</sup>P NMR signal at 21 ppm. The reaction is complete within eight hours. The formation of this red solution was noted in the original report; however, characterization of the compound was not attempted.<sup>151, 170</sup> Slow removal of solvent produces a red crystalline material, which is isolated in high yield and characterized as 9-di-*i*-propylamino-10-di-*i*-propyliminium-9, 10-dihydro-9-phosphaphenanthrene tetrachlorogallate <u>5.18</u>. The presence of a C=N bond is clearly revealed in the IR



spectrum with a strong absorption at 1582 cm<sup>-1</sup>,<sup>171</sup> as well as a distinct signal in the <sup>13</sup>C NMR spectrum at 213.9 ppm (d, <sup>1</sup>J<sub>PC</sub> = 89 Hz). A crystallographic view of the cation is shown in Figure 5.4.

The salt <u>5.18</u> is isomeric to the zwitterionic fluorenyl-phosphonium-GaCl<sub>3</sub> complex <u>5.17a</u>. No intermediates are observed in low temperature <sup>31</sup>P NMR studies; nevertheless, isomerization of the neutral fluorenylphosphonium in polar solvents implicates the formation of a polar intermediate, possibly the tetrachlorogallate salt of the methylenephosphonium cation <u>5.7a</u>. A proposed mechanism is presented Figure 5.5. Solutions of <u>5.17b</u> in CH<sub>2</sub>Cl<sub>2</sub> do not isomerize and remain unchanged for periods of several months. Attempts to isolate the reaction products from these solutions result in the
formation of an oily material. The stability of <u>5.17b</u>, in contrast to the facile isomerization of <u>5.17a</u>, suggests that the reaction is driven by the release of steric stresses in the fluorenyl-phosphonium-GaCl<sub>3</sub> complex due to the increased steric demands of the *N*-alkyl groups (*i. e.*, *i*Pr vs Et).

The quantitative rearrangement is unique in terms of the nature of the bonds broken and formed, and further demonstrates the thermodynamic instability of the tricoordinate phosphonium environment. Ring expansions *via* Beckman rearrangements have been reported for fluorenone oximes.<sup>172</sup> In addition, such an isomerization is unprecedented in the chemistry of phosphorus ylides despite extensive research.<sup>173</sup>



Figure 5.4: ORTEP view of 9-di-*i*-propylamino-10-di-*i*-propyliminium-9,10-dihydro-9phosphaphenanthrene 5.18 (50% probability ellipsoids). The hydrogen atoms have been omitted for clarity. Important bond lengths (Å) and angles (\*) P(1)-N(2) 1.667(6), P(1)-C(1) 1.905(8), P(1)-C(13) 1.822(8), N(1)-C(1) 1.305(9), N(2)-P(1)-C(1) 105.6(3), N(2)-P(1)-C(13) 105.0(3), C(1)-P(1)-C(13) 88.8(3),  $\Sigma$  angles at N(1) 359.9(6),  $\Sigma$  angles at N(2) 359.2(5),  $\Sigma$  angles at C(1) 359.8(6).



Figure 5.5: Proposed mechanism for the isomerization of 5.17a.

.

.

#### Chapter 6 Bis-Arenes as Four Site Donors to Antimony Trichloride

A general feature of arene complexes of the heavier *p*-block elements is structural diversity. Variations include the hapticity of the metal---arene interaction, and the metal:arene ratio (*i. e.*, sandwich, half sandwich, and inverse sandwich structures). The limited data on *p*-element---arene complexes indicate that many elements prefer the inverse sandwich arrangement with an arene donor sandwiched by two acceptors.<sup>174</sup> Despite an extensive study by Schmidbaur,<sup>175</sup> it is still not clear which factors govern the structure and stoichiometry of these complexes. In addition, the *p*-element---arene interaction is weak because of the lack of  $\sigma$  donor/ $\pi$  acceptor synergism, generally associated with transition metal complexes. For the *p*-elements, *d* core electrons are not available for back-bonding.<sup>176</sup>

One of the most extensively studied series of main group element arene complexes are the arene complexes of the antimony trihalides (Menshutkin complexes),<sup>177</sup> but this class is often extended to include all arene complexes of EX<sub>3</sub> (E=As, Sb, Bi, X=Cl, Br, I). "Half sandwich" and "inverse sandwich" <u>6.1</u> complexes of neutral aromatic hydrocarbons and SbX<sub>3</sub> (X=Cl, Br, I), are readily prepared.



Most of the arene ligands examined to date have been monoarenes (*i.e.*, single arene donors such as benzene<sup>178</sup> and various alkylated benzenes).<sup>179,180</sup> There appears to be no obvious reason for the 1:2 or 1:1 stoichiometries that are observed. Poly-arenes

which have been examined include naphthalene,<sup>181</sup> phenanthrene,<sup>182</sup> and pyrene  $(SbBr_3)$ .<sup>183</sup> A few examples are known with bifunctional arene donors that do not contain fused rings, including biphenyl,<sup>184</sup> 9,10-dihydroanthracene  $(SbBr_3)$ ,<sup>185</sup> and diphenylamine;<sup>186</sup> however, maximum coordinating ability of the arenes in such species has not been realized (*i. e.*, not all arene rings of the poly-arene engage two SbCl<sub>3</sub> molecules). For example, biphenyl forms 2 SbCl<sub>3</sub>:biphenyl despite potential for engaging four molecules of SbCl<sub>3</sub>.

## 6.1 Preparation of 4 SbCl<sub>3</sub>:Bis-Arene Complexes

Crystalline adducts of SbCl<sub>3</sub> with the general form 4 SbCl<sub>3</sub>:bis-arene are readily prepared with bis-arenes diphenylacetylene <u>6.2</u>, t-stilbene and bibenzyl. Direct



<u>6.2</u>

combination of the bis-arene with four equivalents of  $SbCl_3$  in  $CH_2Cl_2$  solution produces the adducts in high yield as analytically pure compounds.

A macroscopic characteristic of the 4 SbCl<sub>3</sub>:bis-arene complexes is their beautiful

Table 6.1: Comparison of melting points (°C) for 4 SbCl <sub>3</sub> :bis-arene complexes. N.B: mp. SbCl <sub>3</sub> 73 °C.		
Arene	Complex	Pure Bis-arene
diphenylacetylene	76-78	59-61
t-stilbene	101-103	122-124
bibenzyl	78-79	50-53

crystalline form (clear plates, some as large as 10 mm x 10 mm x 2 mm) which are readily soluble in  $CH_2Cl_2$  and significantly less soluble in hexane. The complexes acquire a light green color upon exposure to moist air; however, they are stable when stored under dry nitrogen. The complexes have been characterized by melting points (Table 6.1), which are sharp and distinctly different from the original starting materials. As pure compounds, both SbCl<sub>3</sub> and the bis-arenes are volatile and are conveniently purified by vacuum sublimation. The 4 SbCl<sub>3</sub>:bis-arene complexes remain unchanged after evacuation (ca. 10<sup>-3</sup> Torr) for 24 hours as shown by elemental analysis.

One of these complexes (bibenzyl) was originally described in an abstract;<sup>187</sup> however, this report was brief, consisting only of cell parameters and the space group of the complex. Full description of the sample preparation and an interpretation of the crystallographic data has not been published. This report also mentions a 2 SbCl<sub>3</sub>:bis-arene complexes (*t*-stilbene and bibenzyl) as well as reference to the "instability" of a 4:1 bibenzyl complex.



Figure 6.1: ORTEP view of the 4 SbCl<sub>3</sub>:diphenylacetylene complex (50% probability ellipsoids). The hydrogen atoms on the arene rings have been omitted for clarity.

# 6.2 Solid State and Spectroscopic Features of 4 SbCl<sub>3</sub>:Bis-Arene Complexes

IR spectra are consistent with the complexes containing monosubstituted arenes, with two strong bands corresponding to C-H out-of-plane vibrations between 770-730 and

Table 6.2: IR data (cm <sup>-1</sup> ) for bis-arenes and 4 SbCl <sub>3</sub> :bis-arene complexes.					
diphenyl acetylene	complex	t-stilbene	complex	bibenzyl	complex
1953s	1974w	1950w	1966w	1945w	1974w
1886s	1913w	1875w	1899w	1870w	1894w
1830s	1854w	1818w	1854w	1811w	1840w
1763s	1798w	1751w	1775w	1750w	1761w
756s	780s	763s	779s	758w	785s
689s	692s	688s	704s	699w	720s
		····			

720-680 cm<sup>-1</sup>. More importantly, these peaks are shifted in the complexes compared to those bands in the pure solid arenes (Table 6.2). Significant

shifts of combination and overtone bands of the C-H out-of-plane deformations are also observed.

No significant differences were observed between the <sup>13</sup>C and <sup>1</sup>H NMR spectra of the complexes and the pure arenes, in accord with earlier studies on the SbCl<sub>3</sub>:hexaethylbenzene 1:1 complex.<sup>180</sup> This complex does not show significant changes for the

<u>Table 6.3:</u> Comparison of Sb-aren (Å) for 4:1 SbCl <sub>3</sub> :arene complexe the arene.	ne and SbC <sub>arene</sub> dista s. Sb-arene measured	nces (minmax.) to the centroid of
Compound	· Sb-centroid (Å)	SbCarene (Å)
2 SbCl <sub>3</sub> :Benzene	3.22, 3.30	
4 SbCl <sub>3</sub> :bibenzyl	3.21(1), 3.36(1)	3.36(3)-3.79(3)
4 SbCl <sub>3</sub> :t-stilbene	3.24(1), 3.35(1)	3.38(3)-3.80(3)
4 SbCl3: diphenylacetylene	3.284(3), 3.385(3)	3.418(8)-3.840(8)

NMR of the arene; however, it exhibits rapid exchange with excess hexaethylbenzene. Crystals of the complexes were

obtained by slow solvent removal from  $CH_2Cl_2$ /hexane solution. The three structures belong to the same space group (P1), and are isotypical. Views of the diphenylacetylene

and *t*-stilbene complexes are presented in Figures 6.1-6.2. In all cases, the molecular structures of the 4:1 SbCl<sub>3</sub>:arene complexes consist of corrugated layers of SbCl<sub>3</sub>, with

Table 6.4: Sb-Cl bond lengths	-Cl bond lengths and SbCl contact distances (Å) for SbCl3:arene complexes.			
Compound	Sb(1)-Cl	Sb(1)Cl	Sb(2)-Cl	Sb(2)Cl
2 SbCl <sub>3</sub> :Benzene	2.353(2)	3.449(2)	2.357(2)	3,502(2)
	2.347(2)	3.401(2)	2.340(2)	3.802(2)
	2.380(2)	3.968(2)	2.378(2)	4.047(2)
4 SbCl <sub>3</sub> :Bibenzyl	2.311(8)	3.499(8)	2.335(7)	3.408(7)
	2.359(6)	3.804(7)	2.352(9)	3.661(8)
	2.381(9)	3.933(7)	2.361(8)	4.002(7)
4 SbCl <sub>3</sub> :t-Stilbene	2.347(7)	3.588(8)	2.343(7)	3.441(7)
-	2.360(7)	3.799(8)	2.355(7)	3.572(7)
	2.367(8)	3.878(8)	2.371(7)	3.923(9)
4 SbCl <sub>3</sub> :Diphenylacetylene	2.345(2)	3.401(2)	2.334(2)	3.622(3)
	2.352(2)	3.519(3)	2.358(2)	3.788(3)
	2.378(2)	3.862(3)	2.372(2)	3.813(3)
SbCl <sub>3</sub>	2.304(2)	2 x 3.457(1)		
-	2 x 2.368(1)	3.609(2)		
		2 x 3.736(1)		

each molecule of SbCl<sub>3</sub> coordinated to one ring of the bis-arene. The result is a sandwich with alternating organic and inorganic layers.

The Sb---arene distance as well as the range of Sb---C<sub>arene</sub> contacts are presented in Table 6.3. An interesting feature is the trend in Sb---arene distance with changes in the arene donor. The bibenzyl complex exhibits the shortest interaction with SbCl<sub>3</sub>, whereas the diphenylacetylene adduct engages in the longest coordination. A possible explanation may be the simple concept that  $sp^3$  hybridized alkyl substituents are electron donating and enhance the interaction with SbCl<sub>3</sub>. In contrast, *sp* hybridized carbon substituents (alkynes) are electron withdrawing. It should be noted, that the effect on the Sb---arene distance is small.



Figure 6.2: ORTEP view of the 4:1 SbCl<sub>3</sub>:*t*-stilbene complex (50% probability ellipsoids). The hydrogen atoms on the arene ring have been omitted for clarity.



Figure 6.3: PLUTO view of SbCl<sub>3</sub> illustrating the local coordination environment for antimony. Antimony atoms are shaded.

It is informative to compare the crystal structures of SbCl<sub>3</sub> <sup>188</sup> with those of the 4:1 SbCl<sub>3</sub>:bis-arene complexes. The structure of SbCl<sub>3</sub> (Figure 6.3) clearly shows that each antimony center achieves a coordination number of eight with three Sb-Cl bonds and five Sb---Cl contacts. Each SbCl<sub>3</sub> in the 4:1 SbCl<sub>3</sub>:bis-arene complexes exhibits a coordination number of seven, with the aromatic ring occupying the axial position of a distorted pentagonal bipyramid. Sb-Cl bond lengths and Sb---Cl contacts for the 4:1 SbCl<sub>3</sub>:bisarene complexes, as well as for 2SbCl<sub>3</sub>:benzene, are presented in Table 6.4. The organic layers of the complex adopt an arrangement in which all of the arene rings are coplanar and parallel.

## **6.3** Factors Governing the Structure of SbCl<sub>3</sub>:Arene Complexes

As noted earlier, the array of structures observed for the SbCl<sub>3</sub>-arene complexes, so far, has appeared to be unpredictable. The 4:1 complexes offer a new insight into factors

Table 6.5: Con 2 SbCl <sub>3</sub> :benze	nparison of crystallogra ne and 4 SbCl <sub>3</sub> : <i>t</i> -stilbe	phic data for ne.
Compound	2 SbCl <sub>3</sub> :benzene	4 SbCl <sub>3</sub> :t-stilbene
System	triclinic	triclinic
Space group	P <u>1</u>	P1
Ζ	2	2
a/Å	8.211(1)	8.363(3)
b/Å	11.83(2)	11.950(3)
c/Å	8.165(1)	8.099(3)
α/*	94.00(1)	95.44(3)
₿ <b>/</b> *	108.55(1)	109.15(3)
γ/*	94.22(1)	96.28(3)
V/Å <sup>3</sup>	746.4(2)	752.6(4)

which govern the structure of these complexes. The protypical SbCl<sub>3</sub>:arene complex is the 2:1 SbCl<sub>3</sub>:benzene,<sup>189</sup> which is isomorphic to the 4:1 SbCl<sub>3</sub>:bisarene complexes (Table 6.5).<sup>178</sup> The structures consist of corrugated layers of antimony trichloride with alternating layers of arene molecules. Packing diagrams of the 2:1 SbCl<sub>3</sub>:benzene

complex<sup>190</sup> and the 4:1 SbCl<sub>3</sub>:*t*-stilbene complex are shown in Figure 6.4 and Figure 6.5, respectively.

The bis-arenes of the 4:1 SbCl<sub>3</sub>:bis-arene complexes (bis-arene= bibenzyl, tstilbene, and diphenylacetylene) exhibit maximum donating ability (i. e., each arene ring coordinates to a molecule of SbCl<sub>3</sub>) and this is in contrast to other examples of SbCl<sub>3</sub>:bis-arene complexes. Structural studies of SbCl<sub>3</sub> complexes of naphthalene,<sup>181</sup> phenanthrene,<sup>182</sup> biphenyl,<sup>184</sup> and diphenylamine<sup>186</sup> reveal that only two molecules of SbCl<sub>3</sub> are engaged by the bis-arene despite potential for four-fold coordination. Alternating inorganic and organic layers are evident in the structures of these complexes, with SbCl<sub>3</sub> involved in extensive Sb---Cl interactions as seen in the 2 SbCl<sub>3</sub>:benzene complex. The longest arene-arene tether for these complexes is one atom long and the distance between adjacent arene rings is too short to incorporate another molecule of SbCl<sub>3</sub> without disturbing the SbCl<sub>3</sub> layer. Therefore, the maximum donating ability of the arene cannot be achieved. The complexes of the bis-arenes diphenylacetylene, t-stilbene and bibenzyl contain a two carbon tether between each arene ring. This distance between arene rings is sufficient to allow four molecules of SbCl<sub>3</sub> to interact with the bis-arene while maintaining the Sb---Cl contacts evident in solid SbCl<sub>3</sub>. Attempts to prepare a 4:1 SbCl<sub>3</sub>:bis-arene complex with diphenylbutadiyne were unsuccessful, and it is possible that the four carbon tether may be too long to facilitate formation of poly-SbCl<sub>3</sub> layers.

SbCl<sub>3</sub> forms a variety of coordination complexes with alkylated benzenes, and it is again likely that the establishment of Sb---Cl contacts is important. Formation of SbCl<sub>3</sub> chains and layers with six and seven coordinate antimony centers are common in these complexes. In cases of extreme steric crowding where formation of poly-SbCl<sub>3</sub> is prevented, alternate modes of coordination are adopted to maintain the key Sb---Cl contacts. For example, hexamethylbenzene forms a complex with 2 SbCl<sub>3</sub> molecules, but the solid state structure does not reveal layers of SbCl<sub>3</sub> molecules as seen in the case of 4:1 SbCl<sub>3</sub>:bis-arene complexes.<sup>191</sup> Rather, tetrameric units are formed, perhaps to maintain high coordination at the antimony center. Hexaethylbenzene, which is substantially more sterically crowded, forms a 1:1 complex.<sup>180</sup> The antimony center is located directly above the centroid of the arene, with a centroid---SbCl<sub>3</sub> contact 2.96 Å and no evident Sb---Cl contacts. This distance is substantially shorter than those observed in other complexes, such as 4:1 SbCl<sub>3</sub>:*t*-stilbene (3.255(1) and 3.315(1) Å) and 2:1 SbCl<sub>3</sub>:hexamethylbenzene (3.15 and 3.13 Å).<sup>191</sup> Without the Sb---Cl interactions which result upon formation of inorganic layers, an isolated SbCl<sub>3</sub> could be significantly more acidic than an SbCl<sub>3</sub> antimony center engaged with other SbCl<sub>3</sub> molecules. This could enhance the Sb---arene interaction. The single strong interaction with one SbCl<sub>3</sub> may prevent coordination of the arene by a second molecule of SbCl<sub>3</sub>. It is interesting to note that 2 AsBr<sub>3</sub>:hexaethylbenzene has been isolated, indicating significant differences between the coordination complexes of arsenic and antimony trihalides.<sup>192</sup>

ł

1



-

.

:

Figure 6.4: Packing diagram of the 2:1 SbCl<sub>3</sub>:benzene complex illustrating the alternating layers of SbCl<sub>3</sub> and arenes.



Figure 6.5: Packing diagram of the 4:1 SbCl<sub>3</sub>:*t*-stilbene complex illustrating the alternating layers of SbCl<sub>3</sub> and arenes.

#### Chapter 7 Future Work

13

1

The full extent of the coordination chemistry of main group element cations has not been realized, and future work should focus on the examination of common ligands, such as PR<sub>3</sub>, NR<sub>3</sub>, R<sub>2</sub>CO, etc., as potential donors to these cations. Bidentate ligands should also be examined. These reactions will not be described; however, more exciting and perhaps more synthetically challenging investigations will be proposed.

In addition to synthetic studies, calculations of strain energies should continue for other systems containing main group elements in low coordinate environments, such as the Lewis base adducts of  $R_3E$  (E = B, Al, Ga, and In).

## **<u>7.1</u>** Novel Ligands with Interesting Cations

σ-Complexes of the main group element cations are well established with donors such as amines and phosphines (*vide supra*). Further investigations into the preparation and reactivity of these complexes are in progress.<sup>193</sup> There are other, perhaps more interesting, ligands available. Possible novel ligands include CO, alkynes, nucleophilic



carbenes 7.1 and silvlenes 7.2. The rich potential of the stable carbenes is perhaps best illustrated by the recent characterization of 7.3.

# 7.2 Synthesis of New Phosphorus Cations

The quantitative insertion of [MesNP]<sup>+</sup> into E-H bonds provides a unique approach to phosphorus cations which the previously eluded synthesis. Particularly promising are reactions of [Mes\*NP]<sup>+</sup> with Mes\*SH, R<sub>3</sub>SnH, and R<sub>2</sub>BH to form novel phosphorus cations <u>7.4</u>, <u>7.5</u>, and <u>7.6</u>.



## 7.3 Nitrenium Cations

The diazonium cation [Mes\*NN]<sup>+</sup> should be prepared and characterized. This key molecule could yield stable  $\pi$  complexes analogous to those described in Chapter 4 for the



phosphoazonium cation. Isolation of the nitrenium cation 7.7 should then be realized using the 1, 3 hydrogen shift reaction.

## 7.4 Polymers Based on the SbCl<sub>3</sub>---Arene Interaction

Preparation of polymers containing transition metals in the backbone are being examined by a number of research groups;<sup>194</sup> however, analogous p-block



7.8

metalloid---arene interactions have not yet been exploited in terms of their potential for the preparation of such novel oligomers and polymers. Recognizing the ease with which SbCl<sub>3</sub> forms arene complexes, <sup>1</sup> eliminary investigations should be attempted to examine the potential of polymer preparation using the Sb---arene interaction.

The polymer  $\underline{7.8}$  is no different than any other polymer. Nylon 6,6 is a copolymer of hexanedioic acid (bifunctional acid) and 1,6-hexanediamine (bifunctional base) with the resulting polymer formed by amide bonds. The requirements for the polymer using the Cl<sub>3</sub>Sb---arene interaction are analogously simple. Unfortunately, bifunctional acids Cl<sub>2</sub>Sb-(R)-SbCl<sub>2</sub> are not readily available.

#### **Chapter 8** Experimental

## **<u>8.1</u>** Handling Air Sensitive Reagents

Cationic main group elements in low coordination environments are extremely reactive, and special apparatus and techniques are required to handle these air and moisture sensitive compounds. Over the past 8 years, our lab has developed a convenient system for handling such sensitive compounds. The reactors which are used to prepare the compounds are referred to as bridges, and their design and use has been described.<sup>195</sup> All reactions, with the exception of the preparation of Mes\*NPC1, were performed in three bulb bridges equipped with two reaction bulbs and an NMR tube.

Before use, the empty bridge was placed under dynamic vacuum for a minimum of 1 hour (ca.  $10^{-3}$  Torr) and was flame dried with a Bunsen burner for 10 minutes. It was then allowed to slowly cool to room temperature. Solid starting materials were prepared for use by exposure to dynamic vacuum ( $10^{-3}$  Torr) for a minimum period of 30 minutes. Their purity was assessed by melting point and/or NMR spectroscopy. Solid reagents were introduced into the bridge inside a nitrogen-filled drybox. Liquid starting materials were distilled before use and degassed using the standard freeze-pump-thaw technique. Liquid reagents with low volatility were introduced into the bridge inside a polyethylene glove bag, which had been flushed three times with dry nitrogen. Volatile reagents and solvents were vacuum distilled into the bridge on the vacuum line.

## **<u>8.2</u>** General Procedures

Phosphorus trichloride (Aldrich) and arsenic trichloride (BDH) were vacuum distilled and stored in evacuated bulbs. Triethylamine was distilled from CaH<sub>2</sub> and degassed. Trimethylsilylbromide and trimethylsilyliodide were vacuum distilled. Iodobenzene (Aldrich) and bromine (Fisher) were distilled from  $P_2O_5$ . Gallium trichloride, antimony trichloride (both Aldrich), and iodine (BDH) were sublimed under vacuum.

Chlorine (Matheson) was used as supplied. Sodium tetrafluoroborate, 2,4,6-tri-tertbutylaniline, 2.4.6-tri-tert-butylphenol, triphenylphosphine, 2.6-dimethylphenol, diphenylamine, pentachlorothiophenol, trifluoromethanesulphonic acid, diphenylacetylene, fluorene, bibenzyl, triphenylcarbenium tetrafluoroborate, triphenylcarbenium hexafluorophosphate, and stilbene (Aldrich) were used as supplied. Dichloromethane was dried over P<sub>2</sub>O<sub>5</sub> and CaH<sub>2</sub>. Toluene, benzene, tetrahydrofuran, and ether were dried over N<sub>1</sub>/benzophenone. Pentane and hexane were dried over CaH<sub>2</sub> and degassed. All solvents were stored in evacuated bulbs. Deuterated solvents were dried over P<sub>2</sub>O<sub>5</sub> or CaH<sub>2</sub>. Mes\*NPCl was prepared using a variation of the literature procedure (vide infra).<sup>31</sup> C<sub>6</sub>H<sub>5</sub>ICl<sub>2</sub>,<sup>196</sup> Mes\*NPBr,<sup>31</sup> and Mes\*NPI <sup>31</sup> were prepared following literature procedures. All solids were manipulated in Vacuum/Atmospheres or MBraun dryboxes containing an atmosphere of 99.998% prepurified nitrogen, which was constantly circulated through water and oxygen scavenging trains. Reactions were performed in evacuated bridges unless otherwise indicated. Melting points were recorded on a Fisher-Johns apparatus and are uncorrected. Elemental analyses were performed by Beller Laboratories, Göttingen, Germany. IR spectra were recorded as Nujol mulls on CsI plates using a Nicolet 510P FT-IR spectrometer. Solution NMR spectra were recorded on either a Bruker AC250 or Bruker AMX400 spectrometer in 5 mm evacuated flame sealed Pyrex tubes. Chemical shifts are reported in ppm relative to TMS (<sup>1</sup>H and <sup>13</sup>C), <sup>15</sup>NH<sub>3</sub> (<sup>15</sup>N) or 85% H<sub>3</sub>PO<sub>4</sub> (<sup>31</sup>P{<sup>1</sup>H}). <sup>1</sup>H and <sup>13</sup>C spectra were calibrated to the internal solvent signal. Variable temperature NMR studies involved equilibrium times of ten minutes at each temperature. Solid state <sup>13</sup>C and <sup>31</sup>P CP MAS NMR spectra were recorded on a Bruker AMX 400 spectrometer in 3 mm zirconium oxide spinners. Spinning rates are given in parentheses after the isotropic chemical shift. Crystals suitable for crystallography were obtained as described for each compound and mounted in Pyrex capillaries in a drybox. Structures were solved by Dr. T. Stanley Cameron, Dr. S. Sereda, P. Bakshi,

K. Robertson, or Dr. John F. Richardson. X-ray crystallographic data were collected on Rigaku AFC5R (TSC) or Enraf-Nonius CAD-4 (JFR) diffractometers with graphitemonochromated Mo K<sub> $\alpha$ </sub> radiation ( $\lambda = 0.71069$  Å or  $\lambda = 0.70926$  Å, respectively). Full details on the data collection and structure solution procedures used are available from the references as indicated in the appropriate discussion section for each compound. Part of the work described below was performed with students as part of their honours degree. Acknowledgement to these people is made where appropriate.

**Cautionary Note:** Many phosphorus compounds are <u>toxic</u> and possibly carcinogenic. Alkyllithium reagents are <u>pyrophoric</u>. Handle with care.

## **8.3** Specific Procedures

**Preparation of 2,4,6-Tri-***tert***-butylphenyliminophosphine chloride:** In a bridge equipped with a porous frit, PCl<sub>3</sub> (17.0 g, 124 mmol) was condensed over a solution of Et<sub>3</sub>N (15.0 g, 149 mmol), pentane ( $\approx$  50 mL), and 2,4,6-tri-*tert*-butylaniline (3.60 g, 13.8 mmol). The reaction mixture was stirred for one week and the solution became bright red with copious pecipitation of Et<sub>3</sub>NHCl. The precipitate was filtered and washed several times by repeated back distillations of the solvent mixture. Volatiles were removed *in vacuo* and the red solid was sublimed twice under dynamic vacuum ( $\approx$  10<sup>-3</sup> Torr) at 60-70°C. The red crystalline solid was characterized as 2,4,6-tri-*tert*-butylphenyliminophosphine chloride, Mes\*NPCl (2.90 g, 8.90 mmol, 65%), mp. 93-93.5°C.

IR: 1601s, 1271s, 1246s, 1246s, 1221s, 1200m, 1137m, 1060w, 927m, 879s, 763s, 651w, 497w, 410vs

NMR ( $CH_2Cl_2$ ): <sup>31</sup>P{<sup>1</sup>H}, 131.

Under similar experimental conditions, AsCl<sub>3</sub> was used in place of PCl<sub>3</sub> in an attempt to

1.

prepare the arsenic analogue, Mes\*NAsCl. These reaction conditions yielded a dark precipitate which was not characterized.

**Preparation of Mes\*NPCl<sub>3</sub>:** A solution of Mes\*NPCl (0.33 g, 1.0 mmol) in CH<sub>2</sub>Cl<sub>2</sub> ( $\approx$  10 mL) was added to a slurry of PhICl<sub>2</sub> (0.32 g, 1.2 mmol) in CH<sub>2</sub>Cl<sub>2</sub> ( $\approx$  10 mL) over a period of 30 minutes. The solvent was removed *in vacuo* and the residue was recrystallized twice from hexane to give yellow crystals characterized as Mes\*NPCl<sub>3</sub> (0.20 g, 0.50 mmol, 50%), mp. 148-149°C.

Elemental Anal. Calcd: C, 54.49; H, 7.37; N, 3.53; Cl, 26.81%. Found: C, 54.39; H, 7.24; N, 3.49; Cl, 26.70%.

IR: 1605m, 1530s, 971w, 922w, 888w, 879m, 768m, 658m, 648m, 530s, 448m.

- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H}, -96 ppm; <sup>1</sup>H, 1.30 (s, 9H), 1.46 (s, 18H), 7.30 (d  ${}^{5}J_{PH} = 3.0$ Hz, 2H); <sup>13</sup>C, 31.8 (s), 32.4 (s), 32.9 (s), 35.7 (s), 122.3 (s), 124.1 (s), 142.4 (d, J<sub>PC</sub> = 7.6 Hz), 145 (d J<sub>PC</sub> = 3.8 Hz); <sup>31</sup>P CP-MAS -103 small impurity at -73 (4000, 7000).
- Reaction Mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectra of the reaction mixtures contained a single signal at -96 ppm.
- X-ray: monoclinic, space group *Cm*, a = 11.960(3) Å, b = 14.696(2) Å, c = 5.954(2) Å,  $\beta = 97.71(2)$  Å, V = 1037.1 Å<sup>3</sup>, Z = 2,  $D_{calcd} = 1.27$  g cm<sup>-3</sup>,  $\mu = 5.2$  cm<sup>-1</sup>,  $100R_w = 4.7$  (JFR).

**Preparation of Mes\*NPBr<sub>3</sub>:** Bromine (0.15 g, 0.93 mmol) in hexane ( $\approx 5$  mL) was added to a solution of Mes\*NPBr (0.33 g, 0.91 mmol) in hexane ( $\approx 5$  mL). After stirring at room temperature for 5 minutes, the solvent was removed *in vacuo* and the solids were recrystallized from a minimum amount of warm hexane giving orange/red crystals. These crystals were characterized as Mes\*NPBr<sub>3</sub> (0.34 g, 71%), mp. 161.5-163 °C.

Elemental Anal. Calcd: C, 40.78; H, 5.51; N, 2.64 %. Found: C, 41.07; H, 5.72; N, 2.57 %.

IR: 1601m, 1504s, 923w, 881w, 513s, 487m, 446s, 425m.

- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H}, -236; <sup>1</sup>H, 1.30 (s, 9H), 1.46 (s, 18H), 7.30 (d, <sup>5</sup>J<sub>PH</sub> = 3.0 Hz, 2H); <sup>13</sup>C, 31.8 s, 32.4 s, 32.9 s, 35.7 s, 122.3 s, 124.1 s, 142.4 (d, J<sub>PC</sub> = 7.6 Hz), 145.0 (d, J<sub>PC</sub> = 3.8 Hz).
- Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectra of the reaction mixture showed a single signal at -236 ppm.

X-ray: monoclinic, space group  $P2_I/c$ , a = 18.841(8) Å, b = 9.480(4) Å, c = 12.186(5) Å,  $\beta = 98.00(3)$  Å, V = 2155.3 Å<sup>3</sup>, Z = 4,  $D_{calcd} = 1.63$  g cm<sup>-3</sup>,  $\mu = 56.4$  cm<sup>-1</sup>,  $100R_w = 7.4$  (JFR).

**Reaction of Mes\*NPC1 with Br<sub>2</sub>:** Bromine (0.32 g, 2.0 mmol) in pentane (~ 15 mL) was added to a solution of Mes\*NPC1 (0.52 g, 1.6 mmol) in pentane (~ 15 mL). After stirring at room temperature for 12 hours 50% of the solvent was removed *in vacuo* and the solid that formed was warmed back into solution. Slow cooling to room temperature resulted in the formation of orange/red crystals; which were characterized as a mixture of Mes\*NPBr<sub>3</sub>, Mes\*NPBr<sub>2</sub>Cl and Mes\*NPBrCl<sub>2</sub> (total recovered mass 0.62 g) mp. range 172-184°C. A single crystal of Mes\*NPBr<sub>3</sub> was isolated from this mixture and characterized by X-ray crystallography (*vide supra*). Calculated composition based on the integrated ratios observed in the <sup>31</sup>P{<sup>1</sup>H} NMR spectrum of the solids: -236 (40) Mes\*NPBr<sub>3</sub>, -183 (50) Mes\*NPBr<sub>2</sub>Cl, -136 (10) Mes\*NPBrCl<sub>2</sub>. (Derek P. Gates) Elemental Anal. Calcd: C, 43.3; H, 5.9 %. Found: C, 43.2; H, 5.7 %.

Reaction Mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectra of the reaction mixture indicated formation of five products: -236 (40) Mes\*NPBr<sub>3</sub>, -213 (minor), -182 (40) Mes\*NPBr<sub>2</sub>Cl,

-162 (minor), -136 (10) Mes\*NPBrCl<sub>2</sub>.

Reaction of Mes\*NPCl with I<sub>2</sub>: A slurry of I<sub>2</sub> (0.28 g, 1.1 mmol) in pentane ( $\approx$  30 mL) was added to a solution of Mes\*NPCl (0.70 g, 2.2 mmol) in pentane ( $\approx$  20 mL). After stirring overnight at room temperature, slow removal of the solvent *in vacuo* resulted in the precipitation of orange/red needle crystals which were isolated in very low yield and characterized as Mes\*NPCl<sub>2</sub>I, mp. 124-130°C (sublimes). (Derek P. Gates) Elemental Anal. Calcd: C, 44.28; H, 5.99; N, 2.87 %.

Found: C, 44.62, H, 5.86, N, 2.74 %.

IR: 1598s, 1476s, 924w, 880m, 527s, 507m, 480s, 431m.

- NMR:  $(CD_2Cl_2)$ : <sup>31</sup>P{<sup>1</sup>H}, -211; <sup>1</sup>H, 1.30 (s, 9H), 1.46 (s, 18H), 7.30 (d <sup>5</sup>J<sub>PH</sub> = 3.0 Hz, 2H); <sup>13</sup>C, 31.8 (s), 32.4 (s), 32.9 (s), 35.7 (s), 122.3 (s), 124.1 (s), 142.4 (d, J<sub>PC</sub> = 7.6 Hz), 145 (d J<sub>PC</sub> = 3.8 Hz).
- Reaction mixture: -211 Mes<sup>\*</sup>NPCl<sub>2</sub>I, 200-160 Mes<sup>\*</sup>NPCl and Mes<sup>\*</sup>NPI broad ( $\delta$  varies depending upon the stoichiometry. With excess I<sub>2</sub>,  $\delta$  = 200, with trace I<sub>2</sub>,  $\delta$  = 164, average signal due to exchange.
- X-ray: monoclinic, space group  $P2_{l}/n$ , a = 9.903(3) Å, b = 11.638(3) Å, c = 19.734(5)Å,  $\beta = 103.34(2)$  Å, V = 2213.0 Å<sup>3</sup>, Z = 4,  $D_{calcd} = 1.47$  g cm<sup>-3</sup>,  $\mu = 17.5$  cm<sup>-1</sup>,  $100R_{w} = 10.2$  (JFR).

NMR Studies: The reactions listed below were examined by combining the reactants in equimolar amounts in pentanes and obtaining a  ${}^{31}P{}^{1}H$  NMR spectrum of the reaction mixture. The results are presented as the chemical shifts (ppm) of the observed signals with their relative integration given as percentages, in parentheses, followed by the assignment for each signal.

- Mes\*NPBr with PhICl<sub>2</sub>: -96 (10) Mes\*NPCl<sub>3</sub>, -135 (70) Mes\*NPBrCl<sub>2</sub>, -183 (20) Mes\*NPBr<sub>2</sub>Cl.
- Mes\*NPBr with I<sub>2</sub>: 177 (>90) Mes\*NPBr and Mes\*NPI average due to exchange.
- Mes\*NPI with PhICl<sub>2</sub>: 220 (1) PCl<sub>3</sub>, 158 (40) Mes\*NPCl and Mes\*NPI average due to exchange, -96 (10) Mes\*NPCl<sub>3</sub>, -211 (40) Mes\*NPCl<sub>2</sub>I.

Mes\*NPI with Br<sub>2</sub>: 227 (100) PBr<sub>3</sub>.

Mes\*NPI with I<sub>2</sub>: No reaction.

Mes\*NPBr with Mes\*NPI: 159 (100) average signal due to exchange.

Mes\*NPCl<sub>3</sub> with Mes\*NPBr<sub>3</sub>: No reaction.

Mes\*NPCl<sub>3</sub> with Ph<sub>3</sub>Sb: no reaction in CH<sub>2</sub>Cl<sub>2</sub>.

Mes\*NPCl<sub>3</sub> with SbF<sub>3</sub>: no reaction in CH<sub>2</sub>Cl<sub>2</sub>.

**Reaction of Mes\*NPCI with [Ph<sub>3</sub>C][BF<sub>4</sub>]:** Mes\*NPCI (0.41 g, 1.26 mmol) in  $\approx$  10 mL of 3:1 CH<sub>2</sub>Cl<sub>2</sub>/hexane was added over a period of 10 minutes to a solution of [Ph<sub>3</sub>C][BF<sub>4</sub>] (0.42 g, 1.28 mmol) in  $\approx$  10 mL of a similar solvent mixture. The solution turned bright yellow after 10 minutes and slow removal of solvent yielded yellow crystals which were characterized as Mes\*NPF<sub>2</sub>CPh<sub>3</sub> (0.56 g, 0.98 mmol, 78%), mp. 157.0-158.5 °C.

Elemental anal: Calcd: C, 77.73; H, 7.76; N, 2.45 %. Found C, 76.53; H, 8.34; N, 2.59 %.

- IR: 3055s, 3035s, 1530s, 1494s, 1295m, 1272m, 1247m, 1215m, 1193m, 1135m, 1083m, 1083m, 1035m, 945m, 922m, 915m, 899m, 879m, 854m, 828s, 818s, 775m, 769m, 762m, 741s, 697s, 643m, 624m, 521s, 487m, 457m.
- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): -38 ppm (triplet; <sup>1</sup>J<sub>PF</sub>: 1221 Hz), <sup>1</sup>H and <sup>13</sup>C NMR spectra were very complex and poorly resolved due to splitting with the PF<sub>2</sub> unit.

Reaction Mixture: <sup>31</sup>P{<sup>1</sup>H} NMR studies of [Ph<sub>3</sub>C][BF<sub>4</sub>] with Mes\*NPCl in

CH<sub>2</sub>Cl<sub>2</sub>/hexane solution indicate immediate quantitative formation of a signal at 146 ppm (triplet;  ${}^{1}J_{PF} = 1105$  Hz) which then decays over three days to give a signal at -38 ppm (triplet;  ${}^{1}J_{PF} = 1221$  Hz) in a solution yield of > 90%.

X-ray: triclinic, space group PL, z = 13.728(3) Å, b = 23.808(5) Å, c = 10.125(2) Å,  $\alpha = 95.69(1)^{\circ}$ ,  $\beta = 101.37(1)^{\circ}$ ,  $\gamma = 91.62(2)^{\circ}$ , V = 3224 Å<sup>3</sup>, Z = 2,  $D_{calcd} = 1.178$  gcm<sup>-3</sup>,  $\mu = 1.17$  cm<sup>-1</sup>,  $100R_w = 4.98$  (TSC).

**Preparation of [Mes\*NP-benzene][GaCl**<sub>4</sub>]: A solution of GaCl<sub>3</sub> (0.21 g, 1.2 mmol) in benzene ( $\approx 15$  mL) was added over a period of  $\approx 5$  minutes to a stirred solution of Mes\*NPCl (0.38 g, 1.2 mmol) in benzene ( $\approx 15$  mL) giving a bright yellow solution. Solvent was removed *in vacuo* overnight to yield a yellow powder which was characterized as [Mes\*NP][GaCl<sub>4</sub>]. Recrystallization from benzene through repeated cycles of warming to  $\approx 60$  °C and slow cooling to room temperature gave [Mes\*NPC<sub>6</sub>H<sub>6</sub>][GaCl<sub>4</sub>], dp.  $\approx 90$  °C. Elemental analysis was not obtained.

- IR: 1979w, 1842w, 1597s, 1267m, 1245m, 1134m, 1066m, 1033m, 927w, 886s, 764m, 699s, 674s, 393s, 361s, 263m.
- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H}, 76; <sup>1</sup>H, 7.47 (d, <sup>5</sup>J<sub>PH</sub> = 2 Hz, 2H), 1.56 (s 18H), 1.33 (s, 9H) [Mes\*], 7.38 (s, 6H) [benzene]; <sup>13</sup>C,123.8 (d, <sup>4</sup>J<sub>PC</sub> = 3 Hz), 30.9 (s), 30.1 ppm (d, <sup>5</sup>J<sub>PC</sub> = 2 Hz) [Mes\*], 128.7 ppm (s) [benzene], quaternary carbon nuclei not observed.
- Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectra of the reaction mixture showed a single signal at 71 ppm.

X-Ray: orthorhombic, space group *Pbca*, a = 24.948(4), b = 20.195(4), c = 11.953(3) Å, V = 6022(4) Å<sup>3</sup>, Z = 8,  $D_{calcd} = 1.279(1)$  gcm<sup>-3</sup>,  $\mu = 13.32$  cm<sup>-1</sup>,  $100R_w = 4.19$ (TSC).

Benzene is quantitatively removed from the crystals under dynamic vacuum for 24 hours to

give [Mes\*NP][GaCl<sub>4</sub>] as characterized by IR and elemental analysis; dp. 122-3.5 °C

- Elemental Anal. Calcd: C, 43.07; H, 5.82; N, 2.79 %. Found: C, 43.02; H, 5.79; N, 2.97 %.
- IR: 1594s, 928m, 885m, 842m, 763m, 693m, 650w, 474w, 464m, 401s, 373s, 360s, 261m.
- NMR:  $(CD_2Cl_2)$ : <sup>31</sup>P{<sup>1</sup>H}, 76; <sup>1</sup>H, 7.46 (d, <sup>5</sup>J<sub>PH</sub> = 2 Hz, 2H), 1.54 (s, 18H), 1.31 (s, 9H); <sup>31</sup>P CP-MAS NMR, 63 ppm (4000, 7000 Hz).

**Preparation of [Mes\*NP-toluene][GaCl4]:** A solution of GaCl<sub>3</sub> (0.08 g, 0.47 mmol) in toluene ( $\approx 2$  mL) was added over  $\approx 5$  minutes to a stirred solution of Mes\*NPCl (0.15 g, 0.47 mmol) in toluene ( $\approx 2$  mL) giving a yellow solution. Solvent was removed *in vacuo* to give a yellow powder which was recrystallized from a minimum amount of toluene to give [Mes\*NPC<sub>7</sub>H<sub>8</sub>][GaCl<sub>4</sub>], dp. 157-160°C. Elemental analysis was not obtained.

- IR: 1594w, 1267m, 1243w, 1133m, 886m, 762w, 722m, 699m, 674w, 400s, 368s, 266s.
- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H}, 76; <sup>1</sup>H, 7.36 (s, 2H), i.45 (s, 18H), 1.28 (s, 9H) [Mes\*],
  7.56 7.47 (multiplet, 5H), 2.48 (s, 3H) [toluene]; <sup>13</sup>C, 123.7 (d <sup>5</sup>J<sub>PC</sub> = 3 Hz),
  30.6, 29.4 [Mes\*], 132.4, 131.4, 128.1, 22.0 [toluene], quaternary carbon nuclei not observed.
- Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectra of the reaction mixture showed a single signal at 78 ppm.

X-ray: orthorhombic, space group *Pbca*, a = 25.006(9), b = 20.358(8), c = 12.126(5) Å, V = 6173(8) Å<sup>3</sup>, Z = 8,  $D_{calcd} = 1.278(2)$  gcm<sup>-3</sup>,  $\mu = 13.01$  cm<sup>-1</sup>,  $100R_w = 7.60$ (TSC).

Toluene is quantitatively removed from the crystals under dynamic vacuum for 24 hours to

give [Mes\*NP][GaCl<sub>4</sub>].

IR: 1596s, 928m, 887m, 842m, 764m, 694m, 464m, 402s, 373s, 362s.

NMR reactions involving Mes\*NPCl with 0.5 equivalents of GaCl<sub>3</sub> in toluene show a single <sup>31</sup>P{<sup>1</sup>H} shift of 112 ppm. NMR reactions involving Mes\*NPCl with 4 equivalents of GaCl<sub>3</sub> in toluene show a single <sup>31</sup>P{<sup>1</sup>H} shift of 92 ppm.

Preparation of [Mes\*NP-benzene] [Ga2Cl7]: A solution of GaCl3 (0.41 g, 2.3

mmol) in benzene ( $\approx 10 \text{ mL}$ ) was added over a period of  $\approx 5 \text{ minutes to a stirred solution}$ of Mes\*NPCl (0.38 g, 1.2 mmol) in benzene ( $\approx 10 \text{ mL}$ ). Upon standing for  $\approx 5 \text{ minutes a}$ small amount of yellow oil formed under a bright yellow solution. Solvent was removed *in vacuo* to yield a yellow powder which was recrystallized from benzene in hexane (75 % v/v) with repeated cycles of warming to  $\approx 60$  °C and slow cooling to room temperature to give [Mes\*NPC<sub>6</sub>H<sub>6</sub>][Ga<sub>2</sub>Cl<sub>7</sub>] (0.71 g, 0.95 mmol, 82%), mp. 78-81 °C.

Elemental Anal. Calcd: C, 38.12; H, 4.67; N, 1.85 %. Found: C, 37.97; H, 4.47; N, 1.89 %.

- IR: 1981w, 1849w, 1597m, 1268m, 1135m, 1068m, 1025w, 986w, 929w, 887m, 763w, 700s, 400s, 367s, 265s.
- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H}, 93; <sup>1</sup>H, 7.40 (s, 2H), 1.48 (s, 18H), 1.30 (s, 9H) [Mes\*],
  7.80 (s, 6H) [benzene]; <sup>13</sup>C, 124.0, 30.6, 29.3 [Mes\*], 132.6 [benzene],
  quaternary carbon nuclei not observed. <sup>31</sup>P{<sup>1</sup>H} CP-MAS NMR, 90 (3000, 7000 Hz). <sup>13</sup>C{<sup>1</sup>H} CP-MAS NMR 129.4 ppm [benzene].
- Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectra of the reaction mixture showed a single signal at 93 ppm.
- X-ray: triclinic, space group *P1*, *a* = 11.579(2), *b* = 16.350(3), *c* = 11.085(3) Å, *a* = 90.74(2),  $\beta = 118.00(2), \gamma = 73.53(2)^{\circ}, V = 1759(1) Å^3, Z = 2, D_{calcd} = 1.428(1) \text{ gcm}^{-3}, \mu = 21.23 \text{ cm}^{-1}, 100R_w = 3.86 \text{ (TSC)}.$

**Preparation of [Mes\*NP-toluene][Ga<sub>2</sub>Cl<sub>7</sub>]:** Mes\*NPCl (0.34 g, 1.1 mmol) in toluene ( $\approx$  30 mL) was added over a period of  $\approx$  5 minutes to a stirred solution of GaCl<sub>3</sub> (0.38 g, 2.1 mmol) in toluene ( $\approx$  30 mL). An orange oil formed on removal of the solvent *in vacuo*. Fresh toluene ( $\approx$  25 mL) was distilled onto this oil and yellow orange crystals formed upon standing overnight. Solvent was decanted from the crystalline material which was washed with hexanes and characterized as [Mes\*NPC<sub>7</sub>H<sub>8</sub>][Ga<sub>2</sub>Cl<sub>7</sub>] (0.65 g, 80%), mp. 87.5-89 °C.

- Elemental Anal. Calcd: C, 38.99; H, 4.84; N, 1.82 %. Found: C, 38.72; H, 5.01; N, 1.81 %.
- IR: 1971w, 1892w, 1842w, 1787w, 1594m, 1266s, 1244m, 1134m, 1067m, 887s, 764m, 419s, 401s, 362s, 263s.
- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H}, 95; <sup>1</sup>H, 7.42 (s, 2H), 1.50 (s, 18H), 1.31 (s, 9H) [Mes\*],
  7.76 7.66 (multiplet, 5H), 2.56 ppm (s, 3H) [toluene]; <sup>13</sup>C, 124.0, 30.6, 29.4
  [Mes\*], 134.2, 133.1, 130.0, 22.3 [toluene], quaternary carbon nuclei not observed. <sup>31</sup>PCP-MAS NMR 89 (3137, 4045 Hz).
- Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectra of the reaction mixture showed a single signal at 91 ppm.
- X-ray: triclinic, space group *P1*, *a* = 11.571(4), *b* = 16.335(5), *c* = 11.326(3) Å, *a* = 91.47(4)\*,  $\beta$  = 117.87(2)\*,  $\gamma$  = 105.34(3)°, *V* = 1797(1) Å<sup>3</sup>, *Z* = 2, *D<sub>calcd</sub>* = 1.423(1) g cm<sup>-3</sup>,  $\mu$  = 20.80 cm<sup>-1</sup>, 100*R<sub>w</sub>* = 4.64 (TSC).

**Preparation of [Mes\*NP-mesitylene][Ga<sub>2</sub>Cl<sub>7</sub>]:** Mes\*NPCl (0.35 g, 1.1 mmol) in mesitylene ( $\approx$  7 mL) was added to a solution of GaCl<sub>3</sub> (0.38 g, 2.2 mmol) in mesitylene ( $\approx$  3 mL) and a yellow oil formed under a clear yellow solution. Hexanes ( $\approx$  10 mL) was distilled onto this solution giving a light yellow precipitate, which was isolated by decantation and recrystallized from a warm solution of hexane/mesitylene (20/80). Two

crops of crystals were isolated and characterized as [Mes\*NP(mesitylene)][Ga<sub>2</sub>Cl<sub>7</sub>] (total yield 0.76 g, 0.97 mmol, 90 %). mp. 105-7 °C.

Elemental Anal. Calcd: C, 40.63; H, 5.18; N, 1.75 %. Found: C, 38.38; H, 5.54; N, 1.71 %.

- IR: 1813w, 1784w, 1759w, 1589m, 1302m, 1265m, 1243m, 1219m, 1194m, 1134m, 885m, 856m, 762m, 676m, 407s, 367m, 360m, 265m.
- NMR:  $(CD_2Cl_2)$ : <sup>31</sup>P{<sup>1</sup>H}, 91; <sup>1</sup>H, 7.41 (d, <sup>5</sup>J<sub>PH</sub> = 2 Hz, 2H), 1.50 (s, 18H), 1.30 (s, 9H) [Mes\*], 7.21 (s, 3H), 2.42 (s, 9H) [mesitylene]; <sup>13</sup>C, 123.8 (s), 31.0 (s), 30.3 (s) [Mes\*], 131.1 (s), 21.6 (s) [mesitylene] (Quaternary carbons not reported).
- Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectra of the reaction mixture showed a single signal at 91 ppm.
- X-ray: triclinic, space group *P1*, a = 12.119(3) Å, b = 16.575(4) Å, c = 11.309(3) Å,  $a = 93.22(2)^{\circ}$ ,  $\beta = 115.62(2)^{\circ}$ ,  $\gamma = 107.64(2)^{\circ}$ , V = 1905(1) Å<sup>3</sup>, Z = 2,  $D_{calcd} = 1.391(1)$  gcm<sup>-3</sup>,  $\mu = 19.64$  cm<sup>-1</sup>,  $100R_w = 4.50$  (TSC).

**Protonation Reaction of Mes\*NP-Ph:** A blue solution of Mes\*NP-Ph (0.61 mmol), prepared *in situ* by the reaction of Mes\*NPCl with NaBPh<sub>4</sub> in CH<sub>2</sub>Cl<sub>2</sub> ( $\approx 10$  mL) was added to a solution of HOSO<sub>2</sub>CF<sub>3</sub> (0.61 mmol) in CH<sub>2</sub>Cl<sub>2</sub> ( $\approx 10$  mL). A <sup>31</sup>P NMR spectrum of the yellow reaction mixture showed a single signal at 187 ppm assigned to [Mes\*N(H)P-Ph][SO<sub>3</sub>CF<sub>3</sub>].

**Preparation of Mes\*N(H)P=fluorenylidene:** A solution of fluorenyl lithium in ether (0 °C, 30 ml) was prepared by reacting fluorene (0.51 g, 3.07 mmol) with BuLi (1.9 ml, 1.6 M BuLi, 3.04 mmol) for 30 minutes and was added to a stirred solution of Mes\*NPCl (1.02 g, 3.14 mmol) in ether (0 °C). The solution was allowed to warm to room

temperature and the volatiles were removed *in vacuo* and the solids were recrystallized from hexane giving orange crystals characterized as Mes\*N(H)P=fluorenylidene (0.55 g, 1.22 mmol, 39 %), dp. 207-9 °C.

Elemental anal. Calcd: C, 81.72; H, 8.41; N, 3.07 %. Found: C, 81.54; H, 8.50; N, 3.31 %.

- IR: 3413m, 1597m, 1308s, 1268s, 1217s, 1115s, 1026m, 1004m, 920s, 904s, 881 s, 762s, 739s, 050m, 620m, 412m. IR spectrum unchanged after 20 minutes in moist air.
- NMR (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H} 256; <sup>1</sup>H 7.92-7.79 and 7.39-7.21 (8H, aromatic), 7.49 (s, 2H), 6.85 (d, 11 Hz, 1H), 1.34 (s, 18H), 1.36 (s, 9H); <sup>13</sup>C 127.3 (d, 3Hz), 126.8 (s), 126.6 (d, 5 Hz), 125.5 (d, 5 Hz), 123.8 (s), 123.2 (d, 4 Hz), 120.1 (s), 118.1 (d, 21 Hz), 33.54 (s), 23.48 (s), 31.4 (s) quaternary carbons not observed.

Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectrum of the reaction mixture indicated immediate (5 minutes) quantitative formation of Mes\*N(H)P=fluorenylidene in solution.

X-ray: monoclinic, space group  $P2_1/c$ , a = 9.586(8) Å, b = 24.25(2) Å, c = 11.77(1) Å,  $\beta$ = 101.38(8)\*, V = 2678(8) Å<sup>3</sup>, Z = 4,  $D_{calcd} = 1.13$  gcm<sup>-3</sup>,  $\mu = 1.161$  cm<sup>-1</sup>, 100 $R_w$ = 4.75 (TSC).

**Preparation of [Mes\*N(H)PN(H)Mes\*][GaCl<sub>4</sub>]:** A solution of Mes\*NH<sub>2</sub> (0.13 g, 0.51 mmol) in  $\approx$  20 mL of CH<sub>2</sub>Cl<sub>2</sub> was added over  $\approx$  20 minutes to a stirred solution of [Mes\*NP][GaCl<sub>4</sub>] (0.26 g, 0.51 mmol) in  $\approx$  25 mL of CH<sub>2</sub>Cl<sub>2</sub>. The solution slowly turned light yellow/orange, and with stirring turned light yellow in four hours. Solvent was removed *in vacuo*, and the solids recrystallized from CH<sub>2</sub>Cl<sub>2</sub> to give pale yellow crystals of crystallographic quality characterized as [Mes\*N(H)PN(H)Mes\*][GaCl<sub>4</sub>] (0.25 g, 0.32 mmol, 64 %), mp. 152-153.5 °C.

Elemental anal. Calcd: C, 56.94; H, 7.43; N, 3.69 %. Found: C, 56.79; H, 7.69; N, 3.57 %.

- IR: 3175sb, 1599m, 1419m, 1396m, 1319m, 1271m, 1242m, 1212s, 1178m, 1103s, 981s, 881s, 694m, 651w, 389s, 376s, 362s.
- NMR (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H} 272; <sup>1</sup>H 9.94 (d, <sup>2</sup>J<sub>PH</sub> = 14 Hz, 2H), 7.54 (s, 4H), 1.57 (s, 36H), 1.32 (s, 18H); <sup>13</sup>C 125.0 (s), 34.1 (s), 31.2 (s), 31.4 (s) quaternary carbons not observed. Solid state <sup>31</sup>P CP MAS 281 ppm (4000, 6000).
- Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectrum of the reaction mixture indicated immediate (5 min.) quantitative formation of [Mes\*N(H)PN(H)Mes\*][GaCl<sub>4</sub>] in solution.
- X-ray: monoclinic, space group C2/c, a = 24.931(2) Å, b = 10.198(4) Å, c = 16.444(2)Å,  $\beta = 93.32(1)^{\circ}$ , V = 4172(8) Å<sup>3</sup>, Z = 4,  $D_{calcd} = 1.215$  gcm<sup>-3</sup>,  $\mu = 9.769$  cm<sup>-1</sup>,  $100R_w = 3.27$  (TSC).

Reaction of [Mes\*NP][GaCl<sub>4</sub>] with Mes\*OH: A solution of Mes\*OH (0.19 g, 0.72 mmol) in  $\approx$  10 mL of CH<sub>2</sub>Cl<sub>2</sub> was added over  $\approx$  10 minutes to a stirred solution of [Mes\*NP][GaCl<sub>4</sub>] (0.36 g, 0.71 mmol) in  $\approx$  20 mL of CH<sub>2</sub>Cl<sub>2</sub>. The solution slowly turned light yellow. Solvent was removed *in vacuo*, and the solids recrystallized from CH<sub>2</sub>Cl<sub>2</sub> to give a bright yellow microcrystalline solid characterized as [Mes\*N(H)POMes\*][GaCl<sub>4</sub>] (0.45 g, 0.70 mmol, 98 %), mp. 160-163 °C. Elemental anal. Calcd: C, 56.57; H, 7.78; N, 1.83 %.

Found: C, 56.00; H, 7.75; N, 1.89 %.

- IR: 120sb, 1599m, 1418s, 1318m, 1305m, 1245m, 1219m, 1207m, 1165m, 1066s, 997s, 881s, 805w, 756m, 737m, 649w, 379s, 361s.
- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H} 296; <sup>1</sup>H 10.9 (broad, 1H), 7.59 (s, 2H), 7.48 (s, 2H), 1.60 (s, 18H), 1.58 (s, 18H), 1.40 (s, 9H), 1.32 (s, 9H); <sup>13</sup>C 125.6 (s), 125.3 (s), 34.4 (s), 33.7 (s), 31.3 (s), 31.2 (s) quaternary carbons not observed. Solid state

<sup>31</sup>P CP MAS 305.7, 302.7 ppm (5000, 8000).

Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectrum of the reaction mixture indicated immediate (5 minutes) quantitative formation of a species with a chemical shift of 296. On standing, a bright yellow powder precipitates.

Reaction of [Mes\*NP][GaCl4] with C<sub>6</sub>Cl<sub>6</sub>SH: A solution of C<sub>6</sub>Cl<sub>6</sub>SH (0.20 g,

0.71 mmol) in  $\approx 20$  mL of CH<sub>2</sub>Cl<sub>2</sub>was added over  $\approx 10$  minutes to a stirred solution of [Mes\*NP][GaCl<sub>4</sub>] (0.35 g, 0.71 mmol) in  $\approx 50$  mL of CH<sub>2</sub>Cl<sub>2</sub>. The solution turned bright yellow with copious precipitation of a light yellow solid which was extremely insoluble. Solvent was decanted and the solids washed with CH<sub>2</sub>Cl<sub>2</sub> to give a bright yellow powder (0.42 g) dec. > 225 °C.

IR: 1592m, 1334s, 1303s, 1210w, 1164w, 494s, 463s, 378s, 364s.

- NMR: Solid state <sup>31</sup>P CP MAS 366 (4000, 8000) weak signal. <sup>1</sup>H and <sup>13</sup>C not obtained because of the extreme insolubility of the compound.
- Reaction mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectrum of the reaction mixture showed quantitative formation of a species with a chemical shift of 360 ppm, but the peak was broad due to the formation of copious precipitate in the NMR tube.

NMR Studies: Reactions of Mes\*N=P-N(H)Mes\* with  $CH_3I$  in  $CH_2Cl_2$  and in neat solution showed only a single peak at 270 ppm [Mes\*N=P-N(H)Mes\*]. No reaction was observed between Mes\*N=P-N(H)Mes\* or Mes\*N(H)P -fluorenylidene with  $CCl_4$ . Mes\*N=P-N(H)Mes\* reacts slowly with [CPh<sub>3</sub>][GaCl<sub>4</sub>] to give a single species with a chemical shift of 15 ppm. There was no immediate reaction between Mes\*NPCl and [CPh<sub>3</sub>][GaCl<sub>4</sub>], but after one week a multitude of species were formed, from which a single crystal of [Mes\*N(H)PN(H)Mes\*][GaCl<sub>4</sub>] was isolated. Reaction mixtures containing [Mes\*NP][GaCl<sub>4</sub>] with Et<sub>2</sub>O, THF, and NEt<sub>3</sub> in CH<sub>2</sub>Cl<sub>2</sub> all exhibited quantitative formation of Mes\*NPCl (132 ppm). Reaction mixtures containing equimolar amounts of [Mes\*NP][GaCl<sub>4</sub>] and Ph<sub>3</sub>P in CH<sub>2</sub>Cl<sub>2</sub> exhibited quantitative formation of Mes\*NPCl (132 ppm) and Ph<sub>3</sub>P- $\rightarrow$ GaCl<sub>3</sub> (-6 ppm). [Mes\*NP][GaCl<sub>4</sub>] with 2,6dimethylphenol gave a single phosphorus containing species with a chemical shift of 27 ppm (soluble in hexanes). [Mes\*NP][GaCl<sub>4</sub>] with Ph<sub>2</sub>NH gave a single species with a <sup>31</sup>P NMR shift of 249 ppm.

Reaction of Mes<sup>\*15</sup>NPCl with LiHNMes<sup>\*</sup>: A sample of <sup>15</sup>N labelled Mes<sup>\*</sup>N=P-N(H)Mes<sup>\*</sup> was prepared by the reaction of Mes<sup>\*15</sup>NPCl with LiHMes<sup>\*</sup> in ether (0<sup>•</sup>C). Solvent was removed rapidly (20 minutes) *in vacuo*, and the light orange solids were washed with hexanes and characterized as a mixture of Mes<sup>\*</sup>N(H)P=<sup>15</sup>NMes<sup>\*</sup> and Mes<sup>\*15</sup>N(H)P=NMes<sup>\*</sup>.

- IR: 3344m, 3337m, 1599m, 1421s, 1303m, 1274s, 1241m, 1217m, 1114m, 888m,
  879m, 768w, 523w.
- NMR: <sup>31</sup>P, 269.9 (d, <sup>1</sup>J<sub>15</sup><sub>N-P</sub> 87 Hz) , 269.8 (d, <sup>1</sup>J<sub>15</sub><sub>N-P</sub> 69 Hz); <sup>15</sup>N-{H}, 307.7 (d, <sup>1</sup>J<sub>15</sub><sub>N-P</sub> 68 Hz), 139.6 (d, <sup>1</sup>J<sub>15</sub><sub>N-P</sub> 88 Hz) <sup>1</sup>J<sub>15</sub><sub>N-H</sub> 85 Hz.</sub>

**Preparation of**  $[i\mathbf{Pr}_2\mathbf{N}]_2\mathbf{P}$ -fluorenyl: Fluorenyllithium was prepared by slow addition of 4.8 ml of 1.6 M butyllithium in hexane solution to fluorene (1.27 g, 7.64 mmol) in 30 ml of ether (0°C). The solution turned bright yellow and was stirred for 30 minutes, and then was added over  $\approx$  10 minutes to a stirred solution (0 °C) of  $[i\mathbf{Pr}_2\mathbf{N}]_2\mathbf{P}$ -Cl in  $\approx$  30 ml of ether. The solution was warmed to room temperature and became colorless with precipitation of LiCl. Volatiles were removed *in vacuo*, the resulting solids were extracted and recrystallized from hexanes. The pink crystals were characterized as  $[i\mathbf{Pr}_2\mathbf{N}]_2\mathbf{P}$ -fluorenyl (2.78 g, 7.02 mmol, 92%), mp. 120-2 °C. Elemental Anal. Calcd: C, 75.72; H, 9.40; N 7.06 %. Found: C, 75.80; H, 9.37; N 6.89 %.

20

- IR: 1343m, 1297s, 1192m, 1175s, 1155m, 1117m, 1092m, 1018m, 948s, 865m, 529m, 520s.
- NMR: (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H}, 71; <sup>1</sup>H: 7.86-7.27 (8H, aromatic); 4.34 (s, 1H); 3.58-3.35 (multiplet, 4H); 1.17 (d, <sup>3</sup>J<sub>HH</sub> = 7 Hz, 12H); 0.77 (d, <sup>3</sup>J<sub>HH</sub> = 7 Hz, 12H); <sup>13</sup>C: 126.7 (s), 126.5 (s), 126.3 (s), 119.7(s), 57.4 (d, <sup>1</sup>J<sub>PC</sub> = 35 Hz), 48.9 (d, <sup>2</sup>J<sub>PC</sub> = 13 Hz), 24.1 (d, <sup>3</sup>J<sub>PC</sub> = 7 Hz), 23.9 (d, <sup>3</sup>J<sub>PC</sub> = 7 Hz), quaternary carbons not reported.
- Reaction Mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectrum of the reaction mixture indicated quantitative formation of [*i*Pr<sub>2</sub>N]<sub>2</sub>P-fluorenyl in solution.
- X-ray: monoclinic, space group  $P2_1/n$ , a = 15.710(5) Å, b = 9.66(1) Å, c = 15.93(2) Å,  $\beta = 91.48(6)$  Å, V = 2417(7) Å<sup>3</sup>, Z = 4,  $D_{calcd} = 1.089$  g cm<sup>-3</sup>,  $\mu = 1.210$ ,  $100R_w = 4.8$  (TSC).

NMR Reactions of  $[iPr_2N]_2P$ -fluorenyl with  $[Ph_3 C][BF_4]$  indicate quantitative formation of a species with a <sup>31</sup>P signal at 27.5 ppm. This species decays quantitatively to give a fluorinated species at 154 ppm (t, <sup>1</sup>J<sub>PF</sub> = 1194 Hz). In one reaction, as single crystal of Ph<sub>3</sub>C-fluorenyl was isolated and characterized by X-ray crystallography.

**Preparation of**  $[Et_2N]_2P$ -fluorenyl: Prepared in the same fashion as  $[iPr_2N]_2P$ -fluorenyl. Elemental analysis not obtained, yield 1.27 g, 3.78 mmol, 50 %, mp. 121-3 °C.

- IR: 1939w, 1901w, 1373w, 1345w, 1290s, 1189s, 1010s, 904s, 791m, 776m,
  688m, 661m, 636m, 504m, 476w, 427w, 411w.
- NMR (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H} 95 ppm; <sup>1</sup>H: 7.82-7.24 (8H, aromatic); 4.75 (d, <sup>2</sup>J<sub>PH</sub> = 4 Hz, 1H), 3.31-3.18 (multiplet, 8H); 1.14 (t, <sup>3</sup>J<sub>HH</sub> = 7 Hz, 12H) <sup>13</sup>C: 126.8, 126.6, 126.4, 119.7, 47.6 (d, <sup>1</sup>J<sub>PC</sub> = 21 Hz), 44.1 (d, <sup>2</sup>J<sub>PC</sub> = 17 Hz), 14.6 (d, <sup>3</sup>J<sub>PC</sub> = 4

Hz), quaternary carbons not reported.

**Preparation of**  $[iPr_2N]_2P(Cl)=fluorenylidene: [iPr_2N]_2P$ -fluorenyl (1.4 g, 3.6 mmol) was dissolved in  $\approx 20$  ml of pentanes. CCl<sub>4</sub> ( $\approx 1$  mL) was poured into this solution and the clear solution was allowed to stand for two days at room temperature during which time yellow block crystals began to form. The supernatant solution was decanted from the crystalline material and the crystals washed with a small amount of pentanes. All volatiles were removed *in vacuo* and the solid was characterized as  $[iPr_2N]_2P(Cl)=$ fluorenylidene (0.47 g, 1.09 mmol, 30%). A second crop of crystals was obtained (0.83 g) by recrystallization of the supernatant solids from pentanes/CCl<sub>4</sub> (30:1) on standing for three days. Total yield of  $[iPr_2N]_2P(Cl)=$ fluorenylidene 1.3 g, 3.0 mmol, 83 %; mp. 207-9 °C (dec.)

Elemental Anal. Calcd: C, 69.67; H, 8.42; N, 6.50 %. Found: C, 69.88; H, 8.38; N, 6.45 %.

- IR: 1609m, 1588m, 1553m, 1420m, 1322m, 1299s, 1278s, 1193s, 1167s, 1148s, 1118s, 1111s, 889m, 875m, 355m, 753s, 694s, 642m, 599m, 558s, 542s, 536s, 525m, 503m, 469s, 452m, 425m.
- NMR:  $(CD_2Cl_2)$ : <sup>31</sup>P{<sup>1</sup>H}: 53 ppm; <sup>1</sup>H (298 K): 8.33-7.31 (H<sub>aromatic</sub>, 8H), 3.84-3.60 (multiplet, 4H), 1.02 (d,<sup>3</sup>J<sub>HH</sub> = 7 Hz, 12H), 0.98 (d, <sup>3</sup>J<sub>HH</sub> = 7 Hz, 12H); <sup>13</sup>C (213 K): 140.7 (d, J<sub>PC</sub> = 21 Hz), 139.5 (d, J<sub>PC</sub> = 19 Hz), 130.7 (d, J<sub>PC</sub> = 18 Hz), 130.0 (d, J<sub>PC</sub> = 19 Hz), 122.8 (d, J<sub>PC</sub> = 29 Hz), 119.1 (d, J<sub>PC</sub> = 58 Hz), 118.2 (d, J<sub>PC</sub> = 68 Hz), 117.0 (d, J<sub>PC</sub> = 20 Hz), 66.3 (d, <sup>1</sup>J<sub>PC</sub> = 201 Hz), alkyl region is broad. At 293 K aryl region is broad, but alkyl region is sharp: 66.3 (d, <sup>1</sup>J<sub>PC</sub> = 201 Hz), 50.0 (d, J<sub>PC</sub> = 4 Hz), 24.2 (d, J<sub>PC</sub> = 4 Hz), 23.4 (d, J<sub>PC</sub> = 4 Hz).
- Reaction Mixture: <sup>31</sup>P{<sup>1</sup>H} NMR spectrum of the reaction mixture containing copious amounts of crystalline [*i*Pr<sub>2</sub>N]<sub>2</sub>P(Cl)=fluorenylidene indicated no presence of

starting material in solution or other phosphorus containing compounds.

X-ray: monoclinic, space group  $P2_1/c$ , a = 8.69(1) Å, b = 16.24(1) Å, c = 17.16(2) Å,  $\beta = 96.0(1)$  Å, V = 2409(8) Å<sup>3</sup>, Z = 4,  $D_{calcd} = 1.188$  g cm<sup>-3</sup>,  $\mu = 2.351$ ,  $100R_w = 4.5$  (TSC).

[Et<sub>2</sub>N]<sub>2</sub>P(Cl)=fluorenylidene: [Et<sub>2</sub>N]<sub>2</sub>P-fluorenyl (1.07 g, 3.14 mmol) was dissolved in 25 mL of hexanes. CCl<sub>4</sub> ( $\approx$  1 mL) was to the solution with immediate formation of a yellow precipitate. All volatiles were removed *in vacuo* and the solid was recrystallized from benzene and characterized as [Et<sub>2</sub>N]<sub>2</sub>P(Cl)=fluorenylidene (1.10 g, 2.93 mmol, 90%) mp. 103-4 °C. Elemental analyses not obtained.

- IR: 1610m, 1588m, 1557w, 1421s, 1356s, 1326s, 1289s, 1234m, 1203s, 1153s, 1115m, 1095m, 1061s, 1035s, 1022s, 892m, 788s, 753s, 704s, 675s, 597w, 528s, 504m, 453m, 422m, 365w, 332w.
- NMR (CD<sub>2</sub>Cl<sub>2</sub>): <sup>31</sup>P{<sup>1</sup>H}: 58; <sup>1</sup>H: 8.09-7.09 (8H aromatic), 3.52-3.20 (multiplet, 8H), 1.29 (t, J<sub>HH</sub> = 7 Hz, 12H); <sup>13</sup>C: 124.4 (s), 119.7 (s), 118.7 (s), 117.8(s), 64.5 (d, <sup>1</sup>J<sub>PC</sub> = 205 Hz) 40.8 (d, <sup>2</sup>J<sub>PC</sub>= 5 Hz), 13.5 (d, <sup>3</sup>J<sub>PC</sub> = 2 Hz). Alkyl peaks become broad on cooling.

**Preparation of**  $[iPr_2N]_2P(CI)=$ fluorenylidene-GaCl<sub>3</sub> complex: A solution of  $[iPr_2N]_2P(CI)=$ fluorenylidene (0.21 g, 0.47 mmol) in benzene ( $\approx 20$  mL) was added to a stirred solution of GaCl<sub>3</sub> (0.085 g, 0.47 mmol) in benzene ( $\approx 20$  mL) in a 10 °C bath. A white precipitate formed immediately and the solvent was decanted from the solid which was washed by back-distillation. The volatiles were removed *in vacuo* and the white powder was characterized as  $[iPr_2N]_2P(CI)=$ fluorenylidene-GaCl<sub>3</sub> complex: mp. 105 °C (dec.).

IR: 1305m, 1195m, 1165m, 1149m, 1140m, 1110m, 994s, 972s, 942m, 799m, 740s,

681m, 606m, 558m, 540s, 366s, 359s, 345s.

- NMR (CD<sub>2</sub>Cl<sub>2</sub>, -80 ° C): <sup>31</sup>P{<sup>1</sup>H} 71; <sup>13</sup>C and <sup>1</sup>H were obtained but peaks were poorly resolved and difficult to interpret due to facile isomerization.
- Reaction Mixture: <sup>31</sup>P NMR reactions in benzene indicate quantitative precipitation of the complex with no starting material remaining in solution.

## Isomerisation of the $[iPr_2N]_2P(Cl)$ =fluorenylidene-GaCl<sub>3</sub> Complex:

 $[i\Pr_2N]_2P(Cl)=$ fluorenylidene-GaCl<sub>3</sub> was prepared as described above using (GaCl<sub>3</sub>: 0.14 g, 0.79 mmol,  $\approx$  30 mL benzene;  $[^i\Pr_2N]_2P(Cl)=$ fluorenylidene: 0.34 g, 0.77 mmol,  $\approx$  30 mL benzene). The resulting white powder of  $[i\Pr_2N]_2P(Cl)=$ fluorenylidene/GaCl<sub>3</sub> complex was then dissolved in ( $\approx$  40 mL) CH<sub>2</sub>Cl<sub>2</sub>/pentanes (75/25). Slow removal of solvent overnight yielded light pink crystals characterized as 9-di-*i*-propylamino-10-di-*i*-propyliminium-9, 10-dihydro-9-phosphaphenanthrene tetrachlorogallate (0.39 g, 0.64 mmol, 83%) mp. 166-7 °C.

- Elemental Anal. Calcd: C, 49.46; H, 5.98; N, 4.61 %. Found: C, 49.23; H, 6.08; N, 4.54 %.
- IR: 1582s, 1195s, 1167s, 1138s, 1113s, 1017s, 964s, 874m, 785m, 758s, 738s, 687m, 678m, 517m, 489m, 456m, 372s.
- NMR: (CD<sub>2</sub>Cl<sub>2</sub>, -80 C): <sup>31</sup>P{<sup>1</sup>H} 22; <sup>1</sup>H 8.03-7.17 (8H, aromatic), 4.79 (multiplet, 1H), 4.42 (multiplet, 1H), 3.22 (multiplet, 1H), 2.74 (multiplet, 1H), 2.01 (d, <sup>3</sup>J<sub>HH</sub> = 6.4 Hz), 1.52 (d, <sup>3</sup>J<sub>HH</sub> = 6.1 Hz), 1.44 (d, <sup>3</sup>J<sub>HH</sub> = 6.8 Hz), 1.39 (d, <sup>3</sup>J<sub>HH</sub> = 6.0 Hz), 1.07 (d, <sup>3</sup>J<sub>HH</sub> = 5.9 Hz), 0.97 (d, <sup>3</sup>J<sub>HH</sub> = 5.9 Hz), 0.48 (d, <sup>3</sup>J<sub>HH</sub> = 5.7 Hz), -0.04 (d, <sup>3</sup>J<sub>HH</sub> = 6.0 Hz) total integration 24H; <sup>13</sup>C 213.9 (d, <sup>1</sup>J<sub>PC</sub> = 89 Hz, C=N), <sup>13</sup>C NMR complex and not fully resolved and assigned.
- Reaction Mixtures: <sup>31</sup>P{<sup>1</sup>H} NMR reactions of [*i*Pr<sub>2</sub>N]<sub>2</sub>P(Cl)=fluorenylidene/GaCl<sub>3</sub> complex in CH<sub>2</sub>Cl<sub>2</sub> indicate quantitative isomerisation. NMR reactions with

 $[iPr_2N]_2P(Cl)$ =fluorenylidene and GaCl<sub>3</sub> in CH<sub>2</sub>Cl<sub>2</sub> solution indicate quantitative formation of  $[iPr_2N]_2P(Cl)$ =fluorenylidene/GaCl<sub>3</sub> which isomerizes quantitatively to give 9-di-*iso*-propylamino-10-di-*iso*-propyliminium-9, 10-dihydro-9phosphaphenanthrene tetrachlorogallate.

X-ray: monoclinic, space group 
$$P2_1/n$$
,  $a = 9.884(3)$  Å,  $b = 21.189(4)$  Å,  $c = 14.639(4)$   
Å,  $\beta = 95.12(2)$  Å,  $V = 3054(2)$  Å<sup>3</sup>,  $Z = 4$ ,  $D_{calcd} = 1.320$  g cm<sup>-3</sup>,  $\mu = 13.18$ ,  
 $100R_w = 3.7$  (TSC).

NMR reactions of  $[Et_2N]_2P(Cl)$ =fluorenylidene and GaCl<sub>3</sub> in CH<sub>2</sub>Cl<sub>2</sub> solution indicate quantitative formation of  $[Et_2N]_2P(Cl)$ =fluorenylidene/GaCl<sub>3</sub> (<sup>31</sup>P{<sup>1</sup>H} 72) which does not isomerize (stable for 1 month).

NMR reactions of  $[Et_2N]_2P(Cl)=$ fluorenylidene with HOSO<sub>2</sub>CF<sub>3</sub> in CD<sub>2</sub>Cl<sub>2</sub> indicate quantitative formation of { $[Et_2N]_2P(Cl)$ -fluorenyl}{OSO<sub>2</sub>CF<sub>3</sub>}: NMR <sup>31</sup>P 72 ppm; <sup>1</sup>H 7.92-7.33 (8H, aromatic), 5.74 (d, <sup>2</sup>J<sub>PH</sub> = 18 Hz, 1H), 3.17-2.93 (m, 8H), 0.97 (t, <sup>3</sup>J<sub>HH</sub> = 7 Hz, 12H); <sup>13</sup>C: 130.7(d, J<sub>PC</sub> = 3 Hz), 128.7(d, J<sub>PC</sub> = 4 Hz), 127.1 (d, J<sub>PC</sub> = 4 Hz), 121.6 (d J<sub>PC</sub> = 2 Hz), 50.4 (d, <sup>1</sup>J<sub>PC</sub> = 89 Hz), 42.1 (s), 12.5 (s), 12.4 (s) (no change after one month). <sup>13</sup>C multiplicity analysis (JMOD) is consistent with the assignment. Reaction of [*i*Pr<sub>2</sub>N]<sub>2</sub>P(Cl)=fluorenylidene with HOSO<sub>2</sub>CF<sub>3</sub> in CH<sub>2</sub>Cl<sub>2</sub> have a single <sup>31</sup>P signal at 71 ppm (no change after one month).

**Preparation of 4 SbCl<sub>3</sub>:Diphenylacetylene Complex:** Diphenylacetylene (0.39 g, 2.19 mmol) in  $\approx 25$  ml of hexanes was added dropwise over a period of 10 minutes to a solution of antimony trichloride (1.98 g, 8.68 mmol) in  $\approx 25$  ml CH<sub>2</sub>Cl<sub>2</sub>. Solvents were removed from the clear solution *in vacuo* over a three hour period, during which clear, shiny plates formed. The crystalline material was recrystallized from 50.50 CH<sub>2</sub>Cl<sub>2</sub>\hexanes, and the clear plate crystals were placed under dynamic vacuum for thirty minutes and characterized as C<sub>14</sub>H<sub>10</sub>Sb<sub>4</sub>Cl<sub>12</sub> (2.02 g, 1,85 mmol, 85 %), mp. 76-78 °C.
Elemental Anal. Calcd: C, 15.42; H, 0.92 %.

- IR: 1974w, 1913w, 1854w, 1798w, 780s, 692s.
- NMR (CD<sub>2</sub>Cl<sub>2</sub>): <sup>1</sup>H: 7.56-7.35 (10H, aromatic); <sup>13</sup>C: 132.0, 129.0, 128.9, quaternary carbons not observed.
- X-ray: triclinic, space group *PL*, a = 8.379(2) Å, b = 11.965(3) Å, c = 8.122(1) Å,  $\alpha = 95.51(2)^{\circ}$ ,  $\beta = 110.37(1)^{\circ}$ ,  $\gamma = 94.69(2)^{\circ}$ , V = 754.0(3) Å<sup>3</sup>, Z = 1,  $D_{calcd} = 2.402$ g cm<sup>-3</sup>,  $\mu = 46.51$  cm<sup>-1</sup>,  $100R_w = 2.90$  (TSC).

**Preparation of 4 SbCl<sub>3</sub>:***t*-Stilbene Complex: *t*-Stilbene (0.55 g, 3.05 mmol) in  $\approx$  15 ml of hexanes was added dropwise to a solution of antimony trichloride (2.80 g, 12.27 mmol) in  $\approx$  15 ml CH<sub>2</sub>Cl<sub>2</sub> over a period of 10 minutes. Solvents were removed from the clear solution *in vacuo* over a three hour period, during which clear, shiny plates formed. The crystalline material was recrystallized from 50/50 CH<sub>2</sub>Cl<sub>2</sub>/hexanes, and the clear plate crystals were placed under dynamic vacuum for thirty minutes and characterized as C<sub>14</sub>H<sub>12</sub>Sb<sub>4</sub>Cl<sub>12</sub> (3.05 g, 2.79 mmol, 91 %), mp. 101-103 °C.

Elemental Anal. Calcd: C, 15.39; H, 1.11 %.

Found: C, 15.24; H, 1.11 %.

- IR: 1966w, 1899w, 1845w, 1775w, 779s, 704s.
- NMR (CD<sub>2</sub>Cl<sub>2</sub>): <sup>1</sup>H: 7.54-7.23 (10H, aromatic), 7.13 (s, 2H); <sup>13</sup>C: 129.2, 129.0, 128.2, 127.0, quaternary carbons not observed.
- X-ray: triclinic, space group *P1*, a = 8.363(3) Å, b = 11.950(3) Å, c = 8.099(3) Å,  $a = 95.44(2)^{\circ}$ ,  $\beta = 109.15(3)^{\circ}$ ,  $\gamma = 96.28(2)^{\circ}$ , V = 752.6(4) Å<sup>3</sup>, Z = 1,  $D_{calcd} = 2.411$  gcm<sup>-3</sup>,  $\mu = 46.15$  cm<sup>-1</sup>,  $100R_w = 8.39$  (TSC).

**Preparation of 4 SbCl<sub>3</sub>:Bibenzyl Complex:** Bibenzyl (0.71 g, 3.90 mmol) in  $\approx$  40 ml of hexanes was added dropwise to a solution of antimony trichloride (3.58 g, 15.69 mmol) in  $\approx$  20 ml CH<sub>2</sub>Cl<sub>2</sub> over a period of 10 minutes. Solvents were removed from the clear solution *in vacuo* over a three hour period, during which clear, shiny plates formed. The crystalline material was recrystallized from 50/50 CH<sub>2</sub>Cl<sub>2</sub>/hexanes, and the clear plate crystals were placed under dynamic vacuum for thirty minutes and characterized as C<sub>14</sub>H<sub>14</sub>Sb<sub>4</sub>Cl<sub>12</sub> (3.94 g, 3.60 mmol, 92 %), mp. 78-79 °C.

Elemental Anal. Calcd: C, 15.36; H, 1.29 %.

Found: C, 15.57; H, 1.37 %.

- IR: 1974w, 1894w, 1840w, 1761w, 785s, 720s.
- NMR (CD<sub>2</sub>Cl<sub>2</sub>): <sup>1</sup>H: 7.30-7.18 (10H, aromatic), 2.92 (s, 4H); <sup>13</sup>C: 129.0, 128.8, 126.4, 38.1, quaternary carbons not observed.
- X-ray: triclinic, space group *PL*, a = 8.384(2) Å, b = 11.950(4) Å, c = 8.103(2) Å,  $\alpha = 94.81(3)^{\circ}$ ,  $\beta = 108.64(2)^{\circ}$ ,  $\gamma = 97.42(3)^{\circ}$ , V = 756.0(4) Å<sup>3</sup>, Z = 1,  $D_{calcd} = 2.404$  gcm<sup>-3</sup>,  $\mu = 45.94$  cm<sup>-1</sup>,  $100R_w = 7.34$  (TSC).

## References

- 1. March, J. Advanced Organic Chemistry: Reactions, Mechanism, and Structure, 4th ed.; John Wiley and Sons: New York, 1992.
- 2. Kutzelnigg, W. Angew. Chem. Int. Ed. Engl. 1984, 23, 272.
- 3. Weber, L. Chem Rev. 1992, 92, 1839.
- 4. Kuchen, W.; Buchwold, H. Chem. Ber. 1958, 91, 2296.
- 5. See, for example (a) Daly, J. J.; Maier, L. Nature 1965, 208, 383. (b) Daly, J. J.; Maier, L. Nature 1964, 203, 1168.
- 6. Norman, N. C. Polyhedron 1993, 12, 2431.
- 7. Yoshifuji, M.; Shima, I.; Inamoto, N. J. Am. Chem. Soc. 1981, 103, 4587.
- 8. A survey of the Cambridge Structural Database in January, 1995 reveals that there have been over three hundred structural reports of molecules, not all of which would be considered low coordinate, containing Mes\*.
- See, for example, (a) Hitchcock, P. B.; Jasim, H. A.; Lappert, M. F.; Williams, H. D. J. Chem. Soc., Chem. Commun. 1984, 662. (b) Atwood, D. A.; Cowley, A. H.; Jones, R. A.; Mardones, M. A. J. Am. Chem. Soc. 1991, 113, 7050. (c) Petrie, M. A.; Ruhlandt-Senge, K.; Power, P. P. Inorg. Chem. 1992, 31, 4038. (d) Schulz, S.; Pusch, S.; Pohl, E.; Dielkus, S.; Herbst-Irmer, R.; Meller, A.; Roesky, H. W. Inorg. Chem. 1993, 32, 3343. (f) Wehmschulte, R. J.; Ellison, J. J.; Ruhlandt-Senge, K.; Power, P. P. Inorg. Chem. 1994, 33, 6300. (g) Wehmschulte, R., J.; Power, P. P. Inorg. Chem. 1994, 33, 3611. (h) Brothers, P. J.; Wehmschulte, R. J.; Olmstead, M. M.; Ruhlandt-Senge, K.; Parkin, S. R.; Power, P. P. Organometallics 1994, 13, 2792. (i) Cowley, A. H.; Gabbaï, F. P.; Deckon, A. Angew. Chem. Int. Ed. Engl. 1994, 33, 1371. (j) Cowley, A. H.; Gabbaï, F. P.; Isom, H. S.; Carrano, C. J.; Bond, M. R. Angew. Chem. Int. Ed. Engl. 1994, 33. 1253.
- 10. Jutzi, P.; Becker, A., Leue, C.; Stammler, H. G.; Neumann, B. Organometallics 1991, 10, 3838.
- 11. Underiner, G. E.; Tan, R. P.; Powell, D. R.; West, R. J. Am. Chem. Soc. **1991**, 113, 8437.
- 12. Modrelli, D. A.; Lahti, P. M.; George, C. J. Am. Chem. Soc. 1991, 113, 6329.
- 13. See, for example, Cowley, A. H.; Kilduff, J. E.; Lasch, J. G.; Mehrotra, S. K.; Norman, N. C.; Pakulski, M.; Whittlesey, B. R.; Atwood, J. A.; Hunter, W. E. Inorg. Chem. 1984, 23, 2582.
- 14. Cowley, A. H.; Lasch, J. G.; Norman, N. C.; Pakulski, M.; Whittlessey, B. R. J. Chem. Soc., Chem. Commun. 1983, 881.

- 15. Cowley, A. H.; Lasch, J. G.; Norman, N. C.; Pakulski, M. J. Am. Chem. Soc. 1983, 105, 5506.
- 16. Märkel, G.; Sejpka, H. Tetrahedron Lett. 1986, 27, 171.
- 17. Märkel, G.; Sejpka, H. Angew. Chem. Int. Ed. Engl. 1986, 25, 264.
- 18. Appel, R.; Knoch, F.; Kunze, H. Angew. Chem. Int. Ed. Engl. 1984, 23, 157. See also Romanenko, V. D.; Ruban, A. V.; Drapailo, A. B.; Chernega, A. N.; Rusanov, E. B. Heteroatom Chem. 1992, 3, 181.
- 19. Bender, H. R. G.; Niecke, E.; Nieger, M. J. Am. Chem. Soc. 1993, 115, 3314.
- 20. Appel, R.; Paulen, W. Angew. Chem. Int. Ed. Engl. 1983, 22, 785.
- 21. Appel, R.; Fölling, P.; Josten, B.; Siray, M., Winkhaus, V.; Knoch, F. Angew. Chem. Int. Ed. Engl. 1984, 23, 619.
- 22. du Mont, W.-W.; Kubiniok, S.; Peters, K.; von Schnering, H.-G. Angew. Chem. Int. Ed. Engl. 1987, 26, 780.
- 23. Healy, M. D.; Power, M. B.; Barron, A. R. Coord. Chem. Rev. 1994, 130, 63.
- See, for example, (a) Healy, M. D.; Leman, J. T.; Barron, A. R. J. Am. Chem. Soc. 1991, 113, 2776. (b) Healy, M. D.; Barron, A. R. Angew. Chem. Int. Ed. Engl. 1992, 31, 921.
- 25. Brook, A. G.; Nyburg, S. C.; Abdesaken, B.; Gutekunst, B.; Gutekunst, G.; Kallury, R. K. M. R.; Poon, Y. C.; Chang, Y. M.; Wong-Ng, W. J. Am. Chem. Soc. 1982, 104, 5667.
- 26. Jutzi, P; Meyer, U.; Krebs, B.; Dartmann, M. Angew. Chem. Int. Ed. Engl. **1986**, 25, 919.
- 27. Cowley, A. H.; Kilduff, J. E.; Newman, T. H.; Pakulski, M. J. J. Am. Chem. Soc. 1982, 104, 5820.
- 28. See, for example, (a) Tokitoh, N.; Matsumoto, T.; Manmaru, K.; Okazaki, R. J. Am. Chem. Soc. 1993, 115, 8855. (b) Tokitoh, N.; Takeda, N.; Okazaki, R. J. Am. Chem. Soc. 1994, 116, 7907.
- For excellent reviews of iminophosphines see (a) Niecke, E.; Gudat, D. Angew. Chem. Int. Ed. Engl. 1991, 30, 217. (b) Niecke, E. In Multiple Bonds and Low Coordination in Phosphorus Chemistry; Regitz, M.; Scherer, O. J., Eds; Thierne: New York, 1990; pp 293-320.
- 30. Niecke, E.; Ruger, R.; Schoeller, W. W. Angew. Chem. Int. Ed. Engl. 1981, 20, 1034.
- 31. Niecke, E.; Nieger, M.; Reichert, F. Angew Chem. Int. Ed. Engl. 1988, 27,

1715. See also, Niecke, E.; Nieger, M.; Reichert, F.; Schoeller, W. W. Angew Chem. Int. Ed. Engl. 1988, 27, 1713.

- See, for example, (a) Niecke, E.; Nieger, M.; Gärtner-Winkhaus, C.; Kramer, B. Chem. Ber. 1990, 123, 477. (b) Niecke, E.; Detsch, R.; Nieger, M. Chem. Ber. 1990, 123, 797. (c) Hein, J.; Gärtner-Winkhaus, C.; Nieger, M.; Niecke, E. Heteroatom 1991, 3, 409. (d) Detsch, R.; Niecke, E.; Nieger, M.; Reichert, F. Chem. Ber. 1992, 25, 321. (e) Niecke, E.; Nieger, M.; Wendroth, P. J. Am. Chem. Soc. 1993, 115, 6989. (f) Ruban, A.; Nieger, M.; Niecke, E. Angew. Chem. Int. Ed. Engl. 1993, 32, 1419.
- See, for example, (a) Niecke, E.; Hein, J.; Nieger, M. Organometallics 1989, 8, 2290. (b) Hein, J.; Niecke, E. J. Chem. Soc., Chem. Commun. 1991, 48. (c) Hein, J.; Nieger, M.; Niecke, M. Organometallics 1991, 10, 13. (d) Hein, J.; Niecke, E.; Meidine, M. F.; Trigo Passos, B. F.; Nixon, J. F. J. Chem. Soc., Chem. Commun. 1991, 41. (e) Hein, J.; Niecke, E. J. Chem. Soc., Chem. Commun. 1991, 48. (f) Niecke, E.; Nixon, J. F.; Wendroth, P.; Trigo Passos, B. F.; Nieger, M. J. Chem. Soc., Chem. Commun. 1991, 48. (f) Niecke, E.; Nixon, J. F.; Wendroth, P.; Trigo Passos, B. F.; Nieger, M. J. Chem. Soc., Chem. Commun. 1993, 846. (g) Igau, A.; Dufour, N.; Mahieu, A.; Majoral, J.-P. Angew. Chem. Int. Ed. Engl. 1993, 32, 95.
- See, for example, (a) Barion, D.; David, G.; Link, M.; Nieger, M.; Niecke, E. Chem. Ber. 1993, 126, 649. (b) Link, M.; Niecke, E.; Nieger, M. Chem. Ber. 1994, 127, 313.
- 35. Niecke, E.; v. d. Gonna, V.; Nieger, M. Chem. Ber. 1990, 123, 2329.
- 36. Cowley, A. H. Acc. Chem. Res. 1984, 17, 386. See also Cowley, A. H.; Norman, N. C. In Progress in Inorganic Chemistry, Vol. 34; Lippard, S. J., Ed.; John Wiley and Sons: New York, 1986; pp 2-63.
- PCMODEL, Serena Software, Bloomington, Indiana.
- 38. Wiberg, K. B. Angew. Chem. Int. Ed. Engl. 1986, 25, 312.
- 39. Carey, F. A.; Sundberg, R. J. Advanced Organic Chemistry Part A: Structure and Mechanism; Plenum Press: New York, 1990; p 158.
- 40. Mes\* was approximated by 2, 6-di-*tert*-butylphenyl since Mes\* produced a molecule containing too many atoms for the program to calculate.
- 41. Tidwell, T. T. Tetrahedron 1978, 34, 1855.
- 42. Maier, G.; Pfriem, S.; Schäfer, U.; Matusch, R. Angew. Chem. Int. Ed. Engl. 1978, 17, 520.
- 43. Carey, F. A.; Sundberg, R. J. Advanced Organic Chemistry Part A: Structure and Mechanism; Plenum Press: New York, 1990; p 12.
- 44. Corbridge, D. E. C. *Phosphorus: An Outline of its Chemistry, Biochemistry, and Technology*, 4th ed.; Studies in Inorganic Chemistry 10; Elsevier: New York,

1990; p 44.

- 45. Schoeller, W. W. In Multiple Bonds and Low Coordination in Phosphorus Chemistry; Regitz, M.; Scherer, O. J., Eds.; Thieme: New York, 1990; pp 5-32.
- 46. Schiffer, H.; Ahlrichs, R.; Häser, M. Theor. Chim. Acta 1989, 75, 1.
- 47. March, J. Advanced Organic Chemistry: Reactions, Mechanism, and Structure, 4th ed.; John Wiley and Sons: New York, 1992; p 20.
- 48. Corbridge, D. E. C. Phosphorus: An Outline of its Chemistry, Biochemistry, and Technology, 4th ed.; Studies in Inorganic Chemistry 10; Elsevier: New York, 1990; p 45.
- 49. March, J. Advanced Organic Chemistry: Reactions, Mechanism, and Structure, 4th ed.; John Wiley and Sons: New York, 1992; p 208.
- 50. Burford, N.; Clyburne, J. A. C.; Mason, S.; Richardson, J. F. Inorg. Chem. 1993, 32, 4988.
- 51. Romanenko, V. D.; Ruban, A. V.; Kalibabchuk, N. N.; Iksanova, S. V.; Markovskii, L. N. Zh. Obshch. Khim. 1981, 51, 1726.
- (a) Scherer, O. J.; Kuhn, N. J. Organomet. Chem. 1974, 82, C3. (b) Markovskii, L. N.; Romanenko, V. D.; Ruban, A. V. Phosphorus and Sulfur 1980, 9, 221. (c) Markovskii, L. N.; Romanenko, V. D.; Klebanski, E. O.; Povolotskii, M. I.; Chernega, A. N.; Antipin, M. Yu.; Struchkov, M. Yu. Zh. Obshch. Khim. 1986, 56, 1721. (d) Markovskii, L. N.; Romanenko, V. D.; Drapailo, A. B.; Ruban, A. V. Zh. Obshch. Khim. 1986, 56, 2231.
- 53. Boisdon, M. -T.; Lopez, L.; Malavaud, C.; Barrans, J.; Chabane, A.; Mathis, R. Can. J. Chem. 1986, 64, 1725.
- 54. Mason, S. Post-Doctoral Fellow, Dalhousie University, 1991-1993.
- (a) Tarassoli, A.; Haltiwanger, R. C.; Norman, A. D. Inorg. Chem. 1982, 21, 2684.
   (b) Thompson, M. L.; Tarassoli, A.; Haltiwanger, R. C.; Norman, A. D. Inorg. Chem. 1987, 26, 684.
- 56. Beachley, O. T., Jr.; Churchill, M. R.; Pazik, J. C.; Ziller, J. W. Organometallics 1987, 6, 2088. See also Schultz, S.; Pusch, S.; Pohl, E.; Dielkus, S.; Herbst-Irmer, R.; Meller, A.; Roesky, H. Inorg. Chem. 1993, 32, 3343.
- 57. Klapotke, T.; Passmore, J. Acc. Chem. Res. 1989, 7, 234.
- 58. Burford, N.; Clyburne, J. A. C.; Gates, D. P.; Schriver, M. J.; Richardson, J. F. J. Chem. Soc., Dalton Trans. 1994, 997.
- 59. See, for example, Corbridge, D. E. C. Phosphorus: An Outline of its Chemistry, Biochemistry and Technology, 4th ed.; Studies in Inorganic Chemistry 10; Elsevier: New York, 1990; Chapter 5.

- 60. Schmidpeter, A. In *The Chemistry of Inorganic Homo- and Hetero-cycles*; Haiduc, I.; Sowerby, D. B., Eds.; Academic Press; New York, 1987; Vol. II, p 617.
- Chernega, A. N.; Antipin, M. Yu.; Struchkov, Yu. T.; Boldeskul, I. E.; Marchenko, A. P.; Pinchuk, A. M. J. Struct. Chem. (Engl. Transl.) 1987, 28, 274.
- 62. Berman, M. Top. Phos. Chem. 1972, 7, 311.
- 63. The structure of [C(N=PCl<sub>3</sub>)][SbCl<sub>6</sub>] has been determined (Müller, v. U. Z. Anorg. Allg. Chem. 1980, 468, 117), but the cationic nature of this species prevents direct comparison to other phosphine imides.
- 64. Antipin, M. Yu.; Struchkov, Yu. T.; Yurchenko, V. M.; Kozlov, E. S. J. Struct. Chem. (Engl. Transl.) 1982, 23, 227.
- 65. Antipin, M. Yu.; Struchkov, Yu. T.; Kozlov, E. S. J. Struct. Chem. (Engl. Transl.) 1986, 26, 575.
- 66. Schriver, M. J. Post-Doctoral Fellow, Dalhousie University.
- 67. Huheey, J. E.; Keiter, E. A.; Keiter, R. L. Inorganic Chemistry: Principles of Structure and Reactivity, 4th ed.; Harper and Row: New York, 1993.
- 68. Burford, N.; Clyburne, J. A. C.; Bakshi, P. K.; Cameron, T. S. Phosphorus, Sulfur Relat. Elem. 1994, in press.
- 69. Haase, M.; Klingebiel, U.; Skoda, L. Z. Naturforsch. 1984, 39b, 1500.
- 70. See, for example, Corbridge, D. E. C. *Phosphorus: An Outline of its Chemistry, Biochemistry and Technology*, 4th ed.; Studies in Inorganic Chemistry 10; Elsevier: New York, 1990; Chapter 4.
- 71. Böhme, E.; Dehnicke, K.; Beck, J.; Hiller, W.; Strähle, J. Z. Naturforsch. 1988, 43b, 138.
- 72. Kulpe, S.; Seidel, J.; Bödeker, J.; Köckritz, P. Cryst. Res. Technol. 1984, 19, 649.
- 73. Cameron, T. S.; Mannan, K; Biddlestone, M.; Shaw, R. A. Z. Naturforsch. 1975, 30b, 973.
- 74. Dehnicke, K.; Strähle, J. Polyhedron 1989, 8, 707.
- 75. Moritani, T.; Kuchitsu, K.; Morino, Y. Inorg. Chem. 1971, 10, 344.
- (a) Glaser, R.; Horan, C. J.; Haney, P. E. J. Phys. Chem. 1993, 97, 1835. (b)
  Glaser, R.; Horan, C. J.; Choy, G. S.-C.; Harris, B. L. J. Phys. Chem. 1992, 96, 3689.

- 77. Buenker, R. J.; Bruna, P. J.; Peyerimhoff, S. D. Isreal J. Chem. 1980, 19, 309.
- See, for example, Godfrey, M.; Kelly, D. G.; Mackie, A. G.; McAuliffe, C. A.; Pritchard, R. G.; Watson, S. M. J. Chem. Soc., Chem. Commun. 1991, 1447; Godfrey, M.; Kelly, D. G.; Mackie, A. G.; McAuliffe, C. A.; Pritchard, R. G.; Watson, S. M. J. Chem. Soc., Chem. Commun. 1992, 355; Bricklebank, N.; Godfrey, M.; Kelly, D. G.; Mackie, A. G.; McAuliffe, C. A.; Prithchard, R. G.; Kobryn, P. J. J. Chem., Soc. Dalton Trans. 1993, 101.
- 79. Power, P. P.; Pestana, D. C. J. Am. Chem. Soc. 1989, 111, 6887.
- 80. Petri, M. A.; Olmstead, M. M.; Power, P. P. J. Am. Chem. Soc. 1991, 113, 8704.
- 81. The Chemistry of Diazonium and Diazo Groups; Patai, S., Ed.; Wiley: New York, 1978.
- 82. Curtis, J. M.; Burford, N.; Parks, T. P. Org. Mass Spectrom. 1994, 29, 414.
- 83. David, G.; Niecke, E.; Nieger, M. Tetrahedron Lett. 1992, 33, 2335.
- 84. David, G.; Niecke, E.; Nieger, M.; v. d. Gonna, V.; Schoeller, W. W. Chem. Ber. 1993, 126, 1513.
- 85. Gudat, D.; Schiffner, H. M.; Nieger, M.; Stalke, D.; Blake, A. J.; Grondey, H.; Niecke, E. J. Am. Chem. Soc. 1992, 114, 8857.
- 86. Meissler, G. L.; Tarr, D. A. Inorganic Chemistry; Prentice Hall: Englwood Cliffs, New Jersey, 1991.
- 87. Elschenbroich, Ch.; Salzer, A. Organometallics: A Concise Introduction; VCH: New York, 1989.
- 88. See, for example, Brayer, G. D.; James, M. N. G. Acta Cryst. 1982, 38b, 654 and references therein.
- See, for example, (a) Sanchez, M.; Mazières, M. R.; Lamandé, L.; Wolf, R. In Multiple Bonds and Low Coordination in Phosphorus Chemistry; Regitz, M.; Scherer, O. J., Eds.; Thieme: New York, 1990; pp 129-148. (b) Cowley, A. H.; Kemp, R. A. Chem. Rev. 1985, 85, 367.
- 90. Reed, C. A.; Xie, Z.; Bau, R.; Benesi, A. Science 1993, 262, 402.
- 91. Burford, N; Parks, T. M.; Royan, B. W.; Borecka, B.; Cameron, T. S.; Richardson, T. S.; Gabe, E. J.; Hynes, R. J. Am. Chem. Soc. 1992, 114, 8147.
- 92. Bahr, S. R.; Boudjouk, P. J. Am. Chem. Soc. 1993, 115, 4514.
- 93. Breliere, C.; Carre, F.; Corriu, R.; Man, M. W. C. J. Chem. Soc., Chem Commun. 1994, 2333.

- 94. Schultz, C. W.; Parry, R. W. Inorg. Chem. 1976, 15, 3046.
- 95. Reed, R.; Reau, R.; Dahan, F.; Bertrand, G. Angew. Chem. Int. Ed. Engl. 1993, 32, 399.
- 96. Burford, N.; Losier, P.; Sereda, S. V.; Cameron, T. S.; Wu, G. J. Am. Chem. Soc. 1994, 116, 6474.
- 97. David, G.; Niecke, E.; Nieger, M.; Radseck, J. J. Am. Chem. Soc. 1994, 116, 2191.
- 98. (a) Cowley, A. H.; Gabbaï, F. P.; Atwood, D. A.; Carrano, C. J.; Mokry, L. M.; Bond, M. R. J. Am. Chem. Soc. 1994, 116, 1559. (b) Olazábal, C. A.; Gabbaï, F. P.; Cowley, A. H.; Carrano, C. J.; Mokry, L. M.; Bond, M. R. Organometallics 1994, 13, 421. (c) Mocker, M.; Robl, C.; Schnöckel, H. Angew. Chem. Int. Ed. Engl. 1994, 33, 862.
- 99. Niecke, E.; David, G.; Detsch, R., Kramer, B.; Nieger, M.; Wendroth, P. *Phosphorus, Sulfur Silicon Relat. Elem.* 1993, 76, 25.
- 100. (a) Jutzi, P. J. Organometal. Chem. 1990, 400 1. (b) Jutzi, P. Chem. Rev. 1986, 86, 983.
- 101. Jutzi, P.; Wippermann, T.; Kruger, C.; Kraus, H.-J. Angew. Chem. Int. Ed. Engl. 1983, 22, 250.
- 102. See, for example, Maslowsky, E. J. Chem. Educ. 1993, 70, 980.
- 103. See, for example, (a) Schmidbaur, H.; Thewalt, U.; Zafiropoulos, T. Organometallics 1983, 2, 1550. 9b) Schmidbaur, H.; Thewalt, U.; Zafiropoulos, T. Angew. Chem. Int. Ed. Engl. 1984, 23, 76. (c) Schmidbaur, H. Angew. Chem. Int. Ed. Engl. 1985, 24, 893. (d) Schmidbaur, H.; Bublack, W.; Huber, B.; Müller, G. Organometallics 1986, 5, 1647.
- 104. Ebenhöch, J.; Müller, G.; Riede, J.; Schmidbaur, H. Angew. Chem. Int. Ed. Engl. 1984, 23, 386.
- See, for example, (a) Weininger, M. S.; Rodesiler, P. F.; Amma, E. L. Inorg. Chem. 1979 18, 751. (b) Lefferts, J. L.; Hossain, M. B.; Molloy, K. C.; v. d. Helm, D.; Zuckerman, J. J. Angew. Chem. Int. Ed. Engl. 1980, 19, 309. (c) Schmidbaur, H.; Probst, T.; Huber, B.; Müller, G.; C. Krüger, C. J. Organometal. Chem. 1989, 365, 53. (d) Probst, T.; Steigelmann, O.; Riede, J.; Schmidbaur, H. Angew. Chem. Int. Ed. Engl. 1990, 29, 1397. (e) Luth, H.; Amma, E. L. J. Am. Chem. Soc. 1969, 91, 7515. (f) Weininger, M. S.; Rodesiler, P. F.; Gash, A. G.; Amma, E. L. J. Am. Chem. Soc. 1972, 94, 2135. (g) Rodesiler, P. F.; Auel, Th.; Amma, E. L. J. Am. Chem. Soc. 1975, 97, 7405.
- 106. Gash, A. G.; Rodesiler, P. F.; Amma, E. L. Inorg. Chem. 1974, 13, 2429.
- 107. Frank, W.; Weber, J.; Fuchs, E. Angew. Chem. Int. Ed. Engl. 1987, 26, 74.

- 108. See, for example, (a) Hassel, O.; Strømme, K. O. Acta Scand. 1958, 12, 1146.
  (b) Hassel, O.; Strømme, K. O. Acta Scand. 1959, 13, 1781.
- 109. Jutzi, P. Adv. Organomet. Chem. 1986, 26, 217.
- 110. Burford, N.; Clyburne, J. A. C.; Bakshi, P. K.; Cameron, T. S. J. Am. Chem. Soc. 1993, 115, 8829.
- 111. As previously noted for the [AlCl4] salt by Curtis, R. D.; Schriver, M. J.; Wasylishen, R. E. J. Am. Chem. Soc. 1991, 113, 1493.
- 112. Klinzing, P.; Willing, W.; Müller, U.; Dehnicke, K. Z. Anorg. Allg. Chem. 1985, 520, 35.
- 113. See, for example, Corbridge, D. E. C. Phosphorus: An Outline of its Chemistry, Biochemistry and Technology, 4th ed.; Studies in Inorganic Chemistry 10; Elsevier: New York, 1990; pp 125-144.
- 114. Burford, N.; Losier, P.; Macdonald, C.; Kyrimis, V.; Bakshi, P. K.; Cameron, T. S. Inorg. Chem. 1994, 33, 1434.
- 115. Bader, R. F. W.; Gillespie, R. J.; MacDougall, P. J. From Atoms to Polymers, in Molecular Structure and Energetics, Liebman, J.; Greenberg, A., Eds.; VCH: New York, 1989; Vol. 11; pp 1-51.
- 116. See, for example, Burford, N.; Mason, S.; Spence, R. E. v. H.; Whalen, J. M.; Richardson, J. F.; Rogers, R. D. Organometallics 1992, 11, 2241.
- 117. Cameron, T. S.; Borecka, B.; Kwiatkowski, W. J. Am. Chem. Soc. 1994, 116, 1211.
- 118. Bellamy, L. J. The Infrared Spectra of Complex Molecules; Chapman and Hall: London, 1975; p 75. Socrates, G. Infrared Characteristic Group Frequencies; Wiley and Sons: New York, 1980; p 85.
- 119. Aldrich Library of Infrared Spectra, Nicolet Instrument Corporation, 1988.
- Reference spectra were obtained from: <sup>13</sup>C; Kalinowski, H.O.; Berger, S.; Braun, S. Carbon-13 NMR Spectroscopy; Wiley: New York, 1981; <sup>1</sup>H The Sadtler Handbook of Proton NMR Spectra; Simons, W. W., Ed.; Sadtler Research Laboratories: Philadelphia, 1978.
- 121. Burford, N.; Clyburne, J. A. C.; Cameron, T. S.; Bakshi, P. K. Organometallics 1995, in press.
- 122. Taylor, R. *Electrophilic Aromatic Substitution*; John Wiley and Sons: New York, 1990.
- 123. Wheland, G. W. J. Am. Chem. Soc. 1942, 64, 900.

- 124. See for example, (a) Olah, G. A.; Lin, H. C.; Mo, Y. K. J. Am. Chem. Soc.
  1972, 94, 3667. (b) Olah, G. A.; Schlosberg, R. H.; Porter, R. D.; Mo, Y. K.; Kelly, D. P.; Mateescu, G. D. J. Am. Chem. Soc. 1972, 94, 2034.
- 125. (a) Lambert, J. B.; Zhang, S.; Stern, C. L.; Huffman, J. C. Science 1993, 260, 1917. (b) Lambert, J. B.; Zhang, S. J. Chem. Soc., Chem. Commun. 1993, 383.
- 126. Lambert, J. B.; Zhang, S.; Ciro, S. M. Organometallics 1994, 13, 2430.
- 127. Rathore, R.; Loyd, S. H.; Kochi, J. K. J. Am. Chem. Soc. 1994, 116, 8414.
- Olah, G. A.; Rasul, G.; Li, X.; Buchholz, H. A.; Sandford, G.; Prakash, G. K. S. Science 1994, 263, 983. For further discussion, see also (a) Pauling, L. Science 1994, 263, 983; (b) Lambert, J. B.; Zhang, S. Science 1994, 263, 984; (c) Reed, C.; Xie, Z. Science 1994, 263, 985.
- (a) Dewar, M. J. S. J. Chem. Soc. 1946, 406. (b) Dewar, M. J. S. J. Chem. Soc. 1946, 777. (c) Dewar, M. J. S. Nature 1945, 176, 784.
- 130. Olah, G. A. Acc. Chem. Res. 1971, 4, 240.
- 131. (a) Caille, S. Y.; Corriu, R. J. P. Tetrahedron 1969, 25, 2005. (b) Christy, P. F.; Ridd, J. H.; Stears, N. D. J. Chem. Soc. B 1970, 797. (c) Ridd, J. H. Acc. Chem. Res. 1971, 4, 248. (d) Taylor, R.; Tewson, T. J. J. Chem. Soc., Chem. Commun. 1973, 836.
- 132. (a) Brownstein, S.; Gabe, E. J.; Lee, F.; Tan, L. J. Chem. Soc., Chem. Commun. 1984, 1566. (b) Brownstein, S.; Morrison, A.; Tan, L. Can. J. Chem. 1986, 64, 265. (c) Brownstein, S.; Gabe, E. J.; Lee, F.; Piotrowski, A. Can. J. Chem. 1986, 64, 1661.
- 133. Prepared in situ as alluded to by Niecke, E.; Nixon, J. F.; Wenderoth, P.; Trigo Passos, B. F.; Nieger, M. J. Chem. Soc., Chem. Commun. 1993, 846.
- 134. See, for example, Eaborn, C. J. Organometal. Chem. 1975, 100, 43.
- 135. Schleyer, P. v. R.; Buzek, P.; Müller, T.; Apeloig, T.; Siehl, H. U. Angew. Chem. Int. Ed. Engl. 1993, 32, 1471.
- 136. Regitz, M. Chem. Rev. 1990, 90, 191.
- 137. Curtis, R. D.; Royan, B. W.; Wasylishen, R. E.; Lumsden, M. D.; Burford, N. *Inorg. Chem.* 1992, 31, 3386.
- 138. Arduenego, A. J. III; Dixon, D. A.; Kumashiro, K. K.; Lee, C.; Power, W. P.; Zilm, K. W. J. Am. Chem. Soc. 1994, 116, 6361.
- 139. Gudat, D.; Hoffbauer, W. H.; Niecke, E.; Schoeller, W. W.; Fleisher, U.; Kutzelnigg, W. J. Am. Chem. Soc. 1994, 116, 7325.

- See for example, (a) Sanchez, M.; Mazières, M. R.; Lamandé, L.; Wolf, R. In Multiple Bonds and Low Coordination in Phosphorus Chemistry; Regitz, M.; Scherer, O. J., Eds.; Georg Thieme Verlag: Stuttgart, 1990; pp 129-148. (b) Cowley, A. H.; Kemp, R. A. Chem. Rev. 1985, 85, 367.
- 141. Jameson, C. J.; De Dios, A.; Jameson, A. K. Chem. Phys. Lett. 1990, 167, 575.
- 142. Niecke, E.; Altmeyer, O.; Barton, D.; Detsch, R.; Gartner, C.; Hein, J.; Nieger, M.; Reichert, F. *Phosphorus, Sulfur and Silicon* **1990**, 49/50, 321.
- 143. Niecke, E.; Gudat, D. Angew. Chem. Int. Ed. Engl. 1991, 30, 217.
- 144. Zurmühlen, F.; Regitz, M. Angew. Chem. Int. Ed. Engl. 1987, 26, 83.
- 145. Hitchcock, P. B.; Lappert, M. F.; Rai, A. K.; Williams, H. D. J. Chem. Soc., Chem. Commun. 1986, 1633.
- 146. Niecke, E.; Detsch, R.; Nieger, M.; Reichert, F.; Schoeller, W. W. Bull. Soc. Chim. Fr. 1993, 130, 251.
- See, for example, (a) Burford, N.; Royan, B.W.; Linden, A.; Cameron, T.S. J. Chem. Soc., Chem. Commun. 1988, 842. (b) Burford, N.; Royan, B.W. J. Chem. Soc., Chem. Commun. 1989, 19. (c) Burford, N.; Dipchand, A. I.; Royan, B. W.; White, P. S. Inorg. Chem. 1990, 29, 4938.
- 148. (a) Couret, C.; Escudié, J.; Satgé, J.; Lazraq, M. J. Am. Chem. Soc., 1987, 109, 4411. (b) Lazraq, M.; Escudié, J.; Couret, C.; Satgé, J.; Dräger, M.; Dammel, R. Angew. Chem. Int. Ed. Engl. 1988, 27, 828.
- 149. Glaser, B.; Nöth, H. Angew. Chem. Int. Ed. Engl. 1985, 24, 416.
- 150. v. d. Knaap, T.; Bickelhaupt, F. Chem. Ber. 1984, 117, 915.
- 151. Appel, R.; Schmitz, R. Chem. Ber. 1983, 116, 3521.
- 152. Igau, A.; Baceiredo, A.; Grützmacher, H.; Pritzkow, H.; Bertrand, G. J. Am. Chem. Soc. 1989, 111, 6853.
- 153. See, for example, (a) Ehrig, M.; Horn, H.; Kölmel, C.; Ahlichs, R. J. Am. Chem. Soc. 1991, 113, 3701. (b) Lohr, L. L; Schlegel, H. B.; Morokuma, K. J. J. Phys. Chem. 1984, 88, 1981.
- 154. Heim, U.; Pritzkow, H.; Fleischer, U.; Grützmacher, H. Angew. Chem. Int. Ed. Engl. 1993, 32, 1359.
- 155. Burford, N.; Spence, R. E. v. H.; Richardson, J. F. J. Chem. Soc., Dalton Trans. 1991.
- 156. Grützmacher, H.; Pritzkow, H. Angew. Chem. Int. Ed. Engl. 1992, 31, 99.

- 157. Corbridge, D. E. C. Phosphorus: An Outline of its Chemistry, Biochemistry and Technology, 4th ed.; Studies in Inorganic Chemistry 10; Elsevier: New York, 1990; p 47.
- 158. Jutzi, P; Meyer, U.; Krebs, B.; Dartmann, M. Angew. Chem. Int. Ed. Engl. 1986, 25, 919.
- 159. Gudat, D.; Nieger, M.; Niecke, E. J. Chem. Soc., Dalton Trans. 1989, 693.
- 160. Reed, R.; Réau, R.; Dahan, F.; Bertrand, G. Angew. Chem. Int. Ed. Engl. 1993, 32, 399.
- 161. Schmidbaur, H.; Schier, A.; Milewski-Mahrla, B.; Schubert, U. Chem. Ber. 1982, 115, 722.
- 162. Schmidbaur, H.; Schier, A.; Frazão, C.; Müller, G. J. Am. Chem. Soc. 1986, 108, 976.
- 163. Excellent reviews of the chemistry and bonding in tetracoordinate phosphorus compounds (a) Gilheany, D. G. Chem. Rev. 1994, 94, 1339. (b) Bachrach, S. M.; Nitsche, C. I. In The Chemistry of Organophosphorus Compounds. Phosphonium Salts, Ylides, and Phosphoranes; Hartley, F. R., Ed.; John Wiley and Sons: Chichester, 1994.
- 164. Gray, G. A. J. Am. Chem. Soc. 1973, 95, 5092.
- 165. Gerkin, R. E.; Lundstedt, A. P.; Reppart, W. J. Acta. Cryst. 1984, C40, 1892.
- 166. Buchholz, S.; Harms, K.; Marsch, M.; Massa, W.; Boche, G. Angew. Chem. Int. Ed. Engl. 1989, 28, 72.
- 167. See, for example, (a) Ylides and Imines of Phosphorus; Johnson, A. W. Ed.; John Wiley and Sons, New York, 1993; p 153. (b) Schmidbaur, H.; Füller, H. -J.; Köhler, F. H. J. Organometal. Chem. 1975, 99, 353-357. (c) Schmidbaur, H. Acc. Chem Res. 1975, 62.
- 168. Grützmacher, H.; Pritzkow, H. Angew. Chem. Int. Ed. Engl. 1991, 30, 709.
- 169. Ylides and Imines of Phosphorus; Johnson, A. W. Ed.; John Wiley and Sons, New York, 1993; pp 53-56.
- 170. The structure of this red compound has puzzled other research groups, but the NMR spectrum at room temperature is complex and lack of crystalline material prevented conclusive identification. Bertrand, G. personal communication, Banff, 1994.
- 171. Socrates, G. Infrared Characteristic Group Frequencies; John Wiley and Sons: New York, 1980.
- 172. Nunn, A. J.; Schofield, K.; Theobald, R. S. J. Chem. Soc. 1952, 2797.

- 173. The Chemistry of Organophospilorus Compounds. Phosphonium Salts, Ylides, and Phosphoranes; Hartley, F. R., Ed.; John Wiley and Sons: Chichester, 1994.
- 174. See, for example, (a) Schier, A.; Wallis, J. M.; Muller, G.; Schmidbaur, H. Angew. Chem. Int. Ed. Engl. 1986, 25, 757. (b) Schmidbaur, H.; Bublak, W.; Huber, B.; Muller, G. Angew. Chem. Int. Ed. Engl. 1987, 26, 234.
- 175. Schmidbaur, H. Angew. Chem. Int. Ed. Engl. 1985, 24, 893.
- 176. Elschenbroich, Ch.; Salzer, A. Organometallics: A Concise Introduction; VCH: New York: 1989, p 91.
- 177. See, for historical interest, (a) Menshutkin, B. N. Chem. Zentr. 1910, 3, 378;
  (b) Menshutkin, B. N. Chem. Zentr. 1912, 4, 1436.
- 178. Mootz, D.; Händler, V. Z. Anorg. Allg. Chem. 1986, 533, 23.
- 179. See, for example, (a) Schmidbaur, H.; Wallis, J. M.; Nowak, R.; Huber, B.; Müller, G. Chem. Ber. 1987, 120, 1837. (b) Schmidbaur, H.; Nowak, R.; Schier, A.; Wallis, J. M.; Huber, B.; Müller, G. Chem. Ber. 1987, 120, 1829.
  (c) Hulme, R.; Mullen, D. J. E. J. Chem. Soc., Dalton Trans. 1976, 9, 804.
- 180. Schmidbaur, H.; Nowak, R.; Huber, B.; Müller, G. Organometallics 1987, 6, 2266.
- 181. Hulme, R.; Szymanski, J. T. Acta Cryst. 1969, 25b, 753.
- 182. Mootz, D.; Händler, V. Z. Anorg. Allg. Chem. 1985, 521, 122.
- 183. Bomberieri, G.; Peyronel, G.; Vezzosi, I. M. Inorg. Chim. Acta 1972, 6, 349.
- 184. Lipka, v. A.; Mootz, D. Z. Anorg. Allg. Chem. 1978, 440, 217.
- 185. Schmidbaur, H.; Nowak, R.; Steiglmann, O.; Müller, G. Chem. Ber. 1990, 123, 19.
- 186. (a) Lipka, v. A. Z. Anorg. Allg. Chem. 1980, 466, 217. (b) Lipka, v. A. Z. Anorg. Allg. Chem. 1978, 440, 224.
- 187. Hulme, R.; Hursthouse, M. B. Acta Cryst. 1966, A143.
- 188. Lipka, A. Acta. Cryst. 1979, B35, 3020.
- 189. The cell dimensions for 2:1 SbCl<sub>3</sub>:benzene have been re-ordered to make them directly comparable to 4:1 SbCl<sub>3</sub>:*t*-stilbene.
- 190. Data for this diagram was obtained from the Cambridge Structural Database, Crystal Data Center, 12 Union Road, Cambridge, UK.
- 191. Schmidbaur, H.; Nowak, R.; Steiglmann, O.; Müller, G. Chem. Ber. 1990, 123, 1221.

- 192. Schmidbaur, H.; Bublack, W.; Huber, B.; Muller, G. Angew. Chem. Int. Ed. Engl. 1987, 26, 234.
- 193. Burford, N.; Losier, P. unpublished results.

 $\mathcal{T}$ 

,

- 194. See, for example, (a) Pipal, R. J.; Grimes, R. N. Organometallics, 1993, 12, 4459 (b) Foucher, D. A.; Edwards, M.; Burrow, R. A.; Lough, A. J.; Manners, I. Organometallics 1994, 13, 4959. (c) Fyfe, H. B.; Mlekuz, M.; Zargarian, D.; Taylor, N. J.; Marder, T. B. J. Chem. Soc., Chem. Commun. 1991, 188. (d) Tenhaeff, S. C.; Tyler, D. R. J. Chem. Soc., Chem. Commun. 1989, 1459. (e) Brandt, P. F.; Rauchfuss J. Am. Chem. Soc. 1992, 114, 1926.
- 195. Burford, N.; Müller, J.; Parks, T. M. J. Chem Educ. 1994, 71, 807.
- 196. Fieser, L. F.; Fieser, M. Reagents for Organic Synthesis; John Wiley and Sons: New York, 1967; Vol. 1.